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I. Zusammenfassung 

Mesoporöse  Silikamaterialien  kommen  heutzutage  in  vielschichtigen  Anwendungsbereichen  zum 
Einsatz. Hierzu zählen die Anwendung von Silika als Packungsmaterialien in der Chromatographie, als 
Supportmaterialien  für  medizinische  Stoffe  für  deren  Transport  im  und  durch  den  Körper,  in  der 
Katalyse  oder  in  Form  von  Nanopartikeln  in  der  Nanomedizin.  Entscheidend  ist  in  allen 
Anwendungsgebieten die Beweglichkeit von Gastmolekülen in dem Mesoporensystem, welches durch 
das feste Silikagerüst räumlich begrenzt ist. Der diffusive Transport innerhalb der und durch die Poren 
ist dabei direkt mit deren Morphologie verknüpft: Aufgrund des begrenzten Porenraums und der daraus 
resultierenden sterischen Wechselwirkungen der Gastmoleküle mit den festen Silikawänden wird der 
Transport  der  Moleküle  im  Vergleich  zur  freien  Diffusion  gehindert.  Die  Hinderung  der  Diffusion 
kommt quantitativ in dem effektiven Diffusionskoeffizienten zum Ausdruck. Um Vorhersagen über die 
Transporteigenschaften  verschiedener  individueller  Silika  treffen  zu  können,  ist  es  notwendig, 
quantitative Ausdrücke für die Beziehung zwischen deren Morphologie und den daraus resultierenden 
Transporteigenschaften zu formulieren. Die Porenmorphologie des betrachteten Materials wird dabei 
oft nur grob mit Hilfe vereinfachter geometrischer Modelle beschrieben. Allerdings werden durch die 
Verwendung  vereinfachter  geometrischer  Modelle  individuelle  morphologische  Eigenschaften  wie 
Engstellen, Unregelmäßigkeiten oder andere Defekte wie sogenannte „dead­ends“ (Sackgassen), welche 

einen entscheidenden Einfluss auf die Transporteigenschaften eines individuellen Materials haben, nicht 
berücksichtigt,  sodass  die  Ergebnisse  solcher  Untersuchungen  von  Morphologie­Transport­
Zusammenhängen oft fehlerbehaftet und ungenau sind. Mit Hilfe von Elektronentomographie (ET) kann 
dagegen die dreidimensionale (3D) Struktur eines Materials exakt  rekonstruiert werden, sodass auch 
morphologische  Details  im  Nanometerbereich  aufgedeckt  werden  können  und  eine  korrekte 
Untersuchung  der  Morphologie  ermöglicht  wird.  Die  Rekonstruktionen  werden  in  dieser  Arbeit 
anschließend als geometrische Modelle für numerische Simulationen gehinderter Diffusion von Tracern 
verschiedener  Größe  durch  das  Mesoporensystem  sowie  innerhalb  des  Systems  verwendet.  Die 
resultierenden, von der Größe der Tracer abhängigen, Funktionen für die zugänglichen Porositäten und 
für  die  entsprechenden  Diffusionskoeffizienten  geben  dann  die morphologiebedingte  Hinderung  der 
Diffusion  im  zugrundeliegenden  individuellen  Mesoporensystem  exakt  wieder.  In  der  vorliegenden 
Arbeit  werden  mit  Hilfe  dieses  Ansatzes  verschiedene  Silikamaterialien  unterschiedlichster 
Morphologie hinsichtlich des Einflusses morphologischer Eigenschaften auf den gehinderten diffusiven 
Transport durch die und innerhalb der Mesoporen des jeweiligen Silikagerüstes analysiert. Ziel ist die 
Formulierung allgemeiner Gleichungen, die die exakte Vorhersage über die Güte eines Materials für ein 
bestimmtes Anwendungsgebiet mit Hilfe weniger materialspezifischen Parameter ermöglichen, ohne 
dass ein vereinfachtes geometrisches Modell von Nöten ist.    
Im  ersten  Kapitel  wird  eine  mit  mesoporösen  Silika­Partikeln  gepackten  chromatographische  Säule 
hinsichtlich  ihrer  inter­  und  intrapartikulären  Morphologie  und  deren  Einfluss  auf  ihre 
Trennungseigenschaften  in  der  Größenausschlusschromatographie  (SEC)  untersucht.  Im  Fokus  steht 
hierbei  eine  2.1 mm  I.D.  Säule  mit  vollporösen  Ethylene­Bridged­Hybride­(BEH­)Partikeln 
(Partikeldurchmesser 1.7 µm) als Packungs­material. Ziele der Studie umfassen (i) die Untersuchung 
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von  Morphologie­Transport/Retention­Zusammenhängen  für  SEC­Säulen,  (ii)  die  Bestimmung 
optimaler experimenteller Bedingungen, um entweder maximale Spitzenkapazität zu erreichen oder die 
Rate der Spitzenkapazität über den gesamten Trennungsbereich konstant zu halten, (iii) die Vorhersage 
und der Vergleich der Leistungsfähigkeit von core­shell­Partikeln (unporöser Kern, poröse Schale) in 
der Größenausschlusschromatographie mit der von vollporösen BEH­Partikeln, wobei die core­shell­
Partikel in der Schale die gleiche Morphologie besitzen wie die BEH­Partikel, und (iv) die Diskussion 
über mögliche Vorteile, welche die Anwendung des hier präsentierten Rekonstruktions­Simulations­
Ansatzes für die Beschleunigung der Methodenentwicklung in der Größenausschlusschromatographie 
mit  sich  bringen  könnte.  Zunächst  wird  die  Packungs­Mikrostruktur  der  Säule  mittels 
Rasterelektronenmikroskopie  mit  fokussiertem  Ionenstrahl  (FIB­SEM)  rekonstruiert,  sodass  etwa 
Packungseigenschaften  der  Bereiche  nahe  der  Säulenwand  mit  denen  im  Säulenzentrum  verglichen 
werden können. Mit Hilfe von Lattice­Boltzmann­Methoden­(LBM­)Simulationen des Flussprofils auf 
der  Makroporenskala  kann  der  Einfluss  der  Packungs­Mikrostruktur  auf  die  Flussgeschwindigkeit 
untersucht  werden.  Dabei  kann  eine  deutliche  radiale  Heterogenität  der  Packungsdichte  beobachtet 
werden,  welche  eine  deutliche  Abhängigkeit  der  (aus  LBM­Simulationen  erhaltenen) 
Flussgeschwindigkeit von der radialen Position in der Säule bewirkt. Der Fluss nahe der Säulenwand ist 
dabei viel schwächer als der im Säulenzentrum, wo die Fließgeschwindigkeit den gleichen Wert besitzt 
wie im unbegrenzten Bulk­Volumen. Diese radiale Änderung des Fließgeschwindigkeit kann mit dem 
aufgrund der Packungsweise ähnlich verlaufenden Druckprofil über die Säulenbreite erklärt werden. 
Während eine erhöhte Druckbelastung auf einzelne Silikapartikel in der Nähe der festen Säulenwand 
wirkt, kann sich der Druck auf beweglichere Partikel im Innern der Säule weiter verteilen. Des Weiteren 
kann  mittels  der  durch  FIB­SEM  erhaltenen  3D  Rekonstruktion  des  Packungsmaterials  im 
Säulenzentrum gezeigt werden, dass es sich bei der Packungsmikrostruktur in diesem Bereich um eine 
zufällige  Kugelpackung  handelt.  Anhand  der  Rekonstruktion  kann  eine  externe  (interpartikuläre) 
Porosität von εe = 0.39 ermittelt werden. Das Mesoporenvolumen innerhalb der BEH­Partikel wird mit 
Hilfe  von  ET  rekonstruiert  und  im  Folgenden  als  geometrisches  Modell  für  Diffusionssimulationen 
verwendet. Numerische Simulationen gehinderter Diffusion basierend auf der  random­walk particle­
tracking­(RWPT­)Technik  werden  im  Mesoporenraum  der  3D  Rekonstruktionen  durchgeführt, 
wodurch die intrapartikuläre Porosität ε sowie der lokale diffusive Hinderungsfaktor Kd als Funktionen 
von  λ,  dem  Verhältnis  zwischen  mittlerer  Porengröße  und  dem  Durchmesser  der  Tracer,  erhalten 
werden. Für Punkttracer wird eine zugängliche intrapartikuläre Porosität von ε0 = 0.49 mit Hilfe dieses 
Vorgehens  bestimmt.  Die  erhaltene  echte  Porositätsfunktion  wird  mit  der  unter  der  Annahme  einer 
zylindrischen oder einer sphärischen Porenform zugänglichen Porosität verglichen, wobei eine starke 
Abweichung  insbesondere  vom  zylindrischen  Porenmodell  beobachtet  wird.  Des  Weiteren  wird  die 
Analogie zwischen der Funktion für den lokalen diffusiven Hinderungsfaktor mit dem wohlbekannten 
RENKIN­Modell geprüft. Hierbei wird insbesondere für große λ eine deutliche Abweichung beobachtet. 
Für Punkttracer wird des Weiteren die intrapartikuläre Tortuosität mit τ0 = 1.95 anhand des effektiven 
Diffusionskoeffizienten für λ = 0 bestimmt. Die 3D Rekonstruktion der Säulenmikrostruktur (mittels 
FIB­SEM) und die des intrapartikulären Mesoporenvolumens (basierend auf ET) werden im Folgenden 
als  geometrische  Modelle  für  Diffusionssimulationen  basierend  auf  dem  hierarchischen 
Diffusionsmodell verwendet, wobei sich Tracer verschiedener Größen sowohl durch das Makroporen­
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volumen zwischen den BEH­Partikeln als auch durch den intrapartikulären Mesoporenraum bewegen, 
sodass sich die effektive Diffusivität Dbed als Funktion der Tracergröße ergibt. Die effektive Diffusivität 
spiegelt das Zusammenspiel zwischen der inter­ und intrapartikulären Diffusivitäten der Tracer wider, 
woraus ein parabolisches Verhalten von Dbed als Funktion der Tracergröße resultiert. Zunächst haben 
kleine  Tracer  sowohl  Zugang  zur  inter­  als  auch  zur  intrapartikulären  Porosität,  sodass  Dbed  mit 
steigender  Tracergröße  kleiner  wird,  bis  es  ab  einer  gewissen  Tracergröße  erneut  steigt,  da  die 
zugängliche  intrapartikuläre  Porosität  deutlich  kleiner  wird  als  das  zugängliche  interpartikuläre 
Volumen  und  Dbed  infolgedessen  hauptsächlich  durch  die  Diffusion  im  externen  (interpartikulären) 
Raum  bestimmt  wird.  In  der  Studie  werden  des  Weiteren  die  gewonnenen  Erkenntnisse  über  die 
Zusammenhänge zwischen Morphologie und Transporteigenschaften der vorliegenden Säule genutzt, 
um Vorhersagen über das Leistungsverhalten von SEC­Experimenten unter verschiedenen Bedingungen 
zu  treffen.  Es  wird  gezeigt,  dass  optimale  experimentelle  Bedingungen  anhand  der  vorhergehenden 
Erkenntnisse  einfach  voraussagbar  sind,  sodass  zentrale  leistungslimitierende  Parameter  in  SEC­
Experimenten, wie etwa die globale Spitzenkapazität oder die Rate der Spitzenkapazität, kontrolliert 
werden  können.  Unter  bestimmten  Voraussetzungen  ist  es  so  möglich,  bei  bekannter  mittlerer 
Porengröße  und  Temperatur  die  optimale  Flussrate  zu  bestimmen,  um  die  maximale  globale 
Spitzenkapazität  zu  erzielen.  Des  Weiteren  wird  der  Einfluss  von  core­shell­Partikeln  als 
Packungsmaterial  auf  die  Rate  der  Spitzenkapazität  für  verschiedene  ρ  (Verhältnis  zwischen  dem 
Durchmesser des unporösen Kerns zu dem des gesamten Partikels) untersucht. Hierbei wird eine höhere 
Spitzenkapazität  für  steigende  ρ  beobachtet,  wobei  sich  jedoch das  Separationsfenster  im  Vergleich 
schneller schließt als bei vollporösen Partikeln. Für die Praxis bedeutet dies, dass die Anwendung von 
core­shell­Partikeln  in  der  Größenausschlusschromatographie  von  Vorteil  ist,  wenn  im  Experiment 
einfache Analytgemische sowie ein geringer Anteil an Analyten mit hohem Molekulargewicht enthalten 
ist.     
Im zweiten Kapitel werden Zusammenhänge zwischen Morphologie und Transporteigenschaften zweier 
mesoporöser Silikamaterialien mit geordneten Porensystemen, SBA­15 und KIT­6, untersucht. SBA­15 
besitzt  ein  eindimensionales  Primärporensystem,  welches  aus  hexagonal  angeordneten 
zylinderförmigen  Poren  besteht,  sowie  ein  sekundäres  Porensystem,  bei  welchem  sich  die  Poren  an 
zufälligen Positionen  in den amorphen Silikawänden zwischen den Primärporen befinden und somit 
diffusiver Transport in alle Richtungen durch das geordnete Porensystem ermöglicht wird. Bei KIT­6 
dagegen wird das dreidimensionale Primärporensystem aus zwei ineinander verwobenen Netzwerken 
zylinderförmiger Poren gebildet, während sich wie bei SBA­15 die sekundären Poren (meist Mikro­ und 
kleine  Mesoporen)  in  den  Silikawänden  befinden  und  somit  eine  hohe  Interkonnektivität  der 
Primärporen gegeben ist. Die Strukturparameter der beiden Materialen werden zunächst mit Hilfe von 
Röntgendiffraktometrie  (XRD)  sowie  mittels  Stickstoffphysisorptionsanalyse  ermittelt.  Mittels  ET 
erhaltene dreidimensionale Rekonstruktionen der Porenräume von SBA­15 sowie von KIT­6 werden 
anschließend als geometrische Modelle für numerische Simulationen diffusiven Transports durch die 
Materialien verwendet. Mit Hilfe der RWPT­Technik können so die zugänglichen Porositäten sowie die 
entsprechenden  effektiven  Diffusionskoeffizienten  ermittelt  und  als  Funktionen  von  λ  dargestellt 
werden. Für Punkttracer (λ = 0) werden zugängliche Porositäten von ε0 = 0.69 und ε0 = 0.70 für SBA­
15 und KIT­6 erhalten. Aus den Diffusionskoeffizienten bei λ = 0 kann direkt die diffusive Tortuosität 
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für  beide  Systeme  errechnet  werden  (SBA­15:  τ0  = 1.41;  KIT­6:  τ0 =  1.31).  Des  Weiteren  wird  der 
Zusammenhang zwischen der Porosität und der zugehörigen Tortuosität als Funktionen von λ näher 
untersucht, indem ε(λ) als Funktion von τ(λ) betrachtet wird. Es wird gezeigt, dass der globale und der 
lokale  Diffusionskoeffizient  für  gehinderte  Diffusion  über  die  Tortuosität  für  Bereiche  kleiner 
Tracergrößen  (kleine  λ)  mit  dem  Gesetz  von  ARCHIE  und  für  größere  Tracer  (große  λ)  mit  der 
WEISSBERG­Gleichung  abgeschätzt  werden  kann.  Diese  Vorgehensweise  ist,  basierend  auf  den  hier 
gewonnenen Erkenntnissen, eine gleichwertige Alternative zur Bestimmung des globalen und lokalen 
Diffusionskoeffizienten mit Hilfe des Rekonstruktions­Simulations­Ansatzes. Unter Zuhilfenahme der 
gewonnen  Erkenntnisse  über  geordnete  Silikamaterialien  werden  die  Morphologie­Diffusions­
Eigenschaften mit denen ungeordneter Silikamaterialien verglichen. Es wird festgestellt, dass geordnete 
Silika hinsichtlich der Selektivität bessere Ergebnisse liefern als ungeordnete. Dies liegt an der engeren 
Porengrößenverteilung von KIT­6 und SBA­15, was zu einem schnelleren Absinken der Diffusion für 
größere  Tracerdurchmesser  führt,  als  es  bei  ungeordneten  Silika  der  Fall  ist.  Der  Transport  durch 
geordnete Materialien ist jedoch auch im Falle einer stark vernetzten dreidimensionalen Porenstruktur 
nicht  signifikant  effizienter  als  bei  ungeordneten  Silika.  Zusätzlich  führt  eine  engere 
Porengrößenverteilung bei geordneten Silika zu einer erhöhten Sensitivität gegenüber Engstellen oder 
anderen  Unregelmäßigkeiten  im  Porensystem,  sodass  in  dieser  Studie  kein  klarer  Vorteil  der 
Verwendung  geordneter  gegenüber  ungeordneten  Materialien  als  Packungsmaterialien  in  der 
Chromatographie oder als Supportstrukturen aufgezeigt werden kann.    
Das dritte Kapitel befasst sich mit der Analyse von Morphologie­Transport­Zusammenhängen für zwei 
ungeordnete  mesoporöse  Silikamaterialien  verschiedener  mittleren  Porengrößen,  Si60  mit  einer 
mittleren Porengröße von dmeso = 5.9 nm und Si100 mit dmeso = 13.0 nm. Mit Hilfe von ET werden die 
Porenräume von Si60 und Si100 rekonstruiert und anschließend als geometrische Modelle für RWPT­
Simulationen  innerhalb  der  Mesoporen  genutzt.  Ähnlich  zu  Kapitel  zwei  liegt  der  Fokus  auf  der 
Bestimmung der zugänglichen Porositäten sowie der effektiven Diffusionskoeffizienten, welche passive 
Tracer verschiedener Größe aufgrund der sterischen und hydrodynamischen Wechselwirkung mit den 
festen Silikawänden bei der Diffusion durch den Mesoporenraum beider Materialien erfahren. Aus den 
reziproken  Diffusionskoeffizienten  für  Punkttracer  (λ  =  0)  können  mit  dieser  Technik  diffusive 
Tortuositäten  ermittelt  werden.  Eine  Betrachtung  der  zugänglichen  Porositäten  sowie  der  effektiven 
Diffusionskoeffizienten beider Materialien als Funktionen von λ zeigt, dass die Diffusion von Tracern 
in Si60 mit zunehmender Größe stärker gehindert wird als es bei Si100 für gleiche Tracergrößen der 
Fall  ist.  Infolgedessen  kann  bei  Si60  von  einer  höheren  Selektivität  bei  der  Zugänglichkeit  der 
Porenräume  für  Moleküle  verschiedener  Größen  ausgegangen  werden,  was  etwa  bei  der 
Immobilisierung oder dem Transport von Schlüsselmolekülen oder der größenselektiven Formierung 
neuer Spezies innerhalb der Mesoporen von zentralem Interesse ist. Im Fokus dieser Studie steht der 
Prozess der Ringschlussmetathese eines α,ω­Diens zum makro(mono)­zyklisierten (MMC) Produkt und 
zum Oligomer mit Hilfe eines innerhalb der Mesoporen immobilisierten Hoveyda­Grubbs­Katalysators 
der  zweiten  Generation.  Mittels  „diffusion ordered“ Kernspinresonanzspektroskopie  (DOSY­NMR) 
können die hydrodynamischen Durchmesser der relevanten Spezies (MMC Produkt, Oligomer, Substrat 
und Katalysator) bestimmt und in die zuvor erhaltenen, von der Tracergröße abhängigen, Funktionen 
der Porosität und des Diffusionskoeffizienten eingesetzt werden, um realistische Vorhersagen über die 
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Zugänglichkeit  der  Poren  für  die  entsprechende  Spezies  und  deren  Beweglichkeit  im  Porensystem 
treffen  zu  können.  Es  wird  gezeigt,  dass  bereits  durch  den  ersten  größenselektiven  Schritt,  der 
Immobilisierung  des  Katalysators  innerhalb  der  Poren,  bei  Si60  durch  die  erhebliche  räumliche 
Begrenzung sowie der infolgedessen langsameren Diffusion des Katalysators innerhalb des und durch 
das  Porensystem  von  einer  (bei  gleicher  Reaktionsdauer)  deutlich  geringeren  Aufnahme  des 
Katalysators auszugehen ist als im Falle von Si100. Aufgrund der Größe des Oligomers im Vergleich 
zu dem kleineren MMC Produkt sowie dem Substrat ist die Entstehung des MMC Produkts gegenüber 
der des Oligomers in beiden Materialien bevorzugt, wobei der Effekt in Si60 stärker ausgeprägt ist. Mit 
der beobachteten höheren Selektivität von Si60 geht auch eine stark reduzierte Mobilität der Spezies 
innerhalb  der Mesoporen  einher,  was  zu  einem  Verlust  in  der  Effizienz  und  Ausbeute  während der 
Reaktion führen kann. Der hier präsentierte Ansatz für die Quantifizierung von Transporteigenschaften 
mesoporöser  Silika  ist  allgemein  anwendbar  auf  verschiedenste  Mikroreaktionssysteme mit  anderen 
Katalysatoren, Substraten und Reaktionsprodukten, um entsprechende Selektivitäten und Effizienzen 
der betrachteten Systeme zu ermitteln. Mit den dadurch erhaltenen Informationen und zusammen mit 
komplementären experimentellen Daten ist es möglich, komplexe Reaktionswege aufzudecken sowie 
Mechanismen der Olefinmetathese besser zu verstehen.  
Der Fokus des vierten Kapitels liegt auf einer Untersuchung der radialen Abhängigkeit der Morphologie 
in  dendritischen  mesoporösen  Silikananopartikeln  (DMSNs).  Es  werden  vier  verschiedene  DMSNs 
miteinander  verglichen,  welche  jeweils  mit  leicht  abgewandelten  Parametern  in  einem  typischen 
Mikroemulsionssystem  hergestellt  wurden.  Bei  der  Synthese  wurde  vor  allem  der  Aspekt  der 
unabhängigen Justierung von Partikelgröße und Porengrößenverteilung berücksichtigt, sodass die in der 
Studie  untersuchten  DMSNs  jeweils  paarweise  miteinander  verglichen  werden  können:  Zwei  der 
Partikel besitzen eine vergleichbare Porengrößenverteilung, jedoch unterschiedliche Partikelgrößen (i), 
während  die  beiden  anderen  DMSNs  bei  gleicher  Partikelgröße  versetzte  Porengrößenverteilungen 
aufweisen  (ii).  Mittels  ET  werden  3D  Rekonstruktionen  der  vier  DMSNs  erhalten,  welche  als 
geometrische Vorlagen für die radiale Analyse zentraler morphologiespezifischer Deskriptoren genutzt 
werden. Zur radialen Analyse werden jeweils konzentrische, hohle Kugeln mit definiertem Radius und 
Manteldicke  um  den  Mittelpunkt  der  DMSNs  ausgeschnitten  und  die  Porosität  des  entsprechenden 
Segments sowie eine Sehnenlängenverteilung (CLD) im Porenraum des Segments wird für verschiedene 
Radien ermittelt. Die Porosität sowie die mittlere Sehnenlänge (d.h. der mittlere Abstand zwischen zwei 
gegenüberliegenden  Silikawänden)  kann  dann  in  Abhängigkeit  von  dem  äußeren  Radius  des 
betrachteten Segments aufgetragen werden. Es wird bei allen DMSNs ein signifikanter Porositätsverlust 
an der Oberfläche der Partikel festgestellt, welcher vermutlich aus dem Anschmelzen der Oberfläche 
beim  Kalzinierungsschritt  während  der  Synthese  resultiert.  Die  geringere  Porosität  an  der 
Partikeloberfläche hat erheblichen Einfluss auf die Zugänglichkeit des intrapartikulären Porenraums für 
Moleküle  von  außen  sowie  deren  Austrittsgeschwindigkeit  aus  den  DMSNs,  was  etwa  bei  der 
Anwendbarkeit  und  Effizienz  der  DMSNs  als  Trägermaterialien  für  Medikamente  und  bei  deren 
kontrollierter Freigabe im menschlichen Körper berücksichtigt werden muss. Durch die Betrachtung der 
mittleren  Sehnenlänge  als  Funktion  des  Radius  kann  des  Weiteren  die  Verbreiterung  des 
Porendurchmessers mit steigendem Radius für eines der untersuchten DMSNs festgestellt werden. Eine 
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solche konische Porenform ist besonders günstig, etwa wenn die Porenraum auch für große Moleküle 
(wie DNS oder RNS) zugänglich sein soll. 

Zusammenfassend  wird  mit  der  vorliegenden  Arbeit  gezeigt,  dass  mit  der  hier  zugrundeliegenden 
Analysemethode, d.h. die Nutzung dreidimensionaler Rekonstruktionen als geometrische Modelle für 
spätere  Diffusionssimulationen  sowie  zur  quantitative  Analyse  der  Morphologie  und  der  damit 
einhergehenden  Transporteigenschaften  eines  individuellen  Materials,  unterschiedlichste 
Silikamaterialien  im  Hinblick  auf  verschiedenste  Fragestellungen  analysiert  werden  können:  Die 
Auswirkung  verschiedener  Syntheseparameter  auf  morphologische  Eigenschaften,  der  Einfluss  der 
Porenordnung  und  ­geometrie  auf  die  Effizienz  und  Leistung  eines  Materials  etwa  in  der 
Chromatographie oder Katalyse, oder die radiale Betrachtung der intrapartikulären Porenmorphologie, 
um  nur  ein  paar  wenige  Beispiele  zu  nennen.  Es  wird  deutlich,  dass  mit  Hilfe  vereinfachter 
geometrischer Modelle (z.B. zylindrischer oder sphärischer Porenmodelle) meist die sehr individuelle 
Porenstruktur mesoporöser Silika nicht  adäquat beschrieben werden kann.  In dieser Arbeit wird  ein 
Ansatz  präsentiert,  welcher  die  Quantifizierung  von  Morphologie  und  Transporteigenschaften 
individueller Materialien auf einfache Art und Weise ermöglicht, sodass im weiteren Schritt exaktere, 
allgemein gültige Ausdrücke für bestimmte Klassen von Silikamaterialien formuliert werden können. 
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II. Abstract 

Today, mesoporous silica are employed in a wide field of applications. They are used for example as 
packing materials in chromatography, as support materials for the transport of medical agents within 
and through the human body, in catalysis or as nanoparticles in nanomedicine. In particular, the mobility 
of  guest  molecules  within  the  mesopore  system,  which  is  spatially  confined  by  the  solid  silica 
framework, is a central aspect in all fields of application. The diffusive transport within and through the 
mesopores is directly connected to the pore morphology: Due to the spatial confinement of the pore 
volume and the resulting steric interactions of the guest molecules with the solid silica walls the transport 
of  the molecules is hindered compared to diffusion within free space. This hindrance of diffusion is 
quantitatively expressed by the effective diffusion coefficient. To make predictions about the transport 
properties of different  individual  silica,  the formulation of quantitative expressions  that describe  the 
relationship between their morphological features and  the resulting  transport properties  is necessary. 
The pore morphology of the investigated material is often just roughly described by using simplified 
geometrical  models.  However,  by  using  simplified  geometrical  models,  individual  morphological 
aspects like constrictions, irregularities or other defects like dead­ends within the pore system, which 
have a  significant  influence on  the  transport properties of  an  individual material,  are not  taken  into 
account,  often  leading  to  defective  and  inaccurate  results  in  investigations  of  morphology­transport 
relationships. By means of electron tomography (ET),  the real three­dimensional  (3D) structure of a 
material can be reconstructed, uncovering morphological details on the nanoscale and making a most 
accurate investigation of the morphology possible. In this work, the reconstructions are subsequently 
employed as geometrical models for numerical simulations of hindered diffusion within and through the 
mesopore  system  for  different  tracer  sizes.  The  resulting  tracer  size  dependent  functions  for  the 
accessible  porosity  and  for  the  corresponding  diffusion  coefficients  then  accurately  reflect  the 
morphology­induced hindrance of diffusion for the underlying individual pore system. In the present 
work,  this  approach  is  employed  for  the  investigation of various  silica with  fundamentally different 
morphology  by  means  of  hindered  diffusive  transport  within  and  through  the  mesopores  of  the 
correspondent silica framework. The goal of this work is the formulation of general expressions, which 
enable  accurate predictions  about  the performance of  the  investigated material  for  a  certain  field of 
application,  using  only  a  few  material­specific  parameters  and  avoiding  the  need  of  an  underlying 
simplified geometrical model.    
In the first chapter a chromatographic column packed with mesoporous silica particles is investigated in 
terms of its inter­ and intraparticle morphology and its influence on separation characteristics in size 
exclusion chromatography (SEC). The work  is  focused on a 2.1 mm I.D. column packed with  fully 
porous  ethylene­bridged­hybride­(BEH)particles  (1.7  µm  particle  diameter).  The  goals  of  the  study 
include (i) the investigation of morphology­transport/retention relationships for SEC­columns, (ii) the 
determination of optimal experimental conditions to either maximize peak capacity or assure a constant 
rate of peak capacity over the whole separation window, (iii) the prediction and comparison of SEC­
performance  of  core­shell  particles  (nonporous  core,  porous  shell)  possessing  the  same  mesopore 
morphology  in  their  shell  as  fully  porous  BEH  particles,  and  (iv)  the  discussion  about  possible 
advantages  if  the  presented  reconstruction­simulation  approach  is  employed  in  order  to  accelerate 
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method development in SEC. First, the packing microstructure is reconstructed by means of focused ion 
beam  scanning  electron  microscopy  (FIB­SEM)  in  order  to  compare  the  packing  characteristics  of 
sections close to the wall of the column with those in the column center. By means of macropore­scale 
lattice­Boltzmann  method  (LBM)  simulations  of  the  flow  profile  the  influence  of  the  packing 
microstructure  on  the  flux  velocity  can  be  investigated.  A  clear  radial  heterogeneity  of  the  packing 
density  is  observed,  causing  an  obvious  dependence  of  the  flux  velocity  (obtained  from  LBM 
simulations) on the radial position within the column. The flux close to the column walls is much weaker 
than in the center of the column, where the flux velocity is equal the velocity within unrestricted bulk 
volume. This radial variation of the flux velocity can be explained by the similar shape of the pressure 
profile that results from the packing procedure. While single silica particle close to the solid column 
wall experience an enhanced compressive stress, the pressure is more distributed over the rather flexible 
particles located in the bulk volume of the column center. Furthermore, the 3D reconstruction of the 
packing material in the column center obtained by FIB­SEM proofs the existence of a random sphere 
packing in this area of the column. From the reconstruction, an external (interparticle) porosity of the 
column of εe = 0.39 is determined. By means of ET, the mesopore volume inside the BEH particles is 
reconstructed and subsequently employed as geometrical model for diffusion simulations. Numerical 
simulations of hindered diffusion are conducted within the mesopore volume based on the random­walk 
particle­tracking­(RWPT­)technique,  and  the  intraparticle  porosity  ε  as  well  as  the  local  diffusive 
hindrance  factor  Kd  can  be  obtained  as  functions  of  λ,  the  ratio  of  mean  mesopore  size  and  tracer 
diameter.  For  pointtracers,  an  accessible  intraparticle  porosity  of  ε0  =  0.49  is  obtained  with  this 
technique.  The  derived  porosity  function  is  compared  to  the  accessible  porosity  when  assuming  a 
cylindrical or  a  spherical pore geometry, demonstrating a particularly high discrepancy  in  case of a 
cylindrical pore model. Furthermore, the analogy between the obtained function for the local diffusive 
hindrance factor and the well­known RENKIN model is certified. Here, a significant deviance is observed 
especially  for  large  λ.  For  pointtracers,  an  intraparticle  tortuosity  value  of  τ0  =  1.95  is  furthermore 
determined from the effective diffusion coefficient at λ = 0. The reconstructed bulk stack (based on FIB­
SEM) and the 3D reconstruction of the intraparticle mesopore volume (based on ET) are subsequently 
employed as geometrical models for diffusion simulations using the hierarchical diffusion model, where 
tracers of different sizes move through both the macropore volume between the BEH particles and the 
intraparticle mesopore space, providing tracer size dependent values for the effective diffusivity Dbed. 
The  effective  bed  diffusivity  reflects  the  interplay  of  the  inter­  and  intraparticle  tracer  diffusivities, 
resulting in a parabolic behavior of Dbed as function of λ. First, small tracers have access to both inter­ 
and intraparticle porosity, causing Dbed to decrease with increasing tracer sizes until at a certain λ, Dbed 
starts to increase again since the accessible intraparticle porosity becomes significant smaller than the 
accessible  interparticle  volume  and  the  diffusion  is mainly  controlled  by  the  external  (interparticle) 
diffusion. The observed morphology­transport relationships are furthermore used to make presumptions 
about  the  performance  of  SEC­experiments  under  certain  experimental  conditions.  It  is  shown  that 
optimal experimental conditions can easily be diagnosed with help of the precedent findings, so that 
central parameters limiting the performance of SEC­experiments, for example the global peak capacity 
or the rate of peak capacity, can be controlled. Under certain conditions it is then possible to determine 
the optimal flow rate for a given temperature and mean pore size in order to maximize the global peak 
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capacity.  Additionally,  the  influence  of  core­shell  particles  as  packing  material  on  the  rate  of  peak 
capacity is investigated for different ρ (ratio between the diameter of the nonporous core to the diameter 
of the whole particle). Here, an increasing peak capacity is found for increasing ρ, though meanwhile a 
more rapidly closing separation window is observed. This means in praxis that the application of core­
shell particles in SEC­experiments is advantageous, if simple analyte mixtures as well as a small amount 
of high molecular mass analytes are present in the experiment.    
In the second chapter morphology­transport relationships of two mesoporous silica with ordered pore 
systems, SBA­15 and KIT­6, are investigated. SBA­15 exhibits a 1D primary pore system, which is 
composed of hexagonally arranged cylindrical pores, as well as a secondary pore system, where  the 
pores are located at random positions within the amorphous silica walls around the primary pores and 
therefore allow for diffusive transport in all directions through the ordered pore system. The 3D primary 
pore system of KIT­6, in contrast, is built of two interpenetrating networks of cylindrical pores, while 
the secondary pores (mostly micro­ and small mesopores) are located like in SBA­15 within the silica 
walls  and  therefore  provide  for  a  high  interconnectivity  of  the  primary  pores.  First,  the  structural 
parameters of both materials are examined by means of X­Ray diffraction analysis (XRD) and nitrogen 
physisorption analysis. 3D reconstructions of the pore spaces of SBA­15 and KIT­6 obtained from ET 
are  subsequently  employed  as  geometrical  models  for  numerical  simulations  of  diffusive  transport 
through the materials. By means of the RWPT­technique the accessible porosity for pointtracers and the 
corresponding  effective  diffusion  coefficients  can  be  determined  and  shown  as  functions  of  λ.  For 
pointtracers (λ = 0), accessible porosities of ε0 = 0.69 und ε0 = 0.70 are obtained for SBA­15 and KIT­
6, respectively. The diffusive tortuosity can be directly determined for both systems from the respective 
diffusion coefficient (SBA­15: τ0 = 1.41; KIT­6: τ0 = 1.31). Furthermore, the relationship between the 
porosity and the corresponding tortuosity as functions of λ are investigated in more detail by considering 
ε(λ) as a function of τ(λ). It is shown that the global and local diffusion coefficient for hindered diffusion 
can be evaluated by ARCHIE`s law for small tracer sizes (small λ), while for larger tracers (large λ) the 
WEISSBERG equation is more appropriate for this purpose. This approach is found to be an equally good 
alternative to the reconstruction­simulation­approach for the determination of the global and the local 
diffusion  coefficient.  With  the  use  of  previous  findings  about  ordered  silica  from  this  study,  the 
morphology­transport relationships are compared with those found for unordered silica. It is shown that 
ordered silica give better results in terms of higher selectivities than observed for unordered silica. This 
comes from the narrower pore size distribution of KIT­6 and SBA­15, which leads to a faster decline of 
the diffusion with increasing tracer sizes than it is the case for unordered silica. However, the transport 
efficiency through and within ordered materials is not significantly better than for unordered silica, even 
in  case  of  a  highly  interconnected  ordered  3D  pore  structure.  Additionally,  a  narrower  pore  size 
distribution causes an enhanced sensitivity against constrictions or other irregularities within the pore 
system of ordered silica, so that no clear advantage of using ordered silica rather than unordered silica 
as packing materials in chromatography or as support structures can be pointed out in this study.  
The  third  chapter  deals  with  the  analysis  of  morphology­transport  relationships  for  two  unordered 
mesoporous silica of different mean pore sizes, Si60 with a mean pore size of dmeso = 5.9 nm and Si100 
with  dmeso =  13.0  nm.  By means  of  ET,  the  pore  volumes  of  Si60 and  Si100  are  reconstructed  and 
subsequently employed as geometrical models for RWPT­simulations within the mesopores. Similarly 
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to the second chapter, the main aspect of this study is the determination of the accessible porosities as 
well as the effective diffusion coefficients experienced by passive tracers of different sizes due to the 
steric  and  hydrodynamic  interaction  with  the  solid  silica  walls  during  their  diffusion  through  the 
mesopore space of each material. With this technique, diffusive tortuosities can be directly determined 
from  the  inverse  diffusion  coefficients  for  pointtracers  (λ  =  0).  The  investigation  of  the  accessible 
porosities and the diffusion coefficients of both materials as functions of λ shows a stronger hindrance 
of the tracer diffusion with increasing tracer sizes for Si60 than observed for Si100 for similar tracer 
sizes. As a result, the access to the mesopores of Si60 can be assumed to be more selective to different 
tracer  sizes,  which  is  of  highest  interest  when  it  comes  to  the  immobilization  or  transport  of  key 
molecules or the size selective formation of new species within the mesopores, for example. This study 
is focused on the ring­closing metathesis of a α,ω­diene to the macro(mono)­cyclization (MMC) product 
and  to  the  oligomer  by  means  of  a  Hoveyda­Grubbs­catalysator  of  the  second generation,  which  is 
immobilized  inside  the  mesopores.  Via  diffusion  ordered  nuclear  magnetic  resonance  spectroscopy 
(DOSY­NMR) the hydrodynamic diameters of the relevant species (MMC product, oligomer, substrate, 
and catalyst) can be determined and inserted into the previously obtained tracer size dependent functions 
of the porosities and the diffusion coefficients to allow for accurate predictions about the accessibility 
of the mesopores for the correspondent species as well as its mobility within the pore system. It is shown 
that already the first size selective step, the immobilization of the catalyst within the pores, lets assume 
a  significantly  smaller  catalyst  uptake  (at  similar  reaction  time)  for  Si60  than  for  Si100  due  to  the 
substantial spatial confinement and therefore much slower diffusion of the catalyst within and through 
the pore space of Si60. Because of the larger oligomer size in contrast to the smaller MMC product as 
well as the smaller substrate, the formation of the MMC product is preferred over that of the oligomer 
for  both  materials,  which  is  more  pronounced  for  Si60  than  for  Si100.  With  the  observed  higher 
selectivity of Si60 comes along a strongly reduced mobility of the species within the mesopores, which 
may cause a smaller reaction efficiency and yield. The presented approach towards the quantification of 
transport  properties  of  mesoporous  silica  is  generally  applicable  to  most  different  micro­reaction 
systems  with  various  catalysts,  substrates  and  reaction  products,  to  determine  the  correspondent 
selectivities and efficiencies of  the considered systems. Together with  the obtained  information and 
complementary experimental data, it is possible to uncover complex reaction paths and to enable a better 
understanding of mechanisms within olefin metathesis.   
The fourth chapter is focused on the investigation of the radial dependence of the morphology within 
dendritic mesoporous silica nanoparticles (DMSNs). Four different DMSNs are compared, which were 
all  synthesized  in  a  typical  microemulsion  system  with  slightly  modified  reaction  parameters.  For 
synthesis,  the  primary  aspect  was  the  independent  adjustment  of  the  particle  size  and  the  pore  size 
distribution,  enabling  a  pairwise  investigation  of  the  four  DMSNs:  Two  of  the  DMSNs  exhibit  a 
comparable pore size distribution, but different particle sizes (i), while the other two DMSNs possess 
shifted PSDs at similar particle size (ii). By means of ET, 3D reconstructions of the four DMSNs are 
obtained,  serving  as  geometrical  models  for  the  radial  analysis  of  essential  morphology­specific 
descriptors. For radial investigation, concentric hollow spheres with defined radius and coat thickness 
are  cut  from  the  center  point  of  the  DMSNs  and  the  segment  porosity  as  well  as  the  chord  length 
distribution (CLD) within the segment mesopore space is determined for different radii. The porosity 
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and  the  mean  chord  length  (i.e.,  the  mean  distance  between  two  opposite  silica  walls)  can  then  be 
considered  as  functions  of  the  outer  segment  radius.  For  all  DMSNs,  a  significant  porosity  loss  is 
observed at  the  surface of  the particles, which  is assumed to  result  from  the calcination step during 
synthesis, leading to a partial melting of the particle surface. The smaller porosity at the particle surface 
has massive  influence on  the accessibility of the  intraparticle mesopore space for molecules coming 
from the outside as well as their exit velocity from inside  the DMSNs, which needs to be taken into 
account when considering the applicability and efficiency of DMSNs as drug carriers and for controlled 
drug release in the human body. Furthermore, the investigation of the mean chord length as a function 
of  the  radius  reveals  a  broadening  of  the  pore  diameter  within  increasing  radius  of  the  considered 
segment  for  one  of  the  DMSNs.  Such  conical pore  shape  is  particularly  convenient,  for  example  if 
especially large molecules (e.g., DNA or RNA) should have access to the intraparticle mesopore space.
   
In summary, the present work shows that with  the underlying approach,  i.e.,  the employment of 3D 
reconstructions as geometrical models for subsequent diffusion simulations as well as the quantitative 
investigation of morphological features and the resulting transport properties of an individual material, 
most different silica materials can be analyzed by means of various problems:  The investigation of the 
effect of  synthesis parameters on morphological  features,  the  influence of  the pore arrangement and 
geometries on the efficiency or performance of a material in chromatography or catalysis, for instance, 
or the radial investigation of the intraparticle pore morphology, to name just a few examples. It is shown 
that by means of simplified geometric models (e.g., cylindrical or spherical pore models) in most cases 
the very individual pore structure of mesoporous silica cannot be adequately described. In this work an 
approach  is  presented,  which  enables  the  quantification  of  morphology  and  transport  properties  for 
individual materials in a simple way, so that in the next step highly accurate general expressions can be 
formulated for different classes of silica materials. 
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III. Introduction 

Retention and transport properties play a role in every system, where analytes or guest molecules need 
to be moved within or through a confined environment in a certain time to get highest efficiency, best 
uptake, controlled release, or highest size­selectivity. However, retention and transport properties are 
directly  connected  to  the  morphological  features  of  an  individual  material.  For  instance,  in  size­
exclusion­chromatography  (SEC)  or  in  microreactor  systems  used  in  catalysis,  a  broad  pore  size 
distribution  (PSD)  of  the  packing  material  leads  to  a  low  size­selectivity  of  the  fixed  bed,  which 
significantly reduces the performance of the experiment, while in the meantime a narrower PSD and 
therefore higher selectivity can lead to a much slower transport through the porous structure due to the 
higher spatial confinement and can, in consequence, cause a degradation of the experimental efficiency. 
After all, the transport characteristics of any material need to match the issue of interest, and the simplest 
way to control transport properties is the adjustment of the material`s morphology and to gain detailed 
knowledge  about  its  morphological  features.1  Under  this  aspect,  the  exact  three­dimensional  (3D) 
architecture needs to be known to draw conclusions about the effect of different synthesis routes on the 
morphology of  the material on one hand, and  to  relate morphological  features  to observed  transport 
properties of the structure on the other hand. Since diffusion takes place within and through pores and 
void spaces of all sizes, only a multiscale view on the porous structure enables a comprehensive analysis 
of  morphology­transport  relationships.  An  overview  over  the  most  important  aspects  of  the  here 
presented  reconstruction­simulation  approach,  focusing  on  the  mesopore­scale  analysis  of  various 
mesoporous silica, is given in Fig. 1.  

 

Fig. 1: Overview of the reconstruction­simulation approach using the example of SBA­15 and KIT­6, two ordered 
mesoporous silica.  

From the reconstruction (Fig. 1, left) most important morphological features can be uncovered by eye, 
for example plugs, constrictions,  intrawall mesopores, or  the basic arrangement of  the primary pore 
system, all being important factors that define the transport properties of the given material. By means 
of  random­walk  particle­tracking  (RWPT)  simulations  of  hindered  diffusion  within  the  mesopore 
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system of the reconstructions, expressions for the global and local hindrance factors can be derived as 
functions of λ (Fig.1, right). More detailed morphological analyses are possible by means of medial axis 
analysis, for example, where the pore connectivity is quantified by reducing the pore system to a pore 
skeleton, showing the accessible pore volume for tracers of different sizes (insets in Fig.1, right, showing 
the  pore  skeleton  in  red).  The  amount  and  the  connectivity  of  the  remaining  accessible  pore  space 
directly influences the hindrance factor for a given tracer size.   
In this work, the main focus lies on the detailed characterization of the intraparticle mesopore space of 
different silica materials (cf. section III.2) based on electron tomographic (ET) data (cf. section III.1). 
The  3D  reconstructions  obtained  from  ET  are  then  employed  as  geometric  models  for  numeric 
simulations of hindered diffusion within the mesopore space of the reconstructions based on the RWPT­
technique (cf. section III.3). From the hereby obtained long­time limit of the time­dependent diffusion 
coefficient the global and the local diffusion coefficients can be determined for an individual material 
and,  together with  the previously gained knowledge about morphological details of  the  investigated 
material, the relationships between morphological and transport properties can be quantified (cf. section 
III.4).  
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III.1 Electron tomography 

Electron  tomography  (ET)  enables  the  reconstruction  of  structures  in  three  dimensions  (3D)  with  a 
resolution  in  the  nanometer  range.2  Especially  for  investigations  involving  porous  materials,  this 
imaging  technique  has  become  a  standard  characterization  method.1  In  electron  tomography,  two 
dimensional  (2D)  projection  images  of  the  specimen are  acquired  at  different  tilt­angles  (usually  in 
increments of 2°).2,3 The tilt­series then needs to be aligned (by means of fiducial markers, e.g. Au), 
before the 3D structure can be derived from the stack of projections using a reconstruction algorithm 
that  combines  the  information  from  the  2D  projection  images.4  In  this  work,  scanning  transmission 
electron microscopy (STEM) is used as imaging mode for the acquisition of the 2D images, however, 
tomography  is  also  possible  with  other  imaging  methods  as  for  example  bright­field  transmission 
electron microscopy (BF TEM)5 or annular dark­field TEM (ADF TEM).6 The principle of STEM is 
illustrated in Fig. 2.  

 

Fig. 2: Schematic illustration of the principle of ET. Adapted from [7]. 

In STEM, the probe is scanned across the sample (as in scanning electron microscopy, SEM) and from 
the  transmitted  beam  the  electron  signals  can  be  detected  in  several  ways,  providing  images  with 
complementary characteristics. With an high angle annular dark field (HAADF) detector, Rutherford 
scattering from the atomic nuclei is collected, and the obtained image is sensitive to the atomic number 
Z (also called Z­contrast image).8 In comparison to other imaging modes like bright­field (BF) or annular 
bright­field (ABF), HAADF STEM, which is employed in this work, renders incoherent images, which 
are less sensitive to sample thickness, crystal tilts, and residual aberrations, and which reproduce the 
positions of  high­Z elements with high accuracy.8    
Due to the geometry of typical sample holders used for electron tomography, the specimen can often be 
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rotated within an angular range of ±80° instead of covering the complete ±90° range. This leads to a 
lack of information, the “missing wedge”, which is the primary reason for a loss of resolution in the 

resulting 3D reconstruction since the resolution highly depends on the number of acquired projections.3,9 
Another  factor  limiting  the  accuracy  of  the  3D  reconstruction  is  the  mathematical  reconstruction 
algorithm that is employed to obtain a 3D model from a stack of 2D projections. One of the most popular 
reconstruction algorithms, which is also used in this work, is the simultaneous iterative reconstruction 
technique  (SIRT).10  In  this  algorithm,  an  initial  reconstruction  is obtained  from  the  tilt­series of  the 
measured  projection  images  by  a  simple  back­projection  as  a  first  step.  Afterwards,  the  initial 
reconstruction  is re­projected along the same angles as used for  the acquisition of the measured  tilt­
series (forward projection). The difference between the re­projections and the measured projections is 
then calculated (projection error). In the last step, the simultaneously back­projected projection error is 
added to the previous reconstruction to refine the reconstructed object.10 This procedure is repeated until 
a certain stop criterion is reached, which is often defined as the number of iterations.    
Another  iterative algorithm, which is especially convenient  for specimens that consist of only a few 
materials  corresponding  to  only  a  small  number  of  distinct  grey  levels  in  the  reconstruction,  is  the 
discrete  algebraic  reconstruction  technique  (DART).  This  algorithm  starts  from  an  iterative 
reconstruction technique (such as SIRT) and uses the existing information on grey levels from the SIRT 
reconstruction  to  select  a  convenient  threshold.  This  threshold  is  then  applied  to  segment  the 
reconstruction. In the segmented reconstruction, all boundary pixels (i.e., pixels that have at least one 
neighboring pixel of another segmentation class) are located, and all non­boundary (i.e., all remaining) 
pixels  are  simultaneously  assigned  to  either  the  background  or  objects  grey  levels.  Then,  the  SIRT 
algorithm  is  applied  again,  but  only  the  boundary  pixels  are  updated.  With  this  technique,  the 
computational effort is significantly reduced due to the reduced number of unknown pixels.10   
In Fig. 3, the basic reconstruction procedure is shown using the example of a dendritic mesoporous silica 
nanoparticle which has been investigated in this work.  

 

Fig. 3: Different steps during reconstruction procedure. From the initial tilt series (A) a first reconstruction (B) is 
obtained by applying the SIRT algorithm, consisting of a stack of greyscale 2D images. With the DART algorithm, 
the number of grey  levels  is significantly reduced (C) and  the final binarized reconstruction (D) contains only 
black (void) and white (solid) pixels. 

The reconstruction starts from the aligned tilt series (A) (the bright dots are gold­nanoparticles used as 
reference  markers)  and  after  application  of  the  SIRT  algorithm,  a  first  greyscale  reconstruction  is 
obtained  (B),  consisting  of  a  stack  of  2D­images.  By  means  of  the  DART  algorithm,  a  new 
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reconstruction is obtained (C), consisting of only a few grey levels that correspond to different material 
classes within the reconstruction. The final binarized reconstruction is shown in (D), containing only 
black (void) and white (solid) pixels. 

III.2 Morphological analyses of mesoporous structures 

In the following sections, different methods for the analyzation of the mesopore morphology as used in 
this work are presented. For all methods, 3D reconstructions obtained from ET serve as geometrical 
models,  enabling  an  accurate  quantitative  investigation  of  the  pore  morphology  of  any  individual 
mesoporous silica on the nanoscale. 

Chord length distribution analysis 

With  chord  length  distributions  (CLDs)  it  is  possible  to  investigate  and  compare  systems  with 
fundamentally different morphology, since no pre­assumptions about the shape or size of the considered 
phase are needed for this analysis method.1,11,12 Especially for disordered or weakly ordered systems, 
where  it  is not possible  to describe  the phase morphology by a well­defined macro­lattice,  the CLD 
analysis is a convenient method for the morphology characterization.13 As there are no limitations to the 
physical  characteristics  of  the  phase  itself  (solid,  liquid,  or  void),  CLD  analysis  is  applicable  to  all 
multiphase media.1   
To derive a CLD within a given material (consisting of at least two phases), first an adequate number of 
seed  particles  (a  too  small  number  would  not  render  reliable  results  and  too  many  enhance  the 
computational effort) is inserted at random positions within the phase of interest (here, the pore volume). 
Then, vectors are spread  from the seed point  in equiangular directions until  they  reach  the  interface 
border. Afterwards,  the  chord  lengths  are  calculated by  adding  the  lengths of  two opposing vectors 
(representing the linear distance between two interphases) and stored, before they are collected within a 
histogram.1  Chords  are  discarded,  if  one  of  the  opposing  vectors  crosses  the  image  border.14  The 
resulting histogram gives information about both the local pore geometries, as most of the vectors scan 
a  single pore, and  the short  range geometry,  if vectors probe a neighboring pore without hitting  the 
interface  border  before.1  From  the  CLD,  qualitative  and  quantitative  information  about  the  pore 
geometry and structural disorder within the considered system is available. Qualitatively, conclusions 
about the nature of structural disorder can be made from the shape of the distribution. Basically, three 
different types of disorder are known: Crystalline structures with a high structural order result in multi­
modal  CLDs,  structures  defined  by  Debye  randomness  follow  an  exponential  decay  function,  and 
structures with correlated disorder can be well described with the k­Γ­function by1,12,15 

𝑓(𝑙c) =
𝑘𝑘

𝛤(𝑘)

𝑙𝑐
𝑘−1

𝜇𝑘 exp (−𝑘
𝑙c

𝜇
)                 (1) 

with lc the chord length, Γ the gamma function, μ the mean chord­length as first­moment parameter and 
k  a  second­moment  parameter  defined  by  the  mean  chord  length  and  the  standard  deviation  with 
k = μ2/σ2. In Fig. 4 the histogram of the CLD within the void space (light blue) of the random mesoporous 
silica Si60 and the resulting k­Γ­fit is shown as an example. 
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Fig. 4: Slice through the reconstruction of Si60 with void in black and silica walls in white (A). Two seedpoints 
located at random positions within the void volume are shown as an example (blue). The resulting CLD from a 
high  number  of  seedpoints  is  shown  in  (B)  in  light  blue.  The  k­Γ­fit  (eqn.  1)  gives  a  mean  chord  length  of 
μ = 12.9 nm and a second moment parameter k = 2.7. 

Correlated  disorder  is  found  for  randomly  packed  beds,  random  mesoporous  silica  or  monolithic 
materials, for example, and therefore of highest interest for the present work.16­19 In eqn. 1, the second­
moment  parameter  k  is  a  measure  of  the  homogeneity  found  for  the  phase  of  interest.  The  more 
homogeneous the structure, the larger the k­value, i.e., the narrower the CLD with respect to μ.14 The 
mean chord length μ in eqn. 1 measures the average dimensions of the considered phase, for example 
the void volume. In this work, CLD analysis is used, amongst others, to compare the mean pore size 
dmeso obtained from nitrogen physisorption measurements by using a cylindrical pore model with the 
mean chord length μ from CLD analysis, which represents the mean pore size as well, but is not based 
on  any  pore  geometry  model.19  This  has  shown  the  drastic  advantage  of  CLD  analysis  over  other 
geometrical analysis methods that underly specific pre­assumptions about the pore geometry, especially 
when it comes to complex pore structures as found for dendritic mesoporous silica nanoparticles, for 
example.19 

Medial axis analysis 

With medial axis  analysis  (MAA),  the phase of  interest within  a material  consisting of at  least  two 
distinct  phases  can  be  investigated  by  means  of  its  topology  and  pore  connectivity.  For  MAA,  an 
iterative  thinning  algorithm  (available  as  an  ImageJ  plug­in bundle; “Skeletonize3D” and 

“AnalyzeSkeleton”)20 is applied to the reconstruction to reduce the considered phase to a skeleton of 
one  voxel  thickness  while preserving  its  geometrical  and  topological  properties.14  Then,  an  average 
number of branches meeting in one junction within the skeleton is calculated from the number of third, 
fourth and higher order junctions (representing junctions, where three, four, or five and more branches, 
respectively, meet at one junction) by 

𝑍 = 3
𝑛t

𝑛j
+ 4

𝑛q

𝑛j
+ 5

𝑛x

𝑛j
                   (2) 
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with Z the average connectivity, and nt, nq and nx the number of nodes connecting three, four or more 
than four branches, respectively. The quotients nt/nj, nq/nj, and nx/nj represent the relative amount of each 
junction type amongst all junctions within the skeleton network with14 

𝑛x

𝑛j
= 1 −

𝑛t

𝑛j
−

𝑛q

𝑛j
                   (3) 

MAA is an important tool to quantify and explain transport phenomena within mesoporous materials, 
where pore interconnectivity is a central aspect limiting the effectiveness and selectivity of molecular 
transport. Especially when  it  comes  to  finite­size  tracers moving  through confined pore volume,  the 
accessible pore space varies significantly in dependence of the tracer size, as visualized in Fig. 5 using 
the example of SBA­15. 

 

Fig. 5: Principle of MAA shown for SBA­15 for two different tracer sizes. The left picture in (A) is a slice through 
the  reconstruction  with  accessible  void  volume  in  black  (here,  for  pointtracers)  and  solid  in  white.  The 
corresponding pore skeleton is shown on the right in red. By means of a dilatation procedure of the solid volume 
(B), the void space is decreased, reproducing the accessible void volume for larger tracers that experience enhanced 
spatial confinement. Here, the accessible void volume and corresponding skeleton (in red) is shown in (B) for a 
tracer with a diameter of 4.70 nm. 

Clearly, the skeleton of the available pore volume for pointsize tracers (Fig. 5A) differs highly from the 
skeleton left for a tracer of 4.70 nm diameter (Fig. 5B) due to enhanced steric hindrance, which causes 
tracers to be excluded from a certain amount of pore space. Especially constrictions and bottlenecks 
within the pore network lead to an exclusion of larger tracers from whole branches of pores. In general, 
MAA  is a  simple method  to quantify  the  interconnectivity of  the pore network and  to visualize  the 
accessible pore space for tracers of different sizes, highlighting  critical points within the mesoporous 
structure, where bottleneck and constrictions cause branches to be cut off for tracers of critical sizes.18  

Nitrogen physisorption analysis 

Since  the  adsorption  behavior  of  porous  materials  allows  for  conclusions  about  basic  structural 
characteristics  of  the  considered  material,  physisorption  analysis  is  a  convenient  tool  to  make  pre­
assumptions about  the overall use of a material  for many applications, such as  their performance as 
packing materials in chromatography.22 Especially for the investigation of nano­/mesoporous structures, 
the usage of gases for adsorption/desorption analyses is advantageous, since it enables the accessibility 
of a broad range of pore sizes, including micropores (up to 2 nm diameter) and mesopores (2 – 50 nm 
diameter). In this work, nitrogen is used as adsorptive, being a standard adsorptive for both micro­ and 
mesopore analysis.22   
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With IUPAC classification of physisorption isotherms (Fig. 6), it is possible to interpret physisorption 
isotherms in order to draw conclusions about basic structural characteristics of the considered material.  

 

Fig. 6: IUPAC classification of physisorption isotherms. Adapted from [22]. 

The isotherms are classified in different types, with microporous materials leading to type I isotherms, 
non­porous  or  microporous  materials  leading  to  type  II  isotherms,  structures  with  weak  adsorbent­
adsorbate  interactions  (i.e.,  no  multilayer  formation  of  gas  molecules)  yielding  type  III  and  type  V 
isotherms,  mesoporous  materials  leading  to  type  IV  isotherms,  and  structures  where  layer­by­layer 
adsorption  occurs  on  a  non­porous  surface  leading  to  type  VI  isotherms.22,23  From  the  shape  of  the 
isotherm hysteresis, more detailed information can be gained about the pore geometry (i.e., the existence 
of  bottleneck­structures  and  constrictions)  and  the  pore  network  connectivity  (for  example  the 
occurrence of intrawall­pores).17,18,22 Hysteresis loops are also classified by IUPAC into different types, 
each reflecting a special pore geometry.23   
The classification of different surface and pore structures into different types of isotherms and hysteresis 
loops  is  possible,  since  all  structures  underly  the  same  adsorption  mechanism  principle  so  that  the 
isotherms directly reflect  the  structural characteristics of  the considered material.22 For example,  the 
filling process of mesopores is fundamentally different to the continuous adsorption­desorption process 
within  micropores  (Fig.  6,  type  I)  due  to  a  multilayer  formation  process  followed  by  capillary 
condensation. This leads to type IV(a) or Type IV(b) isotherms, depending on the mesopore size. In 
larger mesopores, capillary condensation is delayed on the adsorption branch due to the existence of 
metastable adsorption fluid, while the desorption process takes place at the equilibrium state for liquid­
vapor transition, leading to hysteresis.24 This is often found for ordered mesoporous materials such as 
MCM­41 or SBA­15 materials, where the pore network consists of uniform cylindrical pores. For ink­
bottle­shaped pores, the desorption process is different to desorption from simple cylindrical pores and 
includes pore­bocking effects, leading to an either steeper or flatter desorption branch with respect to 
the adsorption branch, depending on the neck diameter relative to the pore cavity diameter.22 Another 
case of desorption occurs, when the neck diameter is smaller than a critical size at a certain temperature, 
leading to desorption via cavitation, where a spontaneous nucleation of a bubble in the pore causes the 
pore body to empty but the neck remains filled.22   
Besides qualitative analysis of physisorption isotherms, quantitative information about the considered 
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mesoporous material is available as well. This includes the Brunauer­Emmet­Teller­(BET­)surface area 
and the pore size as most important quantities. The pore size distribution (PSD) can be derived from the 
nitrogen physisorption isotherm from either the adsorption or the desorption branch by means of non­
local density functional theory (NLDFT).22 By using an appropriate DFT kernel, an accurate PSD of a 
mesoporous material  can be derived,  if  the pore geometry of  the DFT model matches  the  real pore 
geometry of  the given material. As mentioned  in  chapter  4,  the mean mesopore  size obtained  from 
nitrogen  physisorption  measurements  can  be  quite  different  to  the  mean  pore  size  derived  by  CLD 
analysis, if the pore geometry is more complex and cannot be described by a simple cylindrical pore 
model.19 

III.3 Simulation of tracer diffusion 

To derive  time­dependent diffusion coefficients  for  tracer diffusion  through  the pore volume within 
cubical  subsections  of  the  reconstructions,  a  random­walk  particle­tracking  (RWPT)  method  was 
employed in this work.25­27   
The method is based on the equivalence of the diffusion equation 

𝜕𝑐(𝒓,𝑡)

𝜕𝑡
= 𝐷∇2𝑐(𝐫, 𝑡)                   (4) 

where D is the diffusion coefficient and c is the concentration, and the stochastic differential equation 
that describes the random walk of a tracer, which is written in its discrete form as28 

𝐫(𝑡 + δ𝑡) = 𝐫(𝑡) + ∆𝐫                  (5) 

In  eqn.  5,  r(t)  is  the  tracer  position  at  time  t,  Δr  is  the  displacement  of  every  tracer  and  δt  is  the 
elementary time step of the random walk, which is chosen such that the displacement of each tracer does 
not exceed Δh/10 (Δh as the voxel size of the reconstruction). The RWPT method uses a large number 
(typically 106 to 107) of pointlike, inert tracers, which are randomly inserted into the void volume of the 
considered reconstruction that serves as geometrical model for the investigated porous material.27 Then 
the tracers are displaced to random new positions within the void space and the displacement of each 
tracer is calculated after every time step δt by 

∆𝐫 = 𝜸√6𝐷mδ𝑡                   (6) 

where γ is a vector with random orientation in space and length (space and length following a Gaussian 
distribution  with  zero  mean  and  unity  standard  deviation)  and  Dm  is  the  molecular  tracer  diffusion 
coefficient in free space.27,29 The tracer positions are monitored after every time step (according to eqn. 
5) so that the time­dependent diffusion coefficient can be derived by30  

𝐷(𝑡) =
1

6𝑁

d

d𝑡
∑ [∆𝐫𝑖(𝑡)]2𝑁

𝑖=1                  (7) 

where D(t)  is  the  time­dependent diffusion coefficient  (average  in all Cartesian directions),  N  is  the 
number of tracers, and Δri(t) = ri(t) – ri(t = 0) is the displacement of the ith tracer after time t. The value 
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of  the normalized  time­dependent diffusion coefficient D(t)/Dm  starts  from unity and decreases with 
time (that is, the number of iterations) due to steric, passive interactions of the tracers with the solid 
phase until it reaches a long­time, asymptotic limit. The long­time limit then corresponds to the time­
independent effective intraparticle diffusion coefficient, Deff, that represents the geometric confinement 
experienced  by  the  whole  tracer  ensemble  during  the  random  walk.29  The  RWPT  method  uses  a 
multiple­rejection boundary condition at the solid­void interface to restrict the tracer movement to the 
void volume of the reconstruction. That is, the displacement of a tracer into solid phase is rejected and 
recalculated, until the new position is inside the void volume. To avoid displacements of tracers outside 
the  reconstruction,  a  mirror  boundary  condition  was  implemented  at  the  external  faces  of  the 
reconstructed  section  such  that  each  tracer  that  crosses  the  external  boundary  of  the  reconstruction 
continues its movement within the mirror image of the original domain.29     
The principle of the diffusion simulations as conducted in this work is illustrated in Fig. 7. From the 
reconstructions (Fig. 7A), the time­dependent diffusion coefficient is obtained for different tracer sizes 
(Fig. 7B), and from the long­time limit of D(t)/Dm the effective diffusion coefficient Deff can be derived 
for each tracer size (Fig. 7C). The dilatation procedure that is used to depict the available pore space for 
finite size tracers (Fig. 7A, bottom) is explained in the next section.  

 

Fig. 7: Visualization of different steps for hindered diffusion simulations. In (A), two reconstructions are shown 
that are employed as geometrical models for simulation of pointtracers (A, top) and finite size tracers (A, bottom) 
moving  through  the pore volume. The  time­dependent diffusion  coefficient  obtained by RWPT simulations  is 
shown in (B) for both tracer sizes, with the effective (long­time limit) diffusion coefficient indicated by a red line 
for both curves. The obtained values for  the effective diffusion coefficient are  then plotted  in (C) for different 
tracer sizes.  

 

Dilatation Procedure 

To investigate hindered diffusion for different tracer sizes (i.e., different λ), a dilatation procedure is 
employed to simulate enhanced steric hindrance instead of directly varying the tracer diameter, which 
would massively increase the computational effort. Both approaches provide equally accurate results.  
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In the dilatation procedure, which is schematically shown in Fig. 8, a solid layer of one tracer radius 
thickness (i.e., one voxel thickness) is added to the existing silica walls within the reconstruction for 
each λ. The tracer center has no longer access to the original solid silica surface due to the enhanced 
steric  hindrance  compared  to  pointtracers.  With  increasing  λ,  the  additional  solid  layers  cause  the 
accessible pore volume to decrease, which simultaneously leads to a decreasing diffusion coefficient. 
Due to the enhanced amount of solid volume and therefore a higher probability of tracers being displaced 
into solid, tracer displacements are more often rejected and recalculated, so that the distance tracers have 
travelled  through  the  reconstruction  after  time  t  decreases  with  increasing  λ.  By  that,  an  increasing 
hindrance of diffusion is simulated for increasing tracer sizes, resulting in a smaller diffusion coefficient 
for higher λ. 

 

Fig. 8: Dilatation procedure illustrated at the example of SBA­15. The void volume is shown in black and solid in 
white within a slice through the reconstructed image stack (A). Then, an additional solid layer is added (red, B), 
reducing the accessible pore volume and mimicking the exclusion of the tracer center from the additional layer 
volume due to enhanced steric hindrance (C). This has similar effect as if the tracer  radius is increased by one 
voxel size.  

 

III.4 Quantitative description of diffusive transport through mesoporous silica 

The diffusion of spherical tracers (solutes) within an unbounded liquid can be adequately described by 
the STOKES­EINSTEIN  equation.  If  diffusion  takes place within pores of molecular dimensions,  e.g., 
within  in  a  column  packed  with  porous,  non­interacting  particles,  both  tortuosity  (windings  of  the 
pathways) and constrictions (narrowing of the pathways) lead to a decrease of the tracer mobility on its 
way through the packed bed and therefore smaller diffusion coefficients compared to movement through 
bulk liquid (i.e., within an open tube).31,32 Thus, the derivation of a diffusive hindrance factor is needed 
to adequately describe hindered transport phenomena.    
With the local diffusive hindrance factor Kd, local fluxes can be specified, quantifying the hindrance of 
bulk diffusion inside the cylindrical pore due to hydrodynamic interactions between solute and the pore 
wall.32,33 The local diffusive hindrance factor is defined as 

𝐾d(𝜆) = 𝐷eff,K(𝜆)/𝐷m                  (8) 

where Deff,K is the long­time asymptotic, effective diffusion coefficient of the spherical particle inside a 
pore, Dm is the free diffusion coefficient of the particle in bulk liquid, and λ is the ratio between pore 



III. Introduction 

23 
 

diameter and particle size.    
At a global point of view, hydrodynamic effects and steric hindrance of particles moving  through a 
porous medium are described by the global diffusive hindrance factor H, which is defined as  

𝐻(𝜆) = 𝐷eff,H(𝜆)/𝐷m                   (9) 

with Deff,H as the long­time asymptotic, effective diffusion coefficient of the spherical particle through 
the porous structure. The global and the local diffusive hindrance factors are related by32,34 

𝐻(𝜆) = 𝜀(𝜆)𝐾d(𝜆)                   (10) 

with ε(λ) as the accessible porosity (the fraction of the cross­sectional pore area that is accessible to the 
particle center), which is (1 ­ λ)2 for cylindrical pores. Some available numerical and analytical results 
for hindrance factors are presented by DECHADILOK an DEEN for cylindrical and slit pores.32 A well­
known expression for cylindrical pores is the RENKIN equation, 

𝐻(𝜆) = 𝜀(𝜆)𝐾d(𝜆) = (1 −  𝜆)2(1 − 2.104𝜆 + 2.09𝜆3 − 0.95𝜆5)        (11) 

The RENKIN equation uses the centerline approximation, which is a common simplification of diffusion 
models and implies the diffusion of spherical particles that are located symmetrically within the pore, 
such that the drag coefficient along the pore axis is equal the constant drag coefficient over the pore 
cross­section.32   
In the literature, Kd(λ) of eqn. 1 is also known as obstruction factor and referred to as  

𝐾d(𝜆) =
𝐷eff,K(𝜆)

𝐷m
=

1

𝜏(𝜆)
                 (12) 

with τ as the mesopore diffusive tortuosity.31,35­37 Some publishers define eqn. 12 using the square root 
of the tortuosity factor, which is simply a matter of definition and should not confuse the reader. The 
tortuosity is a parameter that describes the complexity of a porous structure and helps to compensate for 
differences  between  models  for  chromatographic  applications  and  experimental  data.37  In  models 
describing transfer properties of porous media, the tortuosity is often used as an adjustable parameter.37 
More adequately, the tortuosity should be employed as a function of the accessible porosity, however, 
no model exists so far that is able to represent experimental data over the whole porosity range. This 
problem will be addressed in chapter 2, where the modified WEISSBERG equation38,39 

 𝜏(𝜆) = 1 − 𝑝ln(𝜀(𝜆))                  (13) 

with p as an adjustable parameter as well as a modification of the equation found by ARCHIE39­41 

𝜏(𝜆) = 𝐴(𝜀(𝜆))
(1−𝑚)                   (14) 

with A and m as fitting parameters are compared with data reflecting the tortuosity­porosity relationships 
found by a RWPT simulation approach for ordered silica.  
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Abstract 

We present relationships between the multiscale structure and the separation properties of size exclusion 
chromatography (SEC) columns. Physical bed reconstructions of wall and bulk regions from a 2.1 mm 
i.d. column packed with fully porous 1.7 μm bridged­ethyl hybrid (BEH) particles, obtained by focused 
ion­beam scanning electron microscopy, serve as geometrical models for the packing microstructure in 
wall and central regions of a typical narrow­bore SEC column. In addition, the intraparticle mesopore 
space  morphology  of  the  BEH  particles  is  reconstructed  using  electron  tomography,  to  ultimately 
construct  a  realistic  multiscale  model  of  the  bed  morphology  from  mesopore  level  via  interparticle 
macropore space to transcolumn scale. Complemented by the results of eddy dispersion simulations in 
computer­generated  bulk  packings,  relationships  between  packing  microstructure  and  transchannel, 
short­range interchannel, as well as transcolumn eddy dispersion are used to analyze the fluid dynamics 
in the interparticle macropore space of the model. Further, we simulate hindered diffusion and accessible 
porosity  for  passive,  finite­size  tracers  in  the  intraparticle  mesopore  space,  to  finally  determine  the 
effective particle and bed diffusion coefficients of these tracers in the hierarchical (macro­mesoporous) 
bed. Retention and transport properties of polystyrene standards with hydrodynamic diameters from 5 
to 95 Å in tetrahydrofuran are subsequently predicted without introducing bias from arbitrary models. 
These properties include the elution volumes of the polystyrene standards, the global peak capacity (over 
the entire separation window), and the rate of peak capacity at any fixed elution volume. Optimal flow 
rates yielding maximal global peak capacity and a nearly uniform rate of peak capacity over the entire 
separation window are close to 0.04 and 0.20 mL/min, respectively. SEC column performance obtained 
for fully porous and superficially porous particles is compared by varying the core­to­particle diameter 
ratio ρ from 0 to 0.95. Because the separation window is narrowing more rapidly than the rate of peak 
capacity is growing with increasing ρ, core­shell particles always provide smaller global peak capacity; 
they  still  can  be  advantageous  but  only  for  simple  sample  mixtures.  The  presented  morphology­
performance approach holds great promise for method development in SEC. 
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1.1 Introduction 

Due to the lack of a comprehensive molecular description of a separation along the multiscale journey 
through a chromatographic column (including column length and diameter, individual particles, and the 
mesopore level with a complex interfacial dynamics), its characteristic retention (equilibrium constant, 
phase ratio) and transport properties (axial diffusion, eddy dispersion, and mass transfer resistance in 
the  stationary  phase)  are  best  estimated  from  theoretical  models  and  empirical  expressions  for  the 
relevant model parameters.1­10 For instance, the general rate model of chromatography relies on three 
adjustable parameters: the longitudinal dispersion coefficient DL, the external mass transfer coefficient 
kf, and the apparent diffusion coefficient Dp of the analytes through the stationary phase volume.11 The 
solution for the different moments are unambiguously determined by Laplace transform of the mass 
balance in the time domain.12 The expressions of DL, kf, and Dp are then either estimated from empirical 
correlations or simply extracted by multilinear regression analysis from the best agreement between the 
calculated and experimental peak profiles.11 However, even excellent agreement between experimental 
and  calculated  peak  profiles  does  not  necessarily  validate  the  physical  relevance  of  the  adjustable 
parameter values in the general rate model. Also, the general stochastic theory of band broadening in 
chromatography is based on a simple random­walk model for all non­equilibrium events.7 This general 
theory  of  non­equilibrium  chromatography  relies  on  arbitrary  guesses  regarding  the  interparticle 
macropore space morphology of a particle­packed bed (and the associated interparticle tortuosity factor), 
the intensity of the velocity biases, and unknown parameters such as the apparent diffusion coefficient 
of  the  analytes  in  the  stationary phase.7–9 This  stochastic description of  the  transport properties  in  a 
chromatographic column accounts very well for the essence of the band spreading and is qualitatively 
sound.  Yet,  it  may  easily  miss  specific  morphological  features  of  any  particular  chromatographic 
column.  In  fact,  a  particle­packed  bed  is  comprehensively  described  by  (1)  its  axial  and  radial  bed 
density profiles  (as a  result of slurry­packing at high pressure),13–16  (2)  its external macropore space 
morphology  (three­dimensional  interparticle  void  volume  distribution),17–20  (3)  the  intraparticle 
mesopore  space  morphology,21,22  and  (4)  the  concentration  and  diffusion  profiles  of  the  analyte 
molecules across an individual mesopore.23 This multiscale information on structure and dynamics is 
re­ quired to fully predict the retention and transport properties of any chromatographic column. On one 
hand, no retention and transport model can adequately capture this complex morphological information. 
Molecular  simulations  have  been  performed  to  probe  more  realistic  pore  structures  and  polymer 
exclusion in size­exclusion chromatography (SEC).24,25 The stochastic molecular theory based on the 
method of characteristic functions was also proposed assuming an arbitrary dependence of the partition 
coefficient in SEC on the size of the analytes.26–31 On the other hand, significant improvements have 
been made in recent years regarding physical reconstruction of the interstitial macropore space in real 
chromatographic  (particle­packed  or  monolithic)  beds13,20,32–42  and  of  the  relevant  intraparticle  and 
intraskeleton  mesopore  space.22,43–46  For  example,  focused  ion­beam  scanning  electron  microscopy 
(FIB–SEM)  has  been  applied  to  analyze  the  different  bed  structures  and  their  impact  on  the  flow 
distribution in wall and bulk regions of a 2.1 mm i.d. × 50 mm length narrow­bore analytical column 
slurry­packed with 1.7 μm bridged­ethyl hybrid (BEH) silica particles.47 That work confirmed that the 
wall  region  is  more  densely  packed  than  the  central  bulk  region,  a  result  supported  by  the  stress 
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distribution during bed consolidation.48,49 This information on the large­scale structural heterogeneity of 
packed columns was subsequently supplied in a stochastic model of transcolumn eddy dispersion to help 
re­define  the  actual  transport  properties  of  modern  columns.50  Furthermore,  complemented  by  very 
detailed pore­scale  simulations of  flow and dispersion  in  computer­generated bulk packings,51,52  the 
conventional  stochastic  models  of  short­range  interchannel  eddy  dispersion  in  chromatographic 
columns7 were re­calibrated. This made possible the accurate measurement of  the  transcolumn eddy 
dispersion contributions by a simple difference method.53 In addition to the analysis of the interstitial 
macropore  space  morphology,  electron  tomography  with  sub­nanometer  spatial  resolution  was 
successful in the reconstruction of the internal mesoporous structure of silica particles and monolithic 
skeleton.  Combined  with  pore­scale  diffusion  simulations  of  finite­size  tracers,  these  realistic 
geometrical models allowed to improve quantitative expressions for the pore volume accessible to the 
tracers in dependence of their size and for the associated hindered diffusion coefficient caused by the 
tortuosity,  constriction,  actual  pore­network  connectivity,  and  wall  friction  in  the  mesopores.44,46 
Finally, molecular simulations have provided a molecular­detail picture of the inter­ facial structure and 
dynamics relevant  to  the complex  interactions between stationary phase, mobile phase, and analytes 
underlying the separation mechanisms, e.g., in hydrophilic interaction liquid chromatography54,55 and 
reversed­phase  liquid  chromatography.56–58  Overall,  the  information  gathered  on  the  interfacial 
dynamics at the single­pore level, combined with realistic geometrical models of the mesopore space 
and macropore space morphologies, has recently been used to construct a general multiscale simulation 
approach to the effective bed diffusion coefficient, Dbed in hierarchically porous media.59 The approach 
connects the molecular­level picture of interfacial structure and dynamics, as obtained from molecular 
dynamics  simulations,  with  mass  transport  (diffusion)  in  the  hierarchical  porosity  addressed  by 
Brownian dynamics. This methodology allows to follow subtle changes in the interfacial dynamics, as 
introduced by  the surface functionalization, solute properties, and  the mobile­phase elution strength, 
from the pore­level concentration and diffusivity profiles, through the effective diffusion coefficient in 
the mesopore space, up to Dbed  in  the hierarchical, macro­mesoporous bed. The  importance of using 
realistic bed reconstructions  in  this approach has been shown for a set of modern packing materials 
through  the  impact  of  bed­specific  morphological  features  (related  to  the  particle  size  distribution, 
particle shape, particle oligomers, surface roughness, and debris) on the external bed obstruction factor 
and  longitudinal  dispersion  coefficient;  even  the  most  sophisticated  theoretical  models  must  be 
necessarily  failing  regarding  adequate  interpretation  of  the  transport  behavior,  because  they  do  not 
consider this advanced level of relevant structural details.60,61 In this work, we employ the FIB–SEM­
based bed reconstructions from wall and bulk regions of a 2.1 mm i.d. column packed with 1.7 μm fully 

porous BEH silica particles,47  to be used  in SEC, and a geometrical model of  the  internal mesopore 
structure  of  these  particles,  derived  by  electron  tomography,  for  the  analysis  of  transcolumn, 
interparticle macropore space, and intra­ particle mesopore space morphologies. Retention (exclusion 
volume) and transport properties (molecular diffusivities, flow distribution, hydrodynamic dispersion) 
are evaluated based on an approach that combines the lattice­Boltzmann method (LBM) for the pore­
scale simulation of fluid flow and a random­walk particle­tracking (RWPT) technique for modeling the 
diffusion and hydro­ dynamic dispersion of  finite­size  tracers, mimicking  the  elution of polystyrene 
standards  (5  to  95  Å  in  size)  in  tetrahydrofuran.  The  objectives  of  this  work  are  (1)  to  establish 
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relationships  between  the  morphology  and  retention/transport  properties  of  SEC  columns,  (2)  to 
determine  optimal  experimental  conditions  (e.g.,  flow  rate  at  room  temperature)  that  will  either 
maximize peak capacity or maintain a uniform rate of peak capacity over the entire separation window, 
(3)  to  predict  the  SEC  performance  of  core­shell  particles  with  the  same  mesopore  space  (shell) 
morphology as the fully porous BEH silica particles, and (4) to discuss the benefits of the employed 
reconstruction­simulation approach for the speed­up of method development in SEC. 

1.2 Theory 

In this section, all the presented equations used for the calculation of the zone retention factor and of the 
various reduced plate heights are given without demonstration. For more details, the reader is referred 
to the cited references. Equations are expressed in the most general case (superficially porous particles, 
adsorption  chromatography,  any  analyte  size)  and  are  directly  applied  to  SEC.  The  core­to­particle 
diameter ratio, ρ, is defined by: 

𝜌 =
𝑑core

𝑑p
                     (1) 

where dcore core is the diameter of the nonporous solid core and dp is the mean Sauter diameter of the 
particle batch. dp is given by:  

𝑑p =
∑ 𝑓𝑖𝑑𝑖

3
𝑖

∑ 𝑓𝑖𝑑𝑖
2

𝑖
                     (2) 

where  fi  is  the  number  fraction  of  particles  of  diameter  di  in  the  particle  batch.  Analyte  size  is 
characterized by the variable λ defined as the ratio of the analyte hydrodynamic diameter, danalyte, to the 
mean mesopore size, dmeso (determined experimentally by physisorption measurement and analysis62):  

𝜆 =
𝑑analyte

𝑑meso
                     (3) 

It is assumed that the size of the analytes is always negligible with respect to the size of the interparticle 
macropores, dmacro. Accordingly,  

𝑑analyte  ≪  𝑑macro                   (4) 

1.2.1 Thermodynamics 

The definition of the zone retention factor, k1, is based on an arbitrary choice for the delimitation between 
stationary and mobile phase volumes. By convention, the mobile phase volume (Vm) is the interparticle 
volume while the stationary phase volume is the volume of the packed particles (Vs). The volumes of 
the mobile and stationary phases are then written:  

𝑉m = 𝜀e𝑉c                     (5) 

𝑉s = (1 − 𝜀e)𝑉c                   (6) 
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where εe is the external (interparticle) porosity of the column and Vc = πrc
2L is the volume of the empty 

column tube (L is the length and rc the inner radius of the column). The average analyte concentration, 
cbed, in the column is then written: 

𝑐bed = 𝜀e𝑐m + (1 − 𝜀e)𝑐s                 (7) 

where cm is the analyte concentration in the mobile phase and cs is the average analyte concentration in 
the stationary phase. cs is then written:  

𝑐s = (1 − 𝜌3)[𝜀p(𝜆) + [1 − 𝜀p(𝜆)]𝐾α]𝑐m             (8) 

where εp(λ) is the fraction of the pore volume in the particle accessible to the analyte (characterized by 
its dimensionless size λ) and Kα is the equilibrium constant for the distribution of the analyte between 
solid skeleton and mobile phase. The mesopore volume is the internal volume accessible to pointlike 
tracers (λ = 0). By definition, the zone retention factor can be written as a function of the dimensionless 
parameters ρ and λ:  

𝑘1(𝜆, 𝜌) =
𝑉s𝑐s

𝑉m𝑐m
                   (9) 

 = 1−𝜀e

𝜀e
(𝜀p(𝜆) + [1 − 𝜀p(𝜆)]𝐾α)(1 − 𝜌3)           (10) 

Accordingly, in SEC (no adsorption, Kα = 0) and for core­shell particles (ρ ≠ 0), k1 is written:  

𝑘1(𝜆, 𝜌) =
1−𝜀e

𝜀e
(1 − 𝜌3)𝜀p(𝜆)                (11) 

For fully porous particles (ρ = 0), the expression of the zone retention factor in SEC becomes:  

𝑘1(𝜆) =
1−𝜀e

𝜀e
𝜀p(𝜆)                   (12) 

1.2.2 Multiscale analyte diffusivities 

The  bulk  diffusion  coefficient,  Dm,  of  a  polystyrene  standard  of  mass  M  (in  Dalton)  in  pure 
tetrahydrofuran (mobile phase) and at a temperature of 24 °C is best estimated by:63 

𝐷m(𝑀) = 3.45 × 10−4𝑀−0.564 cm2/s            (13) 

The  corresponding  hydrodynamic  diameter  (in  nm)  is  based  on  the  Stokes–Einstein  relation  for 
diffusion:63 

𝑑analyte(𝑀) = 2.76 × 10−2𝑀0.564 nm            (14) 

The expression of Dm as a function of the dimensionless size parameter λ is then given by:  

𝐷m(𝜆) =
9.52 × 10−6

λ𝑑meso
     cm2/s                (15) 
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where dmeso is expressed in nm. The effective diffusion coefficient of the analytes along the packed bed 
in the absence of convective flow is referred to as Dbed. Dbed results from the multiscale heterogeneous 
structure  of  a  packed  bed.  The  transverse  dispersion  coefficient  along  the  column  diameter  in  the 
presence of convective flow is called Dt. The apparent diffusion coefficient of the analytes through the 
porous particles is referred to as Dp. The expressions of these relevant analyte diffusivities in SEC will 
be given in the next sections. 

1.2.3 Reduced plate heights: general expressions 

In its most general form, the expression of the total reduced plate height (RPH), h, is written: 

ℎ =
𝐵

𝜈
+ ℎeddy,bulk(𝜈) + ℎeddy,wall(𝜈) + 𝐶abs/esc𝜈           (16) 

where ν is the reduced interstitial linear velocity (or Peclet number) given by  

𝜈 =
𝑢𝑑p

𝐷m
                     (17) 

where u is the average interstitial linear velocity over the column cross­section:  

𝑢 =
𝐹ν

𝜀e𝜋𝑟c
2                     (18) 

where  Fν  is  the  applied  volumetric  flow  rate.  In  Eq.  (16),  B  is  the  reduced  longitudinal  diffusion 
coefficient,  heddy,bulk(ν)  is  the  RPH  associated  with  the  short­range  eddy  dispersion  (including  both 
transchannel and short­range interchannel velocity biases operating over a distance up to a few particle 
diameters)  in  the  bulk  region  of  the  bed,  heddy,wall(ν)  is  the  RPH  associated  with  long­range  or 
transcolumn eddy dispersion  (accounting  for  the  transcolumn velocity profile  and  for  the  transverse 
dispersion coefficient of the analyte) along the column diameter, and Cabs/esc is the reduced mass transfer 
coefficient associated with the absorption/escape kinetics of the analyte entering/leaving the spherical 
particles. In this work, the effect of the polymer dispersity on band broadening is not considered. The 
primary  goal  is  essentially  to  determine  the  diffusivities  of  monodisperse  large  molecules 
(oligonucleotides, proteins, monoclonal antibodies and aggregates, etc.) in and through the mesopore 
space  and  macropore  space  morphologies  of  SEC  columns  and  to  evaluate  their  im­  pact  on  SEC 
performance.  

1.2.3.1. Longitudinal diffusion  

The general expression for the B coefficient is:8,64 

𝐵 = 2(1 + 𝑘1)
𝐷bed

𝐷m
                   (19) 

Dbed can be generally obtained by a four­step methodology59 comprising (1) confocal  laser scanning 
microscopy (for capillary columns33,34,65–67) or FIB–SEM (for the 2.1 mm i.d. analytical size column47) 
reconstruction  of  the  external  macroporous  bulk  structure  of  a  column,  (2)  electron  tomographic 
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reconstruction of the internal mesoporous structure of the fully porous particles (ρ = 0),44,46 (3) molecular 
dynamics simulations in a single mesopore to determine the density and diffusion profiles of the analyte 
from the impermeable pore wall to the bulk pore region,57 and (4) construction of a multiscale diffusion 
model to determine the effective macroscopic bed diffusivity Dbed by using a RWPT method in which 
the physical reconstruction of the macropore space (step 1) represents the two­phase structure and the 
effective  diffusivity  in  the  particle  space  (steps 2  and  3)  is  described  by an  effective  homogeneous 
medium  approach.86  For  identical  but  superficially  porous  particles  (ρ  ≠  0),  the  expression  of  the 
coefficient B can be estimated by Torquato’s model of effective diffusion. This model was validated 
both experimentally69 and numerically70,71 and can be expressed as:72–75 

𝐵 =
2

𝜀e
[

1+2(1−𝜀e)𝛽−2𝜀e𝜉2𝛽2

1−(1−𝜀e)𝛽−2𝜀e𝜉2𝛽2 ]                 (20) 

where 𝜉2 is a unique parameter (for all ρ­ and λ­values) accounting for the actual bed specifics including 
particle shape, size distribution, and the presence of debris in the packed bed. It was determined for the 
FIB­SEM­based bulk bed reconstruction as 𝜉2 = 0.2836.60 In Eq. (20), β is a function of the core­to­
particle diameter ratio ρ and of the analyte size parameter λ:8,73–75 

𝛽 =

1−𝜌3

1+
𝜌3

2

Ω(𝜆)−1

1−𝜌3

1+
𝜌3

2

Ω(𝜆)+2
                    (21) 

where Ω is  the  ratio of  the polymer diffusivity  through  the porous  shell  volume  (Dp(λ))  to  the bulk 
diffusion coefficient Dm(λ) which is only a function of λ. By definition of Dp(λ), Ω(λ) is given by  

Ω(𝜆) = 𝜀p(𝜆)
𝐾d(𝜆)

𝜏0
                   (22) 

where Kd(λ) = Dmeso(λ)/Dmeso(λ = 0) is a diffusive hindrance factor characterizing hindered diffusion in 
the  mesopores.  Here,  Dmeso(λ)  is  the  effective  diffusion  coefficient  in  the  mesopores,  whereas  the 
apparent  diffusion  coefficient  of  the  analytes  through  the  particles  is  Dp(λ)  =  εp(λ)Dmeso(λ).  By 
construction, Kd(λ) does not account for the internal tortuosity factor, τ0 = 1.95, for pointlike analytes 
(λ = 0).  Because  the  actually  experienced  tortuosity  factor  intrinsically  depends  on  λ,  its  relative 
variation with changing λ (λ > 0) is embedded in the coefficient Kd(λ). In this work, the expressions for 
εp(λ) and Ω(λ) were determined with finite­size tracers by RWPT simulations in the reconstructed pore 
network of the BEH particles. Accordingly, 

𝜀p(𝜆) = 𝜀0(1 − 2.46038𝜆 + 1.38734𝜆2)             (23) 

where ε0 = 0.49 (accessible porosity for pointlike tracers) and  

𝐾d(𝜆) = 1 − 1.40339𝜆 − 0.10765𝜆2 − 9.02023𝜆3 + 21.75413𝜆4 − 12.06158𝜆5   (24) 

Overall, Ω(λ) is best described by a 7th order polynomial in λ 
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Ω(𝜆) =
𝜀0

𝜏0
(1 − 3.86377𝜆 + 4.73256𝜆2 − 10.70234𝜆3 + 43.79797𝜆4    

    −78.09904𝜆5 + 59.85647𝜆6 − 16.733401𝜆7)         (25) 

1.2.3.2. Short­range eddy dispersion  

The bulk hydrodynamic dispersion data combining advection and diffusion were derived by fluid­flow 
simulations  (LBM  approach)  together  with  a  RWPT  technique.68  The  FIB–SEM­based  bed 
reconstruction data provided a bulk bed porosity of εe = 0.39.47 The RPH related to short­range eddy 
dispersion accounts for transchannel and short­range interchannel velocity biases. A simultaneous flow 
and diffusion exchange mechanism is considered between the involved velocity extremes. Accordingly, 
the RPH is best written as7,8,51,52  

ℎeddy,bulk(𝜈) =  
𝜔1𝜈

1+
𝜔1
2𝜆1

𝜈
+

𝜔2𝜈

1+
𝜔2
2𝜆2

𝜈
               (26) 

where ω1 = 0.00377, λ1  = 0.418, ω2 = 0.160,  and λ2 = 0.300 are  the best parameters obtained  from 
hydrodynamic dispersion simulations in computer­generated random sphere packings with εe = 0.39 and 
a particle polydispersity similar to that of the BEH particles.  

1.2.3.3. Long­range eddy dispersion  

The RPH related to long­range eddy dispersion is derived from the reconstructions of the bed structure 
in  wall  and  bulk  regions  of  the  2.1  mm  i.d.  column.  From  a  stochastic  approach  considering  three 
homogeneous, coaxial packing regions,15,50,76 this RPH is written as 

ℎeddy,bulk(𝜈) =
1

8

(
𝑟c
𝑑p

)
2

𝐷t
𝐷m

𝑢c

𝑢
 
(𝑓ci+

1

2
)𝜆c

2+𝜆c𝜆w+(1−𝑓ci)𝜆w
2

𝑓ci𝜏c+
1

2

𝑢i
𝑢c

𝜏𝑐+(
1

2
−𝑓ci)

𝑢w
𝑢c

𝜏c

𝜈           (27) 

where the transverse dispersion coefficient Dt is obtained from previously reported transverse dispersion 
simulations.77 Dt is then approximated in a range of reduced velocities from 0.5 to 500 by50,76,77 

𝐷t

𝐷m
=

𝐷bed

𝐷m
+

0.146𝜈0.758

1+𝑘1
                  (28) 

In  Eq.  (27),  uc,  ui,  and  uw  are  the  average  interstitial  linear  velocities  in  the  bulk  region,  in  the 
intermediate region (caused by the slurry­packing process at high pressures), and in the wall region of 
the column, respectively. They are determined from the actual velocity profile obtained by combining 
FIB–SEM tomographic data for the three­dimensional reconstruction of the bed in the wall and bulk 
regions  of  the  column  with  LBM­based  simulations  of  the  associated  flow  velocity  distribution.47 
Accordingly, u = 0.999uc, ui = 0.929uc, and uw = 1.416uc.47,50,76 The coefficient fci is the fraction of all 
the molecular transfers from the bulk region to the intermediate region:50 

𝑓ci =
1

2
[

𝜌i
2

𝜌i
2−𝜌w

2 −
1

2 ln(
𝜌𝑖

𝜌w
)
]                 (29) 
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where ρi = ri/rc is the dimensionless radius delimitating the bulk central region from the intermediate 
region and ρw = rw/rc  is  the dimensionless  radius delimitating  the wall  region  from  the  intermediate 
region. According  to  the FIB­SEM/LBM  investigations on  the 2.1 mm  i.d. column packed with  the 
1.7 μm 130 Å BEH particles,47 ri = 917 μm and rw = 1047 μm for rc = 1050 μm. So, ρi = 0.8733 and ρw = 
0.9971. In Eq. (27), λc and λw are the dimensionless length deviations of a single molecule in central and 
wall regions, respectively, with respect to the mean displacement of all molecules: 

𝜆c =
𝑢c−𝑢

𝑢c
𝜏c                     (30) 

𝜆w =
𝑢w−𝑢

𝑢c
𝜏w                    (31) 

where τc, τi, and τw are the dimensionless average residence times in the bulk, intermediate, and wall 
regions, respectively:50  

𝜏c = 𝜌i
2                     (32) 

𝜏i = 𝜌i
2 + 𝜌w

2 +
𝜌i

2−𝜌w
2

ln(
𝜌w
𝜌i

)
                  (33) 

𝜏w =
−𝜌w

4 +4𝜌w
2 −4 ln(𝜌w)−3

1−𝜌w
3                  (34) 

1.2.3.4. Absorption/escape kinetics  

The particle diffusivity, Dp, depends on the particle porosity, the internal tortuosity, the analyte exclusion 
volume, and on the restricted diffusion caused by the mesopore walls. By definition, Dp is written as (cf. 
Eq. (22))  

𝐷p(𝜆) =  Ω(𝜆)𝐷m                   (35) 

The average escape time tesc of the analytes from the porous region of a core­shell particle (assuming all 
possible initial radial positions from the core to the periphery of the spherical particle) is given by:78 

𝑡esc(𝜌, 𝜆) =
𝜀p(𝜆)𝑑p

2

60Ω(𝜆)𝐷m(𝜆)
(1 − 𝜌)2 1+3𝜌+6𝜌2+5𝜌3

1+𝜌+𝜌2            (36) 

The  corresponding  dimensionless  mass  transfer  coefficient,  Cabs/esc(ρ,λ),  associated  with  the 
absorption/escape kinetics of the analytes entering/leaving the core­shell particle is then obtained from 
either the general formalism of Giddings,7 the solution of the mass balance equation of the general rate 
model  in  the  frequency  domain  (Laplace  transform),79  or  from  a  stochastic  model78  confirmed  by 
numerical simulations.80 All four approaches provide the very same expression for Cabs/esc. Accordingly, 

𝐶abs/esc(𝜌, 𝜆) = 2
𝐷m(𝜆)

𝑑p
2

𝑘1(𝜌,𝜆)

[1+𝑘1(𝜌,𝜆)]2 𝑡esc(𝜌, 𝜆)             (37) 
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Eq. (37) is based on the assumption that sojourn times in the interparticle space (average absorption time 
from the interparticle volume to the particle surface) and the sojourn times in the intraparticle space 
(average  release  time from  the  intraparticle volume to  the  interparticle space)  follow an exponential 
distribution.7,80 

1.2.4 Size exclusion chromatography 

The goal of this section is to provide relevant metrics for quantifying the resolution performance of any 
SEC column. Two different metrics of chromatographic resolution will be discussed in this work: the 
global peak capacity and  the rate of peak capacity. They are both defined and explained in the next 
sections.  

1.2.4.1 Global peak capacity  

The global peak capacity is the maximum number of peaks that can be baseline­separated over the full 
separation window in SEC. It is calculated by considering the 4σ volumetric peak width. Under isocratic 
conditions (as in SEC), the volumetric standard deviation, σ(λ), of the analyte concentration profile is 
given by: 

𝜎(𝜌, 𝜆) = 𝐹𝜈

√ℎ(𝜌,𝜆)𝑑p𝐿

𝑢
[1 + 𝑘1(𝜌, 𝜆)]               (38) 

The general definition of the global peak capacity P c is81 

𝑃C = 1 + ∫
d𝑉

4𝜎(𝑉)

𝑉0

𝑉e
                    (39) 

 where Ve and V0 are the interparticle and hold­up column volumes (Ve < V0), respectively, and V is the 
dummy integration variable representing the variable SEC elution volume of a particular polymer. This 
first metric measures the global performance of the SEC column and the users are often interested in 
finding the optimum flow rate for which the global peak capacity is maximum.  

1.2.4.2 Rate of peak capacity  

A second relevant metric of SEC performance is the local increment, (d𝑃C/d𝑉)𝑉R
, of the peak capacity 

per unit of elution volume at any particular elution volume VR. It is called the rate of peak capacity and 
represents the maximum number of peaks that can be separated per unit volume at the particular elution 
volume VR. By definition, it is written as: 

(
d𝑃C

d𝑉
)

𝑉R

= lim
∆𝑉→0

1

∆𝑉
𝑃C(𝑉R, 𝑉R + ∆𝑉)               (40) 

                =
1

4𝜎(𝑉R)
                   (41) 
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This  second metric  is  highly  relevant because  it  is  important  to know  for which particular  range of 
molecular  weights  a  SEC  column  develops  highest  resolution  power  at  any  fixed  flow  rate  and 
temperature. Additionally, the practitioner may be interested in finding the optimum flow rate for which 
the rate of peak capacity remains quasi­uniform over the entire separation window, e.g., regardless of 
the molecular weight of the polymer.  

1.3 Experimental and modeling 

1.3.1 FIB­SEM imaging 

The 2.1 mm i.d. narrow­bore analytical column packed with the 1.7 μm (brand size) fully porous BEH 

particles  was  prepared  on­site  (Waters,  Milford,  MA).  Geometrical  models  of  the  interparticle 
macropore space for wall region and bulk region of the column packed with the BEH particles have 
been  reported  and  analyzed  in  detail  before.47  These  realistic  models  are  based  on  the  physical 
reconstruction of macroscopic pieces of  the  column packing by FIB–SEM tomography. Two  image 
stacks were acquired, one from the wall region and one from the bulk region of the column. The resulting 
reconstructions had the following dimensions: 27.5×27.4×132 μm3  with  44.6×44.6×150  nm3  voxels 
(wall region) and 21.6×21.2×123.5 μm3 with 41.5×41.5×150 nm3 voxels (bulk region). Mean particle 
diameters (external bed porosities) determined from the reconstructions were 2.0 μm (0.37) for the wall 

region and 1.97 μm (0.39) for the bulk region. The flow distribution in the interparticle pore space of 

the reconstructed wall region of the column (cf. Section 1.2.3.3) was already simulated and analyzed in 
the previous paper.47 For the current study, the reconstructed bulk packing has been employed as model 
to  simulate  diffusion  in  the  hierarchically  structured  (macroporous  interparticle  and  mesoporous 
intraparticle) void space of the column in order to determine Dbed (cf. Sections 1.2.2 and 1.2.3.1). 

1.3.2 Scanning transmission electron microscopy 

The realistic geometrical model of the internal mesoporous structure of the BEH particles was obtained 
from  a  crumb  of  the  particles  by  scanning  transmission  electron  microscopy  (STEM).  We  already 
applied this technique to receive physical reconstructions of various mesoporous materials.44,46 For this 
study, crumbs of the BEH particles were prepared by grounding the sample in a mortar before they were 
dusted  over  a  carbon­coated  100×400  mesh  Cu  grid  (Quantifoil  Micro  Tools,  Jena,  Germany).  Au 
fiducial markers of 6.5 nm diameter were subsequently deposited from an aqueous suspension (CMC, 
University Medical Center, Utrecht, The Netherlands) onto the prepared sample. Electron tomography 
was performed using a Fischione 2020 tomography holder on an image­corrected Titan 80­300 TEM 
(FEI, Hillsboro, OR), deployed in STEM mode with an acceleration voltage of 300 kV and a nominal 
beam diameter of 0.27 nm. A high­angle annular dark­field detector was used to collect STEM images 
over a tilt range of ±76° with tilt increments of 2°. The acquired tilt series was aligned with IMOD 4.782 
by means of Au fiducial markers. The three­dimensional physical reconstruction was executed in several 
steps.  First,  50  iterations  of  the  SIRT  algorithm,83  implemented  in  the  Xplore3D  software  package 
version  3.1  (FEI),  were  applied  to  the  aligned  STEM  images.  Subsequent  processing  of  the  three­
dimensional physical reconstruction (denoising and local contrast enhancement) was performed using 
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the  software  Fiji.84  10  iterations  of  the  DART  algorithm85  were  applied  to  the  refined  SIRT 
reconstruction using the Fiji plug­in TomoJ86 prior to binarization of the grey­scale image stack. The 
resulting  reconstruction  had  dimensions  of  220×202×275  nm3  (with  0.643  nm3  voxels)  and  a 
mesoporosity of ε0 = 0.49. 

1.3.3 Nitrogen physisorption 

Nitrogen physisorption measurements were carried out at 77 K using a Thermo Scientific Surfer gas 
adsorption  porosimeter  (Thermo  Fisher  Scientific,  Waltham,  MA)  for  additional  geometrical 
characterization of the BEH particles. The pore size distribution was derived from the desorption branch 
of the nitrogen sorption isotherm by applying the non­local density functional theory model, assuming 
a cylindrical pore geometry.62,87 From the pore size distribution, a pore diameter (mode) of dmeso = 146 
Å could be determined. Specific surface area (106.1 m2/g) and specific pore volume (0.47 cm3/g) were 
calculated from the nitrogen sorption isotherm. 

1.3.4 Medial­axis analysis 

Medial­axis transformation is an efficient approach to characterize geometrical and topological aspects 
of the pore space morphology.88 The medial axis of a volume­filling object can be considered as the 
“skeleton” of the object running along its geometrical middle.89 In this work, medial axis transformation 
was performed by applying an iterative­thinning algorithm to the reconstructed mesopore space using 
the Fiji plug­in Skeleton (Skeletonize and AnalyzeSkeleton).90 In particular, the accessible mesopore 
space  for  each  value  of  λ,  characterizing  the  ratio  between  the  size  of  the  diffusing  tracer  in  the 
simulations (dtracer) and the mesopore diameter (dmeso), was thereby reduced to a medial axis of one voxel 
thickness under conservation of the topological properties of the material. The obtained branch­node 
network  provides  information  about  the  average  number  of  branches  meeting  in  one  junction.  The 
average connectivity is defined as Z = 3 nt/nj + 4 nq/nj + 5 nx/nj (with nx/nj = 1 − nt/nj − nq/nj), where nt is 
the number of triple­point junctions (connecting three branches), nq is the number of quadruple­point 
junctions (connecting four branches), nx is the number of higher­order (> 4) junctions (connecting five 
or more junctions), and nj is the overall number of junctions.91 Thus, nt/nj, nq/nj, and nx/nj are the fractions 
of triple­ or quadruple­point or higher­order junctions with respect to the overall number of junctions in 
the accessible mesopore network for each λ­value. 

1.3.5 Mesopore­scale simulations 

Brownian dynamics simulations of the motion of inert tracers through a BEH particle were performed 
to investigate the accessible mesopore volume, the effective diffusion coefficient, and the pore network 
connectivity as a function of λ, the ratio between tracer size and mesopore diameter. The simulations 
were conducted by using a RWPT technique, which has already been described in previous papers.22,44,46 
Briefly, N = 106 inert, pointlike tracers were uniformly introduced at random positions into the mesopore 
space  of  the  three­dimensional  physical  (STEM­based)  reconstruction  by  assuming  no  interactions 
among the tracers and between tracers and solid surface. At each time step δt, the tracer displacement 
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due  to random diffusive motion was deter­ mined  from a Gaussian distribution with zero mean and 
standard deviation (2Dmδt)1/2 along each Cartesian coordinate. The movement of the tracers is limited 
by the multiple­rejection boundary condition, that is, the displacement of a tracer into solid is rejected 
and is recalculated until the new tracer position is in the mesopore space. The time­dependent diffusion 
coefficient is determined by means of the tracer positions after every time step according to 

𝐷(𝑡) =
1

6𝑁

d

d𝑡
∑ [∆𝑟𝑖(𝑡)]2𝑁

𝑖=1                  (42) 

where Δri(t) = ri(t) − ri(t = 0) is the displacement of the ith tracer after time t. From the long­time limit 
of  the  transient diffusion coefficient D(t)/Dm,  the effective  intraparticle diffusion coefficient Dmeso(λ) 
was  determined  for  each  λ­value  and  subsequently  used  to  calculate  the  corresponding  apparent 
diffusion coefficient as Dp(λ) = εp(λ)Dmeso(λ) (cf. Eqs. (22) and (35)). The diffusion of finite­size tracers 
through the mesopore space was simulated by employing values of λ > 0. Due to their steric interaction 
with the solid surface, the accessible mesopore space for finite­size tracers is always smaller than for 
pointlike tracers. The motion of finite­size tracers was implemented in the simulation approach (staying 
with  pointlike  tracers)  by  systematically  decreasing  the  mesopore  diameter  by a  layer  of  one  voxel 
thickness for each new λ­value, since the accessible pore volume for finite­size tracers is equal to the 
accessible pore volume for pointlike tracers when the pore diameter is decreased by the actual tracer 
size.92 Thus, the tracer size dtracer was formally increased by eroding the solid structure by a structuring 
element of size dtracer. With this approach, the accessible pore space for finite­size tracers was generated, 
covering 7  dtracer­values  from  1.28  to 8.96  nm  in  steps  of  1.28  nm.  In  addition  to  the  simulation  of 
Dmeso(λ), the eroding procedure was applied also to εp(λ), the actual fraction of the pore volume in the 
particle accessible to the tracers characterized by their dimensionless size λ. εp(λ) was extracted  in a 
straightforward manner as the void volume fraction of an eroded pore space at this λ. 

1.3.6 Determination of Dbed 

Effective  diffusion  coefficients  of  the  tracers  along  the  macro­mesoporous  bed  in  the  absence  of 
convective flow, Dbed, were determined with a (reflection barrier) modification of the RWPT method, 
developed  to  simulate  diffusion  in  systems  with  discontinuous  diffusion  coefficients.93–96  With  this 
modification, asymmetric reflectivity is introduced at the interface between two regions (region 1 and 
region 2) characterized by different diffusion coefficients D1 and D2, e.g., interparticle macropore space 
and  intraparticle  mesopore  space.  When  a  tracer  reaches  the  interface  between  the  regions,  the 
probability p12 that the tracer enters region 2 from region 1 is  

𝑝12 = 1 −
𝑐1√𝐷1

𝑐1√𝐷1+𝑐2√𝐷2
                  (43) 

where c1 and c2 are the tracer concentrations in regions 1 and 2, respectively. Then, p11 = 1 − p12 can be 
interpreted as the probability that a tracer reaching the interface during its move in region 1 is specularly 
reflected. Similarly, the probability p21 that a tracer reaching the interface from region 2 will enter region 
1 is 
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𝑝21 = 1 −
𝑐2√𝐷2

𝑐1√𝐷1+𝑐2√𝐷2
                  (44) 

and the tracer is specularly reflected into region 2 with probability p22 = 1 − p21. If a tracer crosses the 
interface from region 1, its move in region 2 has to be recalculated, because it was computed with D1 
instead of D2. For this purpose, the following approach was adapted.96 The initial move Δr of the tracer 
is divided into Δr1 and Δr2 and the associated timestep of the random walk is split using a nonlinear 
time­splitting scheme  

√𝛿𝑡 = √𝛿𝑡1 + √𝛿𝑡2                   (45) 

where  δt1  and  δt2  denote  the  times  spent  in  regions  1  and  2,  respectively.  The  corrected  tracer 
displacement in region 2 is calculated as 

|∆𝑟2,corrected| = √6𝐷2𝛿𝑡 (1 −
|∆𝑟1|

|∆𝑟|
)               (46) 

This approach was implemented to determine Dbed as a function of λ using the corresponding values of 
Dm(λ) for the interparticle macropore space, represented by the FIB­SEM­based reconstruction of the 
bulk packing microstructure (bulk FIB­SEM stack) in the 2.1 mm i.d. column packed with the 1.7 μm 

130  Å  BEH  silica  particles,47  and  Dmeso(λ)  for  the  intraparticle  mesopore  space,  represented  as  an 
effective homogeneous medium.59 While the total number of tracers in the simulations of D bed was the 
same for all λ­values (N = 107), the numbers of the tracers in the interparticle macropore space (Nmacro) 
and in the intraparticle mesopore space (Nmeso) were determined as  

𝑁(𝜆)macro = 𝑁
𝜀e

𝜀e+(1−𝜀e)𝜀p(𝜆)
                 (47) 

𝑁(𝜆)meso = 𝑁
(1−𝜀e)𝜀p(𝜆)

𝜀e+(1−𝜀e)𝜀p(𝜆)
                 (48) 

The time­dependent bed diffusion coefficients Dbed(λ,t) were calculated with an expression analogous to 
Eq. (42) and the effective Dbed­values at a given λ were determined as the long­time asymptotes of the 
time­dependent curves. 

1.3.7 Calculation of flow velocity profiles and longitudinal dispersion coefficients 

The three­dimensional flow velocity field in the interparticle macropore space of the wall stack and in 
bulk  computer­generated  sphere  packings  was  simulated  using  the  LBM.97–99  During  the  past  two 
decades,  the  LBM  has  been  accepted  as  a  powerful  approach  to  fluid  flow  simulations.  Unlike 
conventional computational  fluid dynamics methods based on  the discretization of  the macroscopic, 
continuous Navier­Stokes equation and its numerical solution, the LBM originates from classical kinetic 
theory.  As  a  consequence,  the  LBM  approach  has  several  attractive  features  including  inherent 
parallelism and relative easiness in dealing with complex geometries specific for porous media. This 
simulation method can be applied to simulate the flow velocity field in the macropore space of both 
physically reconstructed (e.g., FIB–SEM­based) and computer­generated porous media. For this work, 
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we adapted the same implementation and parameters of the LBM scheme as reported previously.18,35,51,52 
Key  information  is  just  briefly  repeated  here.  As  before,  we  used  an  LBM  scheme  with  a  single 
relaxation  time and D3Q19­lattice  to  simulate motion of an  incompressible  fluid  in  the  interparticle 
macropore space in  the  laminar flow regime, which is  relevant for chromatographic practice. At the 
solid­liquid  interface,  a  halfway  bounce­back  rule  was  used  to  implement  the  no­slip  flow  velocity 
boundary condition. The body force associated with the hydrostatic pressure was taken as uniform and 
unidirectional over the whole simulation domain. Then, the calculated three­dimensional velocity fields 
u(r)  were  used  to  simulate  advective­diffusive  transport  with  the  RWPT  technique,  where  a  large 
number (N = 106) of passive, pointlike tracers were initially distributed randomly and uniformly in the 
interparticle  void  space  of  the  packing.  During each  time  step  Δt,  the  displacement  of  a  tracer  was 
determined as the sum of advective and diffusive contributions according to  

∆𝑡 = 𝑢(𝑟)∆𝑡 + 𝛽                   (49) 

where β is a vector with random orientation in space and a length governed by a Gaussian distribution 
with zero mean and standard deviation (2Dm)1/2 along each Cartesian coordinate. Passive interaction of 
the  tracers  with  the  packing  particles  was  also  here  handled  through  a  multiple­rejection  boundary 
condition at  the particle surface. After each time step, all  tracer positions were recorded and a time­
dependent longitudinal dispersion coefficient evolved from  

𝐷𝐿(𝑡) =
1

6𝑁

d

d𝑡
∑ [∆𝑥𝑖(𝑡) − 〈∆𝑥(𝑡)〉]2𝑁

𝑖=1              (50) 

where Δxi(t) = xi(t) – xi(0) and <Δx(t)> the displacement in the longitudinal direction of the ith tracer and 
the  average  displacement  of  the  tracer  ensemble  after  time  t,  respectively.  The  time­independent 
dispersion coefficients DL were determined from the asymptote of the monitored transient dispersion 
curve DL(t), observed in the long­time limit. The reduced plate height is related to DL as follows7  

ℎ = 2
𝐷𝐿

𝐷m𝜈
                     (51)  

1.4 Results and discussion 

1.4.1 General morphology of chromatographic columns 

Fig.  1.1  depicts  the  morphology  and  microscopic  structure  of  a  typical  chromatographic  column 
governing the separation properties.59 The illustrations shown in Fig. 1.1 are for a RPLC column (note 
the presence of C18 chains attached to the silica surface and the distribution of the analyte molecules 
between the bulk pore volume and the C18­bonded layer)57,100 and shed light on the retention mechanism.  
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Fig. 1.1: General multiscale morphology of a chromatographic column required to predict its retention and mass 
transport properties. (Top left) Extruded packed bed made of 1.7 μm BEH particles. The diameter of the rod is the 

column inner diameter of 2.1 mm.47 (Bottom left) 40 μm thick slice of the packed bed cut by FIB in the bulk region 

of the column. Individual particles are now clearly distinguishable.47 (Bottom center) SEM picture zooming into 
the interparticle macropore space generated by the packed 1.7 μm BEH particles. The average particle size around 
2 μm and the particle size distribution are now revealed. (Top center) STEM­based geometrical model revealing 
the  internal  mesopore  structure  of  the  porous  130  Å  BEH  particles.  The  average  pore  size  and  the  pore  size 
distribution are now directly observable. (Top right) Density distribution of solvent molecules (green: acetonitrile; 
red: water) from the pore surface (here silica­C18 bonded phase) to the bulk pore center.57 (Bottom right) Density 
distribution and molecular orientation of the retained model analyte propanol on the silica­C18 bonded phase.100 
(For interpretation of the references to color in this figure legend, the reader is referred to the web version of this 
article.) 

SEC is just a particular case where retention is zero. The top left photograph shows a fragment of the 
extruded  cylindrical  bed (2.1 mm diameter) made of the 1.7 μm 130 Å BEH  spherical  particles 
embedded in a resin. Past analyses of the bed microstructure from wall to center have revealed that the 
local bed density was different for thin (40 μm wide, see the bottom left SEM photograph) bed stacks 
extracted from bulk and wall regions of the cylindrical bed.47 This large­scale structural heterogeneity 
is responsible for the so­called transcolumn or long­range eddy dispersion coefficient (Eq. (27)) of the 
analytes.50,76,101 The external macroporous bed  structure  shown  in  the bottom center SEM picture  is 
controlling  the  intensity  of  the  short­range  (a  few  particle  diameters  length  scale)  eddy  dispersion 
coefficient (Eq. (26)).7,18 The top center image was obtained from STEM reconstruction of the internal 
mesoporous structure of a single 1.7 μm 130 Å BEH particle. It reveals the size of the mesopores (the 
average size is 146 Å102), their geometry (irregular, neither spherical nor cylindrical), and the degree of 
interconnectivity between them. The average pore connectivity determined by medial­axis analysis is 
3.21 (cf. Section 1.3.4).102 The internal mesoporous structure is controlling the accessible pore volume 
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for analytes of different size, so, it determines the retention in SEC (Eq. (12)). It also defines the actual 
tortuosity factor τ0 (passive pointlike tracers) and the diffusive hindrance factor Kd experienced by the 
finite­sized  analyte  molecules  as  they  diffuse  through  the  porous  particles.  The  top  right  image 
represents the equilibrium distribution of the sol­ vent molecules (acetonitrile and water) across a 100 Å 
silica­C18  mesopore.  Finally,  the  bottom  right  panel  visualizes  the  molecular  orientation  of  model 
analytes (propanol) in and near the C18 layer at the surface of the pore. This molecular information is 
necessary to determine the retention factor and the effective particle diffusivity. Note that in SEC only 
one solvent is typically used (so, the eluent composition is uniform in the entire pore volume), surface 
diffusion is not relevant (no retention), and the particle diffusivity (Dp(λ)) is simply a function of the 
fraction of the pore volume accessible to the analyte (εp(λ)), the internal tortuosity factor (τ0), and the 
diffusive hindrance factor (Kd). 

1.4.2 Multiscale structure of a SEC column 

1.4.2.1 Transcolumn structure  

Fig. 1.2 shows the simulated radial velocity profile across wall and intermediate regions (about 130 μm 

or 70 dp wide) of a 2.1 mm i.d. column packed with the 1.7 μm 130 Å BEH particles.47 The black circles 
correspond to the data based on FIB­SEM analysis (for elucidation of packing microstructure) and LBM 
simulation (for the flow profile) at discrete radial positions separated by about 0.14 μm. The blue line is 

the best smooth fit to these data points. Three regions are then distinguishable along the diameter of the 
bed: (1) the wide, central, and bulk packing region, where the local eluent velocity is considered uniform 
(uc), (2) the 130 μm thick intermediate region near the column wall where the velocity is continuously 
decreasing from uc to 0.7 uc at a distance of about 3 dp from the wall, and (3) the thin (3 dp wide) wall 
region, where the local velocity oscillates between minima (0.70 uc, 0.75 uc, and 0.80 uc) and maxima 
(0.90 uc, 1.55 uc, and 2.15 uc), and where the particles are more orderly packed because of the presence 
of the impenetrable and quasi­flat (at particle scale) stainless­steel surface.47 The physical explanation 
for this observed radial heterogeneity of the bed density is the radial stress gradient applied across the 
cylindrical bed formed during its consolidation immediately after the particles have been stacked and 
placed in contact with each other.15,48–50,76 Columns are always slurry­packed at very high pressures to 
prepare stable chromatographic beds and to increase the column lifetime in UHPLC. Due to the presence 
of the fixed stainless­steel wall, the stress applied to individual particles in the intermediate region is 
larger than in the central bulk region. On average, the linear velocity in the thin (3 dp) wall region is 1.4 
times larger than the bulk velocity. It is 5% smaller than the bulk velocity in the intermediate region (70 
dp wide). Overall, the average linear velocity across the entire column diameter is equal to 99.9% of the 
bulk  velocity.  This  segmentation  of  the  velocity  profile  into  three  different  zones  of  uniform  flow 
velocity was used to determine the long­range (transcolumn) eddy dispersion RPH at any flow velocity 
from Eq. (27).50  
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Fig. 1.2: Radial distribution of the interstitial linear velocity u(r) observed in a 2.1 mm i.d. column packed with 
1.7 μm 130 Å BEH­C18 particles47 revealing the delimitation between three regions (wall, intermediate, and bulk 
zones).  The  black  circles  are  the  discrete  data  obtained  by  FIB­SEM  tomography  for  analysis  of  packing 
microstructure and by subsequent macropore­scale LBM simulation of the flow profile. The blue line is the best 
smooth fit to the data points. This result allows for the calculation of the long­range eddy dispersion coefficient 
by using a stochastic model.50 (For interpretation of the references to color in this figure legend, the reader is 
referred to the web version of this article.) 

 

1.4.2.2. Interparticle macropore space  

Fig. 1.3 shows a single SEM picture of the bulk stack, obtained by FIB­SEM tomography, from the 
2.1 mm i.d. column packed with the 1.7 μm BEH particles. From the three­dimensional reconstruction 
the column external porosity was then determined as εe = 0.39.  

 

Fig. 1.3: SEM picture of the bulk (interparticle) macroporous structure in a column packed with 1.7 μm 130 Å 
BEH  particles.47  The  anastomosis  of  the  diffusion  and  flow  paths  between  the  packed  particles  is  directly 
observable. 
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These images confirm the existence of a random sphere packing, which accounts for the true particle 
size distribution, the true particle geometry, and the possible presence of large voids as well as debris. 
All these aspects are affecting to some extent the interparticle diffusion coefficient of the analytes in the 
column.60  This  spatial  arrangement  of  the  BEH  particles  (reconstructed  bulk  stack)  was  used  to 
determine the Dbed(λ)­values with the hierarchical diffusion model (Section 1.3.6). 

1.4.2.3. Intraparticle mesopore space  

Fig.  1.4  shows  the  STEM­based  reconstruction  of  the internal volume from a 1.7 μm 130 Å  BEH 
particle.  Its  internal  porosity  has  been  determined  as  ε0  =  0.49.  This  is  the  void­volume  fraction 
accessible to pointlike analytes (or tracers). The three­dimensional model reveals that the geometry of 
the mesopores is neither spherical nor cylindrical, as often idealized in theoretical models derived to 
determine  the  exclusion  volume  and  diffusive  hindrance  factor  for  analytes  of  different  size  in  a 
mesopore network. In fact, for the 130 Å BEH material, the shape of the mesopores is clearly irregular, 
the  pore  size  distribution  is  somewhat  large,  and  many  constrictions  are  observed  for  the  internal 
diffusion paths. Additionally, the average pore connectivity was determined as Z = 3.21 by medial­axis 
analysis (Section 1.3.4), which is typical for similar mesoporous materials.44,46  

 

Fig. 1.4: Model showing the STEM­based physical reconstruction of the internal mesoporous structure of a 1.7 μm 
130 Å BEH particle. The void volume fraction accessible to passive, pointlike tracers is ε0 = 0.49 and the associated 
internal tortuosity factor is τ0 = 1.95. 

Overall, the intraparticle tortuosity factor (passive, pointlike tracers) was simulated as τ0 = 1.95 by the 
RWPT approach. A dilatation procedure of the pore walls was performed to investigate changes in the 
internal pore volume accessible to the analytes as a function of the dimensionless size parameter λ.102 
This approach is equivalent to keeping the same location for the pore walls while increasing the size of 
the analytes.22,92 With increasing size, the analytes are increasingly excluded from the mesopore volume. 
Fig. 1.5 visualizes the SEC exclusion phenomenon for the branch­node network of the reconstruction 
(see  the  red  interconnected  paths),  highlighting  the  fractions  of  the  particle  volume  that  are  still 
accessible to pointlike analytes (49% porosity for λ = 0, left image), to 77.4 Å­sized analytes (5% for 
λ = 0.53, center  image), and  to a nearly excluded 89 Å analyte (1% only for λ = 0.61, right image). 
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Analytes can no longer diffuse through the particle as the analyte size exceeds 92 Å (λ = 0.63), even 
though some of the pores are significantly larger than 92 Å. In the next sections, the transport properties 
of differently sized analytes (0 < λ < 0.61) in the chromatographic bed (effective longitudinal diffusion) 
and in individual 130 Å BEH particles (effective particle diffusivity) are presented and discussed. 

 

Fig. 1.5: Visualization of the fraction of the particle volume accessible to differently sized analytes. (Left) The 
analyte is a pointlike tracer and has access to 49% of the particle volume. (Center) The analyte size is 77.4 Å and 
access still occurs to 5% of the particle volume. (Right) The analyte size is 89 Å and only 1% of the particle volume 
remains accessible. 

 

1.4.3 Transport properties 

1.4.3.1 Effective bed diffusivity  

The  dimensionless  effective  diffusivity,  Dbed(λ)/Dm,  of  differently  sized  polystyrene  standards  (from 
λ = 0 for pointlike tracers to λ = 0.61 for a hydrodynamic diameter of 89 Å) is plotted as a function of λ 
in Fig. 1.6. The external (interparticle macropore space) obstruction factor is determined as γe = 0.675 
for a polystyrene standard that is nearly fully excluded from the intraparticle pore volume and agrees 
with the simulation of diffusion for pointlike tracers exclusively in the interparticle macropore space of 
the reconstructed bed. The corresponding external (interparticle) tortuosity factor is then τe = 1.48. This 
value  is  typical  for  those determined by peak parking experiments using packed beds of nonporous 
particles.103–105  As  expected,  the  dimensionless  effective  diffusivity  is  maximal  (0.74)  for  pointlike 
tracers (λ = 0), because both the accessible particle volume εp and the local diffusive hindrance factor 
Kd are maximal (εp = ε0 = 0.49 and Kd = 1). It is also necessarily smaller than unity because the apparent 
analyte diffusivity in the particle volume is limited by the accessible particle porosity (ε0 = 0.49) and by 
the non­negligible internal tortuosity factor (τ0 = 1.95). As both the accessible particle volume and the 
local hindrance factor start decreasing towards zero (0 < ε0 < 0.49 and 0 < Kd < 1) with increasing λ, 
Dbed(λ)/Dm inevitably starts decreasing. Eventually, Dbed/Dm passes through a minimum around 0.61 for 
λ = 0.38 (55.4 Å polystyrene standard), increases again, and tends towards γe = 0.675 when the accessible 
particle  volume  becomes  significantly  smaller  than  the  interparticle  volume  despite  the  continuing 
decrease of the diffusive hindrance factor Kd. The contribution from internal diffusion to the effective 
bed diffusivity becomes increasingly negligible. 
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Fig. 1.6: Variation of the effective longitudinal diffusion coefficient, normalized by the bulk diffusion coefficient, 
of differently sized analytes along the macro­ mesoporous bed of 1.7 μm 130 Å BEH particles as a function of the 
dimensionless  size  parameter  λ  =  danalyte/dmeso  (dmeso  =  146  Å).  Note  the  external  obstruction  factor  γe =  0.675 
(tortuosity factor 1.48) and the minimum of Dbed(λ)/Dm at 0.61 for a 55.5 Å analyte size. 

The variations of the normalized (by ε0) particle porosity accessible to the differently sized polymers 
and  of  the  diffusive  hindrance  factors  with  increasing  λ  will  be  discussed  in  the  next  section.  The 
expression for the dimensionless effective diffusivity, Dbed(λ)/Dm, of the analytes along the packed bed 
will  be  used  later  to  calculate  RPHs  associated  with  effective  longitudinal  diffusion along  the  SEC 
column (see Eq. (19)) and transverse dispersion across the column (see Eq. (28)).  

1.4.3.2. Intraparticle diffusivity  

The  top  graph  in  Fig.  1.7  shows  the  best  second­order  polynomial  fit  (Eq.  (23))  to  the  normalized 
intraparticle  porosity  data  measured  for  the  1.7 μm 130  Å  BEH  fully  porous  particles  (black  line), 
εp(λ)/ε0, as a function of λ. The data decrease monotonously from unity to ca. 0.01 (cf. Fig. 1.5) as λ 
increases from 0 to 0.61. It is remarkable that the critical size of the polystyrene at exclusion (no more 
percolation through the particle volume above 92 Å analyte size) is much smaller than the average pore 
size (146 Å) based on physisorption analysis. This result highlights the importance of the polydispersity 
of the pore size, in particular, the presence of a large number of pores smaller than 92 Å, and of the 
constrictivity in the pore network. The present analysis suggests that a simple mathematical model of 
retention for size exclusion chromatography based on ideally cylindrical (red line, εp(λ)/ε0 = (1 − λ)2) or 
spherical (green line, εp(λ)/ε0 = (1 − λ)3) mesopores is not acceptable. Accurate prediction of the retention 
volume can only be achieved through physical reconstruction of the actual mesopore space morphology 
of the packing material. 



Chapter 1 

47 
 

 

 

Fig.  1.7:  (Top)  Fraction  of  the  pore  volume  accessible  to  the  analytes  normalized  by  the  particle  porosity  ε0, 
εp(λ)/ε0, as a function of analyte size normalized by the average pore size, λ = danalyte/dmeso (dmeso = 146 Å). Data 
based on the RWPT simulations in the STEM­based reconstruction of the mesoporous space of 1.7 μm 130 Å BEH 
fully porous particles (black line). They are directly compared with predictions considering cylindrical (red line) 
or spherical (green line) mesopores. (Bottom) Plot of the local diffusive hindrance factor, Kd(λ), in 1.7 μm 130 Å 
BEH fully porous particles as a function of λ (black line) and prediction based on the RENKIN semi­empirical 
model for cylindrical mesopores (blue line). (For interpretation of the references to color in this figure legend, the 
reader is referred to the web version of this article.) 

The  bottom  graph  in  Fig.  1.7  shows  the  best  fifth­order  polynomial  fit  (Eq.  (24))  to  the  diffusive 
hindrance factor Kd(λ) measured for the 1.7 μm 130 Å BEH fully porous particles (black line). These 
experimental data are compared to the results of the semi­empirical model of RENKIN 106 (blue line), 
which is expressed as  

𝐾d(𝜆) = 1 − 2.104𝜆 + 2.09𝜆3 − 0.95𝜆5             (52) 
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Interestingly, the RENKIN model (valid for λ ≤ 0.4) appears to be qualitatively acceptable to describe the 
actual diffusive hindrance factor in the 1.7 μm 130 Å BEH fully porous particles. The maximal absolute 
relative errors between the RENKIN model and the experimental data are only 12% for 0 < λ < 0.44 and 
130% for 0.44 < λ < 0.61. Therefore, it is not an excellent quantitative model in SEC since it always 
underestimates or overestimates the true values of Kd(λ) for a wide mass range of polystyrene standards. 
The bottom graph in Fig. 1.7 reveals how the pore space morphology contributes to the internal diffusive 
hindrance over a size range spanned from a pointlike analyte (zero exclusion) to a polymer size beyond 
92 Å (complete exclusion). The expressions for εp(λ) (Eq. (23)) and Ω(λ) (Eq. (25)) obtained from the 
RWPT simulations in the STEM­based reconstruction of 1.7 μm 130 Å BEH fully porous particles will 
then be used to calculate the RPHs related to the kinetics of absorption/release of the polymers by/from 
the porous BEH particles (Eq. (37)). It should be noted that the relationships shown in Fig. 1.7 for the 
accessible porosity and the diffusive hindrance factor are qualitatively, yet not quantitatively applicable 
to other mesoporous materials. Indeed, these relationships depend sensitively on the specific preparation 
protocol  adapted  for  the  particles.  For  instance,  the  curves  in  Fig.  7  for  the  BEH  material  do  not 
necessarily represent the response of the mesoporous shell in superficially porous particles.46 This will 
be discussed in more detail in Section 1.4.4.3. 

1.4.4 Application: SEC separation performance 

Based  on  the  previous  sections,  the  combination  of  morphological  reconstruction  of  the  multiscale 
structure of a SEC column and pore­scale numerical simulation of the transport properties allows  to 
determine thermodynamic (retention) and kinetic (mass transfer) properties of a SEC column packed 
with the 1.7 μm 130 Å BEH fully porous particles. Given the separation problem of the practitioner 
(nature of sample mixture) and associated goals (resolution level and speed), these direct morphology­
transport relationships can be used to develop the most adequate separation method in SEC. For the sake 
of illustration, in the next sections important questions will be answered regarding the optimal SEC flow 
rates that will (1) maximize the global peak capacity (see definition in Section 1.2.4.1) and (2) maintain 
a quasi­uniform rate of peak capacity (see definition in Section 1.2.4.2) over the entire separation range. 
Additionally,  the  advantage/disadvantage  of  core­shell  particles  in  SEC  is  discussed  based  on  the 
variation of the expected uniform rate of peak capacity (and corresponding global peak capacity) when 
changing the core­to­particle diameter ratio from ρ = 0 (fully porous particles) to ρ = 0.95 (pellicular 
particles).  

1.4.4.1. Global peak capacity  

As defined in Section 1.2.4.1, the global peak capacity (Eq. (39)) in this work is the maximum number 
of monodisperse polystyrene standards that can be baseline separated by a 2.1 mm × 150 mm long SEC 
column assumed to be packed with the same 1.7 μm 130 Å BEH particles as characterized in the previous 
sections. Fig. 1.8 is showing the plot of the global peak capacity as a function of the applied flow rate 
which varies from 0.001 to 2 mL/min.  
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Fig. 1.8: Global peak capacity  (given by Eq.  (39)  for a mixture of monodisperse polystyrene  standards)  for a 
2.1 mm i.d. × 150 mm long column packed with the 1.7 μm 130 Å BEH particles as a function of the applied flow 
rate, which varies from 0.001 to 2 mL/min. Note the optimal flow rate at only about 0.04 mL/min. 

Remarkably, the global peak capacity passes through a maximum at a flow rate close to 0.04 mL/min, 
which  is  relatively  low  for  a  2.1  mm  i.d.  UHPLC  column  packed  with  sub­2 μm particles (the 

corresponding optimal flow rate is around 0.4 mL/min for typical small­molecule analytes). Such a low 
optimal flow rate is required to keep the peak width of high­molecular­weight polystyrene standards 
narrow enough (due to their low intraparticle diffusivity Dp), while that of the low­molecular­weight 
polystyrene standards is not drastically increasing (due to their high longitudinal diffusion coefficient 
Dbed). If the flow rate is too low (below 0.04 mL/min), the global peak capacity drops because of the 
excessive longitudinal diffusion of low­mass polystyrene standards. If the flow rate is too large (above 
to 0.04 mL/min), then the global peak capacity soon becomes limited by the slow mass transfer of the 
high­mass polystyrene standards. The optimal flow rate around 0.04 mL/min is the best compromise 
between  these  extreme  situations  as  it  maximizes  the  number  of  baseline­separated  polystyrene 
standards over the entire separation window. This is the first choice for the practitioner if the polymer 
mixture is unknown and time is not a significant development constraint. 

1.4.4.2. Rate of peak capacity  

Alternatively, anticipating that all polymers in the sample mixture will be uniformly distributed from 
near the interparticle volume to near the hold­up volume of the column, then the practitioner may prefer 
to maintain  constant  the SEC  resolution  level  regardless of  the polymer mass while minimizing  the 
possible loss in global peak capacity. This translates into maintaining constant the rate of peak ca­ pacity 
during the SEC run, as defined by Eq. (41). Fig. 1.9 shows the rate of peak capacity for the same 2.1 
mm  i.d.  ×  150  mm  long  SEC  column  as  a  function  of  the  retention  volume,  VR,  corrected  for  the 
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interparticle volume, Ve. Three different flow rates are considered: (1) Fv = 0.04 mL/min for which the 
global peak capacity is maximal around 33. Note that this global peak capacity is measured as the area 
under the black line. It can be seen that the rate of peak capacity continuously decreases from high­ to 
low­molecular­weight polystyrene standards: the resolution level of the lightest polystyene standards is 
definitely not optimal. (2) Fv = 0.22 mL/min for which, remarkably, the rate of peak capacity is nearly 
independent of the polymer mass around 180 cm−3, except for the most excluded polymers which have 
very limited access to the particle volume (it is even higher, up to 260 cm−3, due to the small variations 
of Kd for λ > 0.55 as shown in the bottom graph in Fig. 1.7). Interestingly, the global peak capacity has 
just dropped from 33 to 26 according to the area under the red curve. (3) Fv = 2 mL/min, the highest 
flow rate set for a typical UHPLC column configuration. Except for the highest masses, the rate of peak 
capacity  slightly  increases with  increasing  elution volume  from 80  to  125 cm−3  but  the  global peak 
capacity  markedly  drops  to  only  14.5  due  to  the  slow  absorption/release  kinetics  for  most  of  the 
polymers regardless of their mass.  

 

Fig. 1.9: Rate of peak capacity  (given by Eq.  (41)  for a mixture of monodisperse polystyrene  standards) of a 
2.1 mm i.d. × 150 mm long column packed with the 1.7 μm 130 Å BEH particles as a function of the corrected 

elution volume, VR − Ve. Three different flow rates were applied: 0.04, 0.22, and 2.0 mL/min. Note  the quasi­
constant rate of peak capacity at a flow rate of 0.22 mL/min. 

The advantage of the plots shown in Fig. 1.9 is that the practitioner can easily develop the right SEC 
method  based  on  the  range  of  molecular  weights  that  defines  the  sample  mixture.  It  becomes  very 
convenient to track the evolution of the rate of peak capacity in any given range of elution volume (or 
mass  range)  when  changing  the  flow  rate.  In  the  end,  the  best  compromise  between  the  expected 
resolution achieved and the analysis time has also to be decided by the practitioner. 
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1.4.4.3. Extension to core­shell particles  

In  this  section,  the  analysis of SEC performance of  the  fully porous particles  (FPPs)  is extended  to 
superficially porous particles (SPPs), assuming isomorphological materials with respect to the mesopore 
space in both types of particles. At this point, we seek a simple comparison, which necessarily neglects 
subtle  differences  between  the  two  morphologies  that  can  arise  in  practice  due  to  the  different 
preparation methods.  For  instance,  broader  pore  size distributions  (PSDs)  were observed  for  100  Å 
Kinetex­C18 SPPs than for 100 Å Symmetry­C18 FPPs,107 while the PSD of 100 Å Halo­C18 SPPs has 
been found tighter than that of 100 Å Kinetex­C18 SPPs.108 In another study, comparable widths of the 
PSD  were  measured  for  Luna­C18  FPPs  and  Halo­C18  SPPs.109  Small  angle  X­ray  scattering  and 
physisorption experiments performed on FPPs (Atlantis, Luna, and Zorbax) and SPPs (Halo, Kinetex, 
and Poroshell) with similar average mesopore size (100 ± 20 Å) have revealed similar widths of their 
PSDs.110  Finally,  very  similar  PSDs  were  also  measured  for  90  Å  Symmetry­C18  FPPs  and  100  Å 
CORTECS­C18 SPPs.111 In the end, there is not a clear trend regarding a correlation between the width 
of a PSD and the underlying process used to prepare the mesoporous particles. Therefore, a quantitative 
comparison of the transport properties of the two particle types is not straightforward without actually 
knowing their 3D mesopore space morphologies, as demonstrated by REICH et al..46 There, it was shown 
that the SPPs contained an increased number of narrow and constricted as well as closed pores,  i.e., 
structural  features reflecting compacted and sintered packings  that most  likely form during  the shell 
consolidation step of the particle preparation process. These features determined the diffusive transport 
behavior of the SPPs and marked the key difference to the dynamics in the studied FPPs.46 For the sake 
of simplicity and to assess only the intrinsic impact of a nonporous core present in BEH­based core­
shell particles on column performance, short­range and long­range interchannel eddy dispersion terms 
in the RPH equation were kept identical for both types of particles. The core­to­particle diameter ratio, 
ρ, is then varied from 0 to 0.1, 0.2, 0.3, 0.4, 0.5, 0.6, 0.7, 0.75, 0.8, 0.85, 0.9, and to 0.95. The mesoporous 
structure is that of the BEH FPP and is assumed the same for all SPPs. Different SEC performance will 
therefore  only  originate  from  differences  in  the  absorption/release  kinetics  of  the  polymers 
entering/leaving  the  porous  shell  of  the  particles  and  in  the  effective  longitudinal  diffusion  of  the 
polymer along the column. For each SPP or ρ­value, the operating flow rate is chosen such that the rate 
of peak capacity remains quasi­independent of the polymer mass. All these results are collected in Fig. 
1.10. It is noteworthy that the flow rate at which the rate of peak capacity remains quasi­constant for all 
the polymer masses varies from 0.225 to 0.225, 0.225, 0.230, 0.235, 0.260, 0.285, 0.280, 0.270, 0.250, 
0.200, 0.160, and to 0.110 mL/min. As expected, the rate of peak capacity is continuously increasing 
with  increasing  ρ  because  both  the  average  escape  time  and  the  effective  longitudinal  diffusion 
coefficient of the polymer are decreasing with increasing ρ. The rate of peak capacity increases from 
about  175  cm−3  (ρ  =  0)  to  320  cm−3  (ρ = 0.95).  However,  at  the  same  time,  the  width  of  the  SEC 
separation window is decreasing from about 0.14 cm3 (ρ = 0) to only 0.02 cm3 (ρ = 0.95). Overall, the 
increase in the rate of the peak capacity is overcompensated by the reduction of the separation window 
as the global peak capacity is decreasing from 26.1 to 26.1, 26.0, 25.8, 25.5, 24.4, 23.0, 21.1, 19.6, 17.7, 
15.4, 12.0, and to 7.4 only. 
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Fig. 1.10: Rate of peak capacity (given by Eq. (41) for a mixture of monodisperse polystyrene standards) of a 
2.1 mm i.d. × 150 mm long column packed with hypothetical 1.7 μm 130 Å superficially porous BEH particles 

(ρ = 0, 0.1, 0.2, 0.3, 0.4, 0.5, 0.6, 0.7, 0.75, 0.80, 0.85, 0.90, and to 0.95) as a function of the corrected elution 
volume, VR − Ve. For each ρ­value, the flow rate was selected to maintain quasi­constant the rate of peak capacity 
over the entire separation window. Note the increase of the rate of peak capacity, the decrease of the volume width 
of the separation window, and the decrease of the global peak capacity with increasing ρ. 

To summarize, core­shell SEC particles are definitely advantageous over fully porous SEC particles if 
a very narrow range of molecular weights and a small number of polymers are considered in the sample 
mixture. In contrast, core­shell SEC particles cannot outperform conventional fully porous particles in 
terms of global peak capacity if long­range eddy dispersion is comparable for both types of columns. 
Similar conclusions were reached for wide­pore SPPs.112,113 

1.5 Conclusions 

The physical reconstruction of the multiscale structure of a 2.1 mm i.d. SEC column slurry­packed with 
1.7 μm 130 Å BEH fully porous particles was demonstrated. Transcolumn, interparticle macropore, and 
intraparticle mesopore structures were imaged and digitized by using FIB–SEM and STEM tomography. 
These geometrical models, together with computer­generated packings studied previously, have been 
employed  to  simulate  retention  and  transport  properties  of  a  SEC  column,  without  depending  on 
arbitrary  and  often  erroneous  correlations,  to  express  the  SEC  retention  volume,  effective  bed 
diffusivity, bulk longitudinal dispersion coefficient, the transverse dispersion coefficient, as well as the 
(intraparticle) hindered diffusivities for differently sized analytes. These direct morphology­retention 
and morphology­transport relationships constitute a wealth of opportunities for the separation scientist: 
The demonstration was made that optimal experimental conditions (e.g., flow rate at fixed temperature) 
can be easily predicted in order to maximize the SEC peak capacity or to maintain nearly constant the 
rate of peak capacity over the entire separation range. For specific applications and a narrow range of 
molecular  weights  relative  to  the  full  range  of  molecular  weights  the  SEC  column  can  handle,  the 
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practitioner can simply adjust the average pore size, the temperature, and find out the optimal flow rate 
that will maximize the rate of peak capacity around the average elution volume of the sample mixture. 
This work confirms that the performance of core­shell particles in SEC is unattractive if the practitioner 
is  only  interested  in  the  global  peak capacity.  The  underlying  reason  is  that  the  separation  window 
narrows more rapidly than the rate of peak capacity grows when increasing the core­to­particle diameter 
ratio of  these particles. They could become advantageous  if  the  sample mixture  is not complex and 
contains  just  a  few  high­molecular­weight  com­  pounds.  In  a  forthcoming  investigation,  the  initial 
results of this work will be used to determine the retention and mass transport properties of various SEC 
columns  with  different  lengths  (15  or  25  cm)  packed  with  the  same  BEH  silica  particles  but  with 
different particle size (1.7 and 2.5 μm) and larger nominal pore size (125, 200, and 450 Å). As a first 
approximation, the mesoporous structures of the 125, 200, and 450 Å BEH particles will be considered 
self­similar to that of the 130 Å BEH particles. This will allow to transfer directly the results obtained 
for the 130 ˚A BEH particles to the other BEH particles regarding the pore exclusion volume and the 

restricted diffusion coefficient in the mesopores. The optimization approach presented in this work will 
then  be  used  to  baseline  re­  solve  in  the  shortest  possible  time  real­life  samples  such  as  proteins, 
oligonucleotides, and mixtures of monoclonal antibodies and their aggregated forms. 
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Abstract 

Quantitative morphology–transport  relationships  are derived  for ordered mesoporous  silicas  through 
direct  numerical  simulation  of  hindered  diffusion  in  realistic  geometrical  models  of  the  pore  space 
obtained  from  physical  reconstruction  by  electron  tomography.  We monitor  accessible  porosity  and 
effective diffusion coefficients  resulting  from steric  and hydrodynamic  interactions between passive 
tracers and the pore space confinement as a function of λ = dtracer/dmeso (ratio of tracer diameter to mean 
mesopore diameter) in SBA­15 (dmeso = 9.1 nm) and KIT­6 (dmeso = 10.5 nm) silica samples. For λ = 0, 
the pointlike  tracers reproduce the true diffusive tortuosities. For 0 ≤ λ < 0.5,  the derived hindrance 
factor quantifies the extent to which diffusion of finite­size tracers through the materials  is hindered 
compared with free diffusion in the bulk liquid. The hindrance factor connects the transport properties 
of the ordered silicas to their mesopore space morphologies and enables quantitative comparison with 
random mesoporous silicas. Key feature of the ordered silicas is a narrow, symmetric mesopore size 
distribution (~10% relative standard deviation), which engenders a sharper decline of the accessible­
porosity  window  with  increasing  λ  than  observed  for  random  silicas  with  their  wide,  asymmetric 
mesopore size distributions. As support structures, ordered mesoporous silicas should offer benefits for 
applications where spatial confinement effects and molecular size­selectivity are of prime importance. 
On the other hand, random mesoporous silicas enable higher diffusivities for λ > 0.3, because the larger 
pores carry most of the diffusive flux and keep pathways open when smaller pores have closed off. 

2.1 Introduction 

The growing interest in ordered mesoporous silicas as supports for heterogeneous catalysis, selective 
adsorption, and controlled drug release builds on their morphological properties that govern hindered 
diffusion of finite­size solutes at the mesopore scale.1–10 Ordered mesoporous silicas possess a primary 
pore system with a defined pore shape and size, available in a variety of architectures.11–13 The primary 
pore  system  of  SBA­15  silica,  for  example,  consists  of  hexagonally  arranged,  cylindrical  pores,14 
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whereas  the  primary  pore  system  of  KIT­6  silica  is  a  pair  of  interpenetrating,  bicontinuous,  3D, 
cylindrical  pore  networks.15  A  secondary  system  of  random  pores  (usually  micropores  and  smaller 
mesopores) exists in the amorphous silica walls around the primary pores. Modulation of this intrawall 
porosity  allows  to  tune  the  pore  size  distribution  and  pore  interconnectivity  of ordered mesoporous 
silicas, and consequently, their transport properties as support structures. Closing the intrawall pores 
eliminates connections between the primary pores, which in the case of SBA­15 leads to a pseudo­1D 
pore system, whereas widening the intrawall pores towards the size of the primary pores generates a 
highly interconnected, 3D pore network.16    
A well­defined pore size distribution for the support structure is necessary to achieve high selectivities 
in  a  given  application.  Catalysis  under  confinement  is  a  prime  example  in  this  regard,  as  recently 
highlighted  for  the  ring­closing  metathesis  of  various  dienes,  where  the  catalyst  was  selectively 
immobilized  inside  the primary pores of SBA­15  silica.17 To ensure  that macro(mono)cyclization  is 
favored over oligomerization, the pores should ideally allow unhindered access to the catalyst by one 
substrate molecule, but prevent simultaneous access of  two substrate molecules. A narrow pore size 
distribution is essential, because larger pores quickly increase the risk of oligomerization (decreasing 
selectivity), whereas smaller pores not only ultimately impede the macro(mono)cyclization itself, but 
also  limit  diffusive  transport  into  and  out  of  the  mesopore  network  (decreasing  conversion).  A 
relationship  between  the  size  of  a  substrate  and  its  propensity  to  undergo  macrocyclization  with  a 
catalyst located inside silica mesopores of defined shape and size (cylindrical in the case of SBA­15) 
was  established,  thereby  identifying  a  narrow  operational  window  for  optimal  macrocyclization 
selectivity. This example shows  that knowledge of  the morphology–transport characteristics,  that  is, 
how hindrance  to diffusion depends on  the steric and hydrodynamic  interactions between  finite­size 
substrates and  the  spatial confinement of  a  catalyst  support,  is  instrumental  to unravel  the  interplay 
between  reaction,  confinement,  and  transport  and  thus  to  improve  efficiency,  yield,  and  selectivity.  
The hindrance to diffusion of passive, finite­size tracers locally within and globally into and through a 
mesopore network can be quantitatively expressed by  the corresponding hindrance  factors Kd(λ) and 
H(λ) as a function of λ = dtracer/dmeso, the ratio of the hydrodynamic diameter of the passive tracers (dtracer) 
and the mean mesopore diameter (dmeso):18 

𝐻(𝜆) =
𝐷eff,H(𝜆)

𝐷m
= 𝐾d(𝜆)𝜀(𝜆)                (1) 

Here, Deff,H(λ) is the effective coefficient characterizing the global process of diffusion into and through 
the mesopore network (subscript H) and H(λ) is the associated global hindrance factor relating Deff,H(λ) 
to the coefficient Dm for unhindered diffusion in the bulk liquid. By contrast, Kd(λ) = Deff,K(λ)/Dm is the 
local  hindrance  factor  reflecting  effective  diffusion  locally,  already  within  the  mesopore  network 
(subscript K). ε(λ) in eqn (1) is a function providing the accessible mesoporosity for a given λ. From eqn 
(1) it follows that 

𝐷eff,H(𝜆) = 𝐷eff,K(𝜆)𝜀(𝜆)                (2) 

Furthermore, with pointlike tracers (λ = 0) the expression for H(λ) reduces to 
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𝐻(𝜆 = 0) =
𝐷eff,H(𝜆=0)

𝐷m
=

𝜀0𝐷eff,K(𝜆=0)

𝐷m
=

𝜀0

𝜏0
            (3) 

where ε0 = ε(λ = 0) and τ0 = τ(λ = 0) denote the porosity and tortuosity, respectively, underlying the 
transport of passive, pointlike tracers (subscript 0) into and through the mesopore space. Consequently, 
Kd(λ = 0) = 1/τ0.    
Compared with the extensive literature on zeolites,19–21 data about hindered transport in SBA­15 and 
KIT­6 silicas remain scarce,22–28 and quantitative hindrance factor expressions for these materials are 
entirely lacking. Ideally, this information would be provided together with an explanation of how the 
relevant morphological properties of a material shape the actual hindrance factor expression. We have 
previously begun research into the morphological details relevant to hindered diffusion through ordered 
mesoporous silicas, starting with two SBA­15 and KIT­6 silica samples whose pore size distributions 
contained a sharp peak at dmeso = 9.4 nm with a low tail from ~7 to ~3.5 nm.29 Physical reconstruction 
of the two samples by electron tomography showed a highly interconnected, 3D network of primary and 

smaller secondary mesopores, but also the presence of obstacles in the primary pores (Fig. 2.1). Direct 

(pore-scale) simulations of the diffusion of finite­size tracers in the reconstructions revealed diffusive 
transport to decline sharply with increasing dtracer, coming to rest as dtracer approached one third of dmeso. 
The somewhat unexpected transport dynamics was linked to the narrow constrictions (both samples) 
and dead­ends (SBA­15 only) in the primary pore system. The problem was more severe for SBA­15 
silica,  where  connectivity  relies  entirely  on  the  intrawall  pores  due  to  the  pseudo­1D  primary  pore 
system. The intrawall pores were, however, typically too small to function as bypasses for tracers around 
obstacles in the primary pores (Fig. 2.1). Overall, the reconstruction‒simulation approach demonstrated 

that even a low­amplitude tailing of the pore size distribution to the left, representing a modest amount 
of smaller pores, can have serious consequences for hindered diffusion  through ordered mesoporous 
silicas. 

 

Fig. 2.1: Section from a reconstructed SBA­15 silica sample investigated in REICH et al.,29 highlighting obstacles 
(dead­ends and narrow constrictions) in the primary mesopore system as well as smaller secondary throughpores 
in the silica wall (intrawall mesopores). 

Considering that applications such as heterogeneous catalysis and controlled drug release rely on the 
efficient diffusive  transport of  (often) bulky species through the support structure,  the dominance of 
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minor  morphological  features  over  the  transport  dynamics  of  ordered  mesoporous  silicas  is  highly 
undesirable. In this regard, the results of the initial study suggest the following morphological targets 
for SBA­15 and KIT­6 silicas: (i) opening the narrow constrictions in the primary pores and (ii) widening 
the smaller intrawall pores towards dmeso to provide efficient bypasses around the remaining obstacles 
in the primary pore system. In this work we apply our established reconstruction‒simulation approach 

to  SBA­15  and  KIT­6  silica  samples  that  were  prepared  under  conditions  adapted  to  the  identified 
morphological  targets.  We  investigate  in  how  far  the  morphological  changes  improve  the  diffusive 
transport dynamics of  the ordered mesoporous silicas, particularly compared  to random mesoporous 
silicas. 

2.2 Methodology 

2.2.1 Material preparation 

The SBA­15 silica material was prepared following essentially the synthesis route of CHOI et al.30 8.1 g 
of Pluronic triblock copolymer (P123, EO20PO70EO20, MW = 5800, Sigma–Aldrich) were dissolved in 
8.9  g  of  hydrochloric  acid  (Sigma–Aldrich,  37%)  and  142.5  g  of  distilled  water  up  to  complete 
dissolution. Then, 17.3 g of tetraethoxysilane (TEOS, Fisher Scientific, 98%) were added at once, and 
the mixture was stirred for 24 h at 30 °C. Hydrothermal treatment was carried out at 140 °C for 24 h 
under static conditions. The obtained solid was filtered hot and dried – first at 100 °C for 2 h, then at 
140 °C for 12 h. The template was removed by slurrying the as­synthesized powder in an ethanol–HCl 
mixture prior to filtering and calcination of the product at 550 °C in air.   
The KIT­6 silica material was prepared following an established route.15,31 The synthesis was carried 
out in a closed propylene bottle. 5.1 g of Pluronic P123 were dissolved under intense stirring in 185.3 g 
of distilled water and 9.9 g of hydrochloric acid (37%). After complete dissolution, 5.1 g of 1­butanol 
(Sigma­Aldrich, 99%) were added at once. The mixture was stirred at 35 °C for 1 h prior to addition of 
11 g of TEOS (98%) at once. Further stirring of this starting mixture at 35 °C for 24 h was followed by 
hydrothermal treatment at 140 °C for 48 h under static conditions. The solid product was filtered and 
dried first for 2 h at 100 °C prior to further drying for 12 h at 140 °C. For template removal, the as­
synthesized powder was slurried in an ethanol–HCl mixture for a short time. The obtained product was 
filtered and then calcined at 550 °C in air. 

2.2.2 Material bulk characterization 

Low­angle X­ray diffraction  (XRD) spectra of  the calcined and powdered SBA­15 and KIT­6 silica 
samples  were  acquired  in  an  angular  range  of  2θ  =  0.5–5°  in  steps  of  0.01°  using  a  PANalytical 
EMPYREAN diffractometer  (Malvern PANalytical, Malvern, UK).  The  signals were  collected by a 
PIXcel3D detector (Malvern PANalytical)  in reflection geometry (Bragg–Brentano HD) using Cu Kα 
radiation (0.15406 nm) at 45 kV and 40 mA.    
Nitrogen physisorption measurements were carried out at –196 °C on an Autosorb­iQ2 sorption analyzer 
(Quantachrome  Instruments,  Boynton  Beach,  FL).  Prior  to  the  measurements  powder  samples  were 
outgassed under vacuum at 150 °C for 12 h. Total pore volumes of the samples (Vt) were obtained with 
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the Gurvich rule at a relative pressure of p/p0 = 0.95. Specific surface areas (SBET) were determined by 
means of the Brunauer–Emmet–Teller equation in a range of 0.05 ≤ p/p0 ≤ 0.3. Pore size distributions 

were  calculated  from  the  adsorption  and  desorption  branches  of  the  nitrogen  sorption  isotherms  by 
applying the kernel of (metastable) equilibrium non­local density functional theory (NLDFT) adsorption 
isotherms  and  the  kernel  of  equilibrium  NLDFT  desorption  isotherms,  respectively.32–34  For  this 
purpose, a cylindrical pore model and an amorphous silica surface were assumed. Autosorb 5.2 software 
provided by Quantachrome Instruments was used for conducting the NLDFT calculations. 

2.2.3 Electron tomography 

For physical reconstruction of the mesopore spaces of the prepared SBA­15 and KIT­6 silica materials 
by electron  tomography,35–37  the powder  samples were ground  in  a mortar  and  the obtained powder 
crumbs disposed onto a holey Cu grid (Quantifoil Micro Tools, Jena, Germany). Au nanoparticles were 
added  to  the  crumbs  from  aqueous  suspension  (CMC,  University  Medical  Center,  Utrecht,  The 
Netherlands).  Electron  tomographic  data  were  acquired  using  an  image­corrected  Titan  80–300 
transmission electron microscope (FEI, Hilsboro, OR) operated at an acceleration voltage of 300 kV in 
scanning transmission electron microscopy (STEM) mode with a nominal beam diameter of 0.2 nm. A 
high­angle annular dark­field detector was used to collect STEM images over a tilt range of ±76° with 
angular increments of 2° for both samples. Images were aligned with IMOD 4.738 using Au nanoparticles 
as fiducial markers. The average aligment error was < 1 pixel. 3D reconstructions were obtained by 
applying 25 iterations of the SIRT algorithm, employing the Xplore3D software package version 3.0 
(FEI), followed by ten iterations of the DART algorithm implemented in the ImageJ plugin TomoJ.39–42 
Images were denoised by applying the ImageJ plugins Filters/Median 3D and Enhance Local Contrast 
(CLAHE).  The  reconstructions  are  irregularly  shaped  objects  (Fig.  2.2)  with  maximum  (x,  y,  z) 
dimensions of 691, 593, and 294 nm for SBA­15 and 588, 756, and 521 nm for KIT­6 silica.  

 
Fig. 2.2: 3D reconstructions based on STEM tomography of irregularly shaped crumbs from the prepared SBA­
15 (A) and KIT­6 (B) silica powder samples. 

Five  blocks  were  cut  from  different  regions  of  each  crumb  and  used  as  final  stacks  of  aligned, 
reconstructed, and denoised images with a voxel size of 0.473 nm3. The ESI† shows a surface rendering 
of the ten image stacks (Fig. S2.1 and S2.2) and lists their dimensions (Table S2.1). The image stacks 
serve as geometrical models for the pore­scale simulations of hindered diffusion. 
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2.2.4 Diffusion simulations 

Diffusion was modeled directly in the reconstructions using a random­walk particle­tracking (RWPT) 
method.43,44 It is based on the equivalence of the diffusion equation (where D is the diffusion coefficient 
and c concentration) 

∂𝑐(𝐫,𝑡)

∂𝑡
= 𝐷∇2𝑐(𝐫, 𝑡)                  (4) 

and the stochastic differential equation describing the random walk of a  tracer, presented in discrete 
form as: 

𝐫(𝑡 + δ𝑡) = 𝐫(𝑡) + ∆𝐫 = 𝐫(𝑡) + 𝛂√6𝐷δ𝑡            (5) 

In eqn (5), r(t) denotes tracer position at time t, δt is the elementary time step of the random walk, and 
𝛂 is a vector with random orientation in space and a length governed by a Gaussian distribution with 
zero mean and unity variance. At the start of a simulation, a number of N = 106 passive (nonadsorbing, 
nonreacting) tracers was distributed randomly in the entire mesopore space of a reconstruction, with the 
mirror­boundary  condition  imposed  at  its  external  faces.  This  approach,  which  has  been  validated 
before, assumes that tracers that cross any external face of a reconstruction continue their motion in the 
mirror image of the original domain.44 In this approach, the tracer ensemble samples the entire accessible 
pore  space  at  any  time  and  therefore  reflects  the  spatially  averaged  morphological  properties  of  a 
reconstruction  along  with  their  impact  on  the  overall  diffusive  transport.  During  each  time  step  δt, 
adjusted  such  that  the  mean  diffusive  displacement  did  not  exceed  Δh/10  (where  Δh  is  the  spatial 
resolution  of  a  reconstruction),  the  displacement  of  every  tracer  by  random  diffusive  motion  was 
determined according to eqn (5). Passive interaction of tracers with the pore walls was handled through 
a multiple­rejection boundary condition at the surface, that is, when a tracer hit the impermeable wall 
during an iteration, this displacement was rejected and recalculated until the tracer position remained in 
the void space.45 The accuracy of the RWPT­approach realized with the above boundary conditions has 
been confirmed by comparing the diffusion coefficients simulated in regular arrays of spheres46,47 with 
values determined by the analytical approach.48 After each time step, all tracer positions were monitored 
and a time­dependent diffusion coefficient D(t) was calculated from 

𝐷(𝑡) =
1

6𝑁
 

d

d𝑡
∑ [∆𝑟𝑖(𝑡)]2𝑁

𝑖=1                 (6) 

where Δri(t) is the displacement of the ith tracer after time t. The effective diffusion coefficients Deff,K 
were extracted from the asymptotes of the transient diffusion curves observed in the long­time limit.  
Provided there are no closed pores, which applies to the investigated samples, the pore space accessible 
to an ensemble of pointlike tracers equals the entire void volume of a reconstruction. Finite­size tracers 
have access to only parts of that volume (depending on λ = dtracer/dmeso) due to their steric interactions 
with the pore walls. With an elegant approach49,50 the reduction of the accessible pore space for finite­
size  tracers  can  be  accounted  for  by  eroding  the  pore  space  accessible  to  pointlike  tracers  with  a 
structuring element of size dtracer. For example, in the case of diffusion in a cylindrical pore, the center 
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of a hard spherical  tracer particle of diameter dtracer  is  excluded  from an annular  region of  thickness 
dtracer/2 at the wall. In other words, the void space accessible to the center of a finite­size tracer becomes 
identical to the void space accessible to a pointlike tracer if the pore diameter is reduced by dtracer. The 
approach is illustrated in two dimensions in Fig. 2.3 for a spherical tracer particle in a slice from one of 
the  SBA­15  image  stacks.  We  implemented  this  mathematical  morphology  operation  to  generate 
accessible mesopore space in the ten SBA­15 and KIT­6 silica blocks (Fig. S2.1 and S2.2, ESI†) for 
dtracer­values of 0.94, 1.88, 2.82, 3.76, and 4.70 nm.  

 

Fig. 2.3: Steric interaction between different­sized tracers and the solid wall in SBA­15 silica. (A) Original slice 
from the image stack. Pointlike tracers (λ = dtracer/dmeso = 0) have access to the entire pore space (black). (B) Eroded 
pore space (red) represents the dtracer/2­wide region adjacent to the pore wall, from which the center of a spherical, 
finite­size  tracer  (λ > 0)  is excluded  (C). The  situation  is  therefore equivalent  to using pointlike  tracers  in  the 
remaining pore space. 

By importing this  information  into the RWPT­simulations,  the accessible porosity ε(λ) and  the  local 
effective diffusion coefficient Deff,K(λ) could be determined for different tracer sizes as a function of λ = 
dtracer/dmeso,  the  ratio  between  tracer  size  and  mean  mesopore  diameter.  The  simulation  of  Deff,K  for 
dtracer = 2.82 nm in one of the SBA­15 silica blocks is shown in Fig. 2.4, assuming a tracer diffusion 
coefficient in the bulk liquid of Dm = 1.0 × 10–10 m2s–1 (compatible with, for example, a small protein51). 
The figure shows a steep decrease of D(t) at the beginning of the simulation (inset, Fig. 2.4) due to initial 
contact of the tracers with the impenetrable walls. But within less than five microseconds, the tracers 
have  experienced  the  microstructural  heterogeneity  of  the  pore  space  and  asymptotic  behavior  is 
observed. The diffusion time required  to reach asymptotic behavior translates into a diffusion length 
ldiff = (2Dmt)0.5 of a few tens of nanometers or 3–4 pore diameters (indicated in Fig. 2.4). This diffusion 
length reflects a pore space characterized by a short­ranged heterogeneity, which is quickly sampled by 
the  tracers  to  attain  quasi­homogeneous  behavior52  and  reveal  effective  diffusion  properties.  The 
asymptotic D(t)­value corresponds to Deff,K (dashed line). Please note that the assumed Dm­value serves 
only for illustration purposes, because we are mainly interested in dimensionless relationships in this 
work, cf. eqn (1).    
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Fig. 2.4: Transient diffusion coefficient D(t) simulated for dtracer = 2.82 nm in a reconstructed SBA­15 silica block. 
After a short diffusion time (~5 µs), the tracers have sampled the microstructural heterogeneity of the pore space 
and  the effective diffusion coefficient  Deff,K  (dashed  line)  is  reached. For  recording  the diffusive dynamics  vs. 
observation  time  t  and  diffusion  length  ldiff  a  tracer  diffusion  coefficient  in  the  bulk  liquid  of  
Dm = 1.0 × 10–10 m2s–1 was assumed. 

It should also be pointed out that tracers employed in this approach are entirely passive (neither interact 
with each other nor with the surface), so that the effective diffusivities exclusively sense the hindrance 
to diffusion caused by the steric and hydrodynamic interactions between the finite­size tracers and the 
pore  space  confinement  in  the  SBA­15  and  KIT­6  silicas.  Using  λ­dependent  values  of  the  local 
hindrance factor, Kd(λ) = Deff,K(λ)/Dm, the associated global parameters Deff,H(λ) and H(λ) can be derived 
(cf. eqns (1) and (2)) through the accessible porosity ε(λ), determined as the void volume fraction of an 
eroded pore space at this λ. 

2.3 Results and discussion 

2.3.1 Material preparation and bulk characterization 

Preparation  of  SBA­15  and  KIT­6  silica  samples  for  this  study  relied  on  our  experience  with 
systematically  varied  preparation  conditions  to  induce  the  desired  morphological  changes.16,53  The 
synthesis starts by adding an organic silica source (TEOS) at a precisely controlled temperature (30 °C 
for SBA­15, 35 °C for KIT­6) to a dilute, acidic, aqueous, micellar solution of an amphiphilic, nonionic, 
triblock copolymer (Pluronic P123, for SBA­15) or a blend of Pluronic 123 and n­butanol (for KIT­6) 
employed as structure­directing agent. The synthesis mixture is kept for ~24 h at the starting temperature 
before hydrothermal treatment at higher temperature occurs. Afterwards, the structure­directing agent 
is removed through extraction, and the synthesis is completed through calcination. The intrawall pores, 
which are not ordered, originate from the penetration of poly(ethylene oxide) chains of the Pluronic 
P123 copolymer into the silica framework.54–57 Due to interaction of these hydrophilic chains with the 
polymerizing silica species during mesophase formation, some chains are finally trapped in the silica 



Chapter 2 

66 
 

walls.58,59 After  removal of  the copolymer,  the material exhibits  intrawall pores. Size and associated 
volume of the intrawall pores are highly dependent on details of the preparation process, particularly 
temperature and duration of the hydrothermal treatment. Targeting a narrow pore size distribution with 
negligible tailing, hydrothermal treatment conditions of 140 °C for 24 h (SBA­15) or 48 h (KIT­6) were 
chosen. According to the morphological analysis of the physically reconstructed samples from our initial 
study,29  the preparation protocol of SBA­15 silica  required more adaption  than  that of KIT­6 silica, 
which is why for preparation of the SBA­15 silica sample the molar ratios of the components of the 
synthesis mixture were also changed according to our experience with systematically varied preparation 
conditions.16 Additional details can be found in Section 2.2.1.    
The  silica  powder  samples  were  characterized  through  XRD  spectra  (Fig.  S2.3,  ESI†)  and  nitrogen 
physisorption analysis (Fig. 2.5), with the derived structural parameters summarized in Table 2.1. The 
mesoporosity of the bulk materials was calculated from the physisorption data according to 

𝜀meso =
𝑉t

𝑉t + 
1

𝜌silica

                  (7) 

where Vt is the total pore volume and ρsilica the skeletal density of amorphous silica, taken as 2.1 g cm–3 
(based on helium pycnometry measurements).60 The two samples have a very similar mesoporosity of 
meso = 0.72 (SBA­15) and meso = 0.74 (KIT­6). Plots of the cumulative pore volume (Fig. S2.4, ESI†) 
confirm the absence of microporosity.    
Fig. 2.5A shows the nitrogen physisorption isotherms recorded at –196 °C. The two materials exhibit 
type IVa isotherms with hysteresis loops, revealing steep capillary condensation and evaporation steps 
at high relative pressures, in a p/p0­range of about 0.7–0.8. The hysteresis loops are relatively narrow 
and the hysteresis branches parallel and almost vertical. This signifies homogeneous pore structures and 
underscores that the SBA­15 silica sample  is characterized by a highly interconnected 3D mesopore 
network, mimicking the pristine 3D pore structure of KIT­6.16 Subtle differences in the physisorption 
isotherms are related to the slightly higher εmeso of KIT­6 (larger maximum adsorbed volume) and the 
smaller dmeso of SBA­15 (earlier capillary condensation and evaporation steps, cf. Fig. 2.5A). 

Table 2.1: Structural properties of the mesoporous silica powder samples.a 

 
SBA­15  KIT­6 

     SBET [m2 g–1]  632  524 

Vt [cm3 g‒1]  1.22  1.37 

εmeso [–]  0.72  0.74 

dmeso [nm]  9.1  10.5 

a [nm]  11.1  24.4 

twall [nm]  2.0  1.7 

 
a Specific surface area (SBET), total pore volume (Vt), mesoporosity (εmeso), and mean mesopore diameter (dmeso) 

from nitrogen physisorption analysis; unit cell parameter (a) from XRD analysis; silica wall thickness (twall) 

calculated from a and dmeso as twall = a ‒ dmeso for SBA-15 and twall = a/2 ‒ dmeso for KIT-6. 
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Fig.  2.5:  Characterization  of  calcined  SBA­15  and  KIT­6  silica  powder  samples.  (A)  Nitrogen  adsorption–

desorption isotherms and (B) NLDFT pore size distributions. 

Pore size distributions were derived from both branches of the nitrogen sorption isotherms. Values for 
dmeso were taken from the desorption­based data (Fig. S2.5, ESI†), as equilibrium conditions (desorption) 
are  considered  to  give  a  more  accurate  estimate  of  the  mean  mesopore  size,  whereas  capillary 
condensation  (adsorption)  in  3D  cylindrical  pore  structures  reflects  pore  network  effects.16,53  The 
adsorption­based  pore  size  distributions  (Fig.  2.5B)  give  the  first  indication  that  the  enhanced 
hydrothermal  treatment  suppresses  the  occurrence  of  narrower  pores  in  the  two  materials  rather 
successfully. Lower pore sizes are practically absent from the pore size distribution of KIT­6 (lower 
inset, Fig. 2.5B) and tailing in the pore size distribution of SBA­15 stops at ~6.5 nm (upper inset, Fig. 
2.5B). The pore  size distributions of  the prepared SBA­15 and KIT­6  silica  samples  reflect ordered 
mesoporous  materials  whose  intrawall  pores  approach  dmeso  in  size,  as  intended,  but  only  a  careful 
inspection of the 3D reconstructions will show presence and extent of constrictions in the primary pores 
(cf. blue circles in Fig. 2.1). 

2.3.2 Morphological analysis based on physical reconstructions 

We  discuss  the  morphological  features  of  the  prepared  materials  as  revealed  by  their  physical 
reconstruction  through  STEM  tomography,  focusing  on  detecting  improvements  on  the  known 
morphological challenges of the materials as detailed in Fig. 2.1. For SBA­15 (Fig. 2.6), we note that 
the hexagonally arranged primary pores are laterally connected by intrawall pores to a substantial degree 
(Fig. 2.6A). Fig. 2.6B visualizes that the connecting intrawall pores (yellow) are of similar size as the 
primary mesopores (red), which leads to the expectation that transport of larger tracers through the SBA­
15 silica sample should be possible. A particularly beneficial feature in this respect is also shown in Fig. 
2.6B: two plugs (black circles) in the primary pores (red) are each flanked by two intrawall mesopores 
(yellow),  which  provide  bypasses  around  the  dead­ends.  Fig.  2.6B  and  2.6C  highlight  a  further 
morphological improvement with respect to Fig. 2.1 regarding the constrictions in the primary pores.  
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Fig. 2.6: Sections from two reconstructed SBA­15 silica blocks, highlighting important morphological features of 
the prepared SBA­15 silica sample. (A) Hexagonally arranged primary mesopores (red) are laterally connected by 
secondary (intrawall) mesopores (yellow); dashed lines indicate that maximum pore dimensions are a little deeper 
inside the material. (B) Black circles identify plugs (dead­ends) in the primary pore system (red); intrawall pores 
are  colored yellow. Blue disks highlight  four  constrictions  in  the primary pore  system. Associated white bars 
represent local pore width in the front­view plane (B) and in the center of the constrictions (a little deeper inside 
the material), as retrieved through the inside­out view (C). 

To visualize the effect, we estimated the maximum size of four selected constrictions (numbered in the 
figure). White bars indicate the local dimensions of the constrictions, which appear quite small in the 
front­view plane of Fig. 2.6B. Because the center of all constrictions is located a little deeper inside the 
material, only the inside­out view of Fig. 2.6C offers a realistic picture of their maximum openings. 
They are sized within 7.7–10.2 nm, which shows that  the adapted preparation conditions sufficed  to 
widen not only the intrawall pores, but also the narrow constrictions found in the primary pore system 
of previous SBA­15 silica sample.29    
The  primary  pore  system  of  cubic  KIT­6  silica  is  a  highly  interconnected  network  based  on  two 
continuous  and  interwoven channel  sub­networks, which  are  separated by an amorphous  silica wall 
believed to follow the infinite periodic minimal surface.15,31 The latter is considered as dividing the space 
into two infinite, unconnected, but mutually intertwined, periodic channel networks.61,62 Connections 
between  the  two  sub­networks  are  enabled  by  intrawall  pores,  as  can  be  seen  in  the  physical 
reconstruction  of  the  KIT­6  silica  sample  (Fig.  2.7).  Because  the  silica  wall  is  relatively  thin 
(twall = 1.7 nm,  Table  2.1),  these  larger  connections  (~5  nm  diameter)  appear  as  wall  openings  (red 
circles in Fig. 7A), that  is, as windows between the closely spaced  twin pore systems rather than as 
extended cylindrical pores. Visual inspection of the reconstructed KIT­6 silica block (Fig. 2.7B) reveals 
the primary pore system as relatively open, uniform, and unobstructed (Fig. 2.7C). Narrow constrictions 
as found in a previous KIT­6 silica sample are absent from the reconstruction. Determining the wall­to­
wall distances in the primary pore system for the section of the reconstruction in Fig. 2.7C gave a lower 
bound of ~8.5 nm, in accordance with light (relaxed) constrictions that are likely uncritical for hindered 
diffusion. 
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Fig. 2.7: Sections from one of the reconstructed KIT­6 silica blocks, highlighting important morphological features 
of the prepared KIT­6 silica sample. (A) Intrawall mesopores appear as windows (red circles) in the thin silica 
wall between the two continuous, interwoven mesopore networks (B). (C) Close­up view of the 3D pore system, 
showing light (mostly uncritical) constrictions and relatively open transport pathways. 

Overall,  the  physical  reconstructions  of  the  SBA­15  and  KIT­6  silica  samples  (Fig.  2.6  and  2.7) 
demonstrate that the morphological targets were met by the adapted preparation conditions.  A better 
connected  and  less  constricted  pore  space  promises  less  hindrance  to  the  diffusion  of  larger  tracers 
through the ordered mesoporous silicas, which is analyzed next. 

2.3.3 Accessible porosity, hindered diffusion, global hindrance factor 

The five blocks from each reconstruction (cf. Fig. S2.1 and S2.2, ESI†) served as geometrical models in 
pore­scale simulations of diffusion, which allowed us to quantify the accessible porosity of the material 
(Fig. 2.8A) as well as the effective diffusion coefficient (Fig. 2.8B) for spherical tracers in dependence 
of λ = dtracer/dmeso. The data in Fig. 8 represent the average from the five blocks with error bars indicating 
95% confidence  intervals. For  comparison,  data  from our previous work29 with differently prepared 
SBA­15 and KIT­6 silica samples are included. For dimensionless plots, the data were normalized by 
the respective values for pointlike tracers (λ = 0),  that is,  the accessible porosities ε(λ) by ε0 and the 
effective diffusivities Deff,K(λ) by Deff,K(λ = 0). The effective diffusivities for pointlike tracers were also 
used to calculate the diffusive tortuosities τ0 = Dm/Deff,K(λ = 0) for the two materials (cf. eqn (3)). Values 
for ε0 and τ0 are summarized in Table 2.2.  

Table 2.2: Transport properties of the reconstructed mesopore spaces.a 

 
SBA­15  KIT­6 

     
ε0 [–]  0.69 ± 0.01  0.70 ± 0.01 

τ0 [–]  1.41 ± 0.05  1.31 ± 0.04 
 

a Porosity  (ε0)  and  diffusive  tortuosity  (τ0)  underlying  the  transport  of  passive,  pointlike  tracers  through  the 
mesoporous  silicas  (cf.  eqn  (3));  obtained by pore­scale  simulations of  diffusion  in  the  five blocks  from each 
reconstruction (cf. Fig. S2.1 and S2.2, ESI†). 
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Mean ε0­values from the five blocks of each reconstruction fall within ~5% of the εmeso­values (Table 
2.1) derived from nitrogen physisorption analysis of the powder samples ‒ an excellent agreement 

reflecting that representative sections of the physical reconstructions recover the bulk porosity of the 
materials.  Because  the  accessible  porosity  data  from  the  current  SBA­15  and  KIT­6  silica  samples 
display very similar behavior, the data from the two samples were treated as one set to derive an equation 
describing the accessible porosity of the materials as a function of λ.    
The black dotted line in Fig. 2.8A represents the best fit to the porosity data of both materials by the 
following equation: 

𝜀(𝜆)

𝜀(𝜆=0)
= 1 − 2.62𝜆 + 1.31𝜆2                (8) 

Eqn (8) predicts zero accessible porosity for λ = 0.51 and that the mesopore space of the two materials 
closes off much slower with increasing λ than observed for previous SBA­15 and KIT­6 silica samples, 
reflecting widened intrawall pores and relaxed constrictions in the primary pores of the current materials. 
Further comparison is made in Fig. 8A with sol–gel based, random mesoporous silicas (grey line), for 
which the following accessible porosity relationship was earlier derived:63,64 

𝜀(𝜆)

𝜀(𝜆=0)
= 1 − 2.20𝜆 + 1.25𝜆2                (9) 

Although eqn (8) and eqn (9) are qualitatively similar, the accessible porosity relationship for random 
mesoporous silicas, eqn (9), predicts a slower decrease with increasing λ than the accessible porosity 
relationship for ordered mesoporous silicas, eqn (8). At λ = 0.51, the point of zero accessible porosity in 
the current ordered mesoporous silica samples,  the random mesoporous  silicas still permit access  to 
~20% of the pore space (that is, of ε0) and continue to permit access even up to λ = 0.9. Access to the 
pore space over such a wide range of λ­values can become a decided disadvantage in applications where 
spatial confinement effects and molecular size­selectivity are of prime importance, as discussed in the 
Introduction.   
This observation can be rationalized by the width and shape of the pore size distributions typical for 
random mesoporous silicas. The distributions have relative standard deviations of 30–50% and higher 
(compared  with  ~10%  for  the  current  SBA­15  and  KIT­6  silicas)  and  may  feature  a  prominent  tail 
towards larger pore sizes.63–66 Indeed, experimental evidence shows that the pore size distributions often 
can  be  described  by  a  log­normal  function.65  These  larger  pores  (absent  in  the  current  ordered 
mesoporous silicas) establish pathways that remain open when other pores have closed off at higher λ­
values.     
The benefit from the morphological improvements made with the current SBA­15 and KIT­6 silicas for 
diffusive transport becomes obvious in Fig. 2.8B. As observed for the accessible porosities in Fig. 2.8A, 
the effective diffusion coefficients at a given λ (and ε(λ)) of the two materials are so similar that they 
were treated as one data set. The combined diffusivities are best fitted by the following equation (black 
dotted line, Fig. 2.8B):  

𝐷eff,K(𝜆)

𝐷eff,K(𝜆=0)
= 1 − 0.49𝜆 − 5.89𝜆2 + 6.12𝜆3                     (10) 
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At λ ≈ 0.3, when diffusive transport in previous SBA­15 and KIT­6 silica samples is almost zero, finite­
size tracers in the current materials retain about 50% of Deff,K(λ = 0), the effective diffusivity observed 
for pointlike tracers. Up to λ ≈ 0.3, the current ordered mesoporous silicas have comparable transport 
dynamics to random mesoporous silicas, which are characterized by the following effective­diffusivity 
relationship (grey line in Fig. 2.8B):63,64 

𝐷eff,K(𝜆)

𝐷eff,K(𝜆=0)
= 1 − 1.22𝜆 − 0.58𝜆2 − 5.20𝜆3 + 13.35𝜆4 − 7.46𝜆5               (11) 

For λ > 0.3, however, the random mesoporous silicas enable higher diffusivities than the current ordered 
materials, for the reasons outlined above (width and shape of the pore size distribution).  

 

Fig. 2.8: Dependence of accessible porosity (A) and effective diffusivity (B) in physical reconstructions of SBA­
15  and  KIT­6  silica  on  λ,  the  ratio  of  tracer  size  to  mean  mesopore  diameter.  Data  are  normalized  by  the 
corresponding values for pointlike tracers, ε0 and Deff,K(λ = 0). Solid and open circles represent data from current 
and previous29 samples, respectively. 

The implications that emerge from Fig. 2.8 with respect to the different pore size distributions of the 
ordered and random mesoporous silicas are supported by theoretical work about the effect of differently­
shaped  pore  size  distributions  (truncated  Gaussian  and  log­normal)  on  membrane  selectivity  and 
transport:67 Sieving coefficients and hindered diffusivities both increased significantly with increasing 
width of the distribution because of the large contribution from solute flux through the largest pores. 
   
The expressions for accessible porosity and effective diffusivity, eqns (8) and (10) (cf. Fig. 2.8), were 
used  to  construct  finally  the  global  hindrance  factor  H(λ)  for  the  studied  materials.  Since 
Kd(λ) = Deff,K(λ)/(Deff,K(λ = 0)τ0), we obtain: 

𝐻(𝜆) = 𝐾d(𝜆)𝜀(𝜆) =
𝜀0

𝜏0
(1 − 3.11𝜆 − 3.29𝜆2 + 20.94𝜆3 − 23.80𝜆4 + 8.04𝜆5)      (12) 
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Inserting the respective values for ε0 and τ0 for SBA­15 or KIT­6 silica from Table 2.2 into eqn (12) 
allows to quantify hindered diffusion of finite­size, passive tracers for any given λ through the respective 
ordered  mesoporous  silica  material.  Using  the  two  panels  in  Fig.  2.8,  the  global  hindrance  factor 
H(λ) = Deff,H(λ)/Dm is obtained by multiplying the ε(λ)/ε0­value at a λ of interest in Fig. 2.8A with the 
corresponding value for Deff,K(λ)/Deff,K(λ = 0) in Fig. 8B, the result of which represents the value given 
by the bracket expression in eqn (12), and then multiplying the value with ε0/τ0 from Table 2.2 for a 
given material.   
At this point it should be recalled that the tracers employed in the approach leading us to eqn (12) were 
intentionally passive, i.e., they neither interact with each other nor with the surface. Similarly, although 
diffusion  in  this  work  addresses  liquid­phase  transport,  no  differentiation  is  made  regarding 
physicochemical properties of the liquid and solid–liquid interactions. As a consequence, the effective 
diffusivities do not reflect interfacial phenomena such as adsorption, partitioning, and surface diffusion, 
which can be complex and are usually highly specific for a given combination of solvent, solute, and 
surface.68–73 Although  interfacial phenomena offer  a wealth of  information,  they are undesirable  for 
establishing  hindrance  factor  expressions  that  exclusively  characterize  the  hindrance  to  diffusion 
originating in the steric and hydrodynamic interactions between tracers and pore space confinement. 
Further, in the limiting case of λ = 0,  the effective diffusivities Deff,K(λ = 0) of the passive, pointlike 
tracers provide access to the true diffusive tortuosities calculated as τ0 = Dm/Deff,K(λ = 0) = 1/Kd(λ = 0) 
(cf. eqn (3)).74 However, in a subsequent step, interfacial phenomena may be embedded into the present 
methodology  through a multiscale  simulation approach using,  for example, a microscopic stochastic 
model of sorption at the solid–liquid interface,75 or by implementing molecular dynamics simulations 
that allow to resolve interfacial structure and dynamics in great (molecular) detail.76 This approach can 
also be extended by covering multiscale porosities76  to study,  for example, mass  transfer within and 
between intracrystalline and intercrystalline pore spaces in beds of hierarchical zeolite particles.77–80 

2.3.4 Evolving pore networks, tortuosity–porosity relationships 

With the established relationships for accessible porosity and effective diffusivity, culminating in the 
quantitative expression  for  the global hindrance  factor, eqn  (12),  it becomes possible  to analyze  the 
tortuosity–porosity  relationship  for  the  finite­size  tracers,  namely  the  dependence  of  τ(λ)  on  ε(λ),  in 
detail. The tortuosity–porosity relationship characterizes the apparent tortuosity τ(λ) = Dm/Deff,K(λ) that 
tracers  experience  in  an  increasingly  inaccessible  pore  space  as  represented  by  ε(λ).  Recalling  the 
implemented  mathematical  morphology  operation  (cf.  Fig.  2.3),  this  situation  is  analogous  to  the 
diffusive tortuosity that pointlike tracers experience in an increasingly eroded pore space. “Effective” 

pore networks evolve in the first scenario through increasing the tracer size in a conserved pore space 
and  in  the  alternative  scenario  through  increasing  the  consolidation  of  the  pore  space  at  conserved 
(pointlike) tracer size. For both scenarios, the simulations offer a systematic tortuosity–porosity data set, 
plotted in Fig. 2.9 for the finite­size tracers as τ(λ)–ε(λ). Interestingly, the combined data from the current 
SBA­15 and KIT­6 silica samples can be fitted by the following equation: 

𝜏(𝜆) = 𝐴(𝜀(𝜆))
(1−𝑚)                           (13) 
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In  its  essence,  eqn  (13)  reflects  ARCHIE’s empirical law,81  which  has  been  used  for  a  long  time  in 
petroleum industry to relate the formation factor F from electrical resistivity measurements to sediment 
porosity via F = ε–m, with τ = εF for the electrical tortuosity.82 The extra parameter A in eqn (13) is a 
modification added by WINSAUER et al.83 to ARCHIE’s law. This variant form of ARCHIE’s law generally 

fits the experimental data much better than the original form (A = 1), as discussed by GLOVER.84 The 
best fit of eqn (13) to the data in Fig. 2.9 (grey line) gives A = 0.58 and m = 2.08 (m = 1.91 with A = 1). 
   
Values of m = 1.2 ‒ 4.4 have been reported for the exponent in ARCHIE’s law.85 Because m increases 
with cementation, it is called the cementation exponent. In consolidated media, the effect of cementation 
on m was revealed by comparing non­fused (m ≈ 1.35–1.4) and fused spherical particles (m ≈ 1.5–1.8) 
of uniform size (Table 1  in FRIEDMAN85). For example, m = 1.5 was reported for  fused glass beads 
studied over a porosity range of 0.02–0.38.86 The good fit of eqn (13) to the data in Fig. 2.9 may therefore 
be  unsurprising  considering  that  the  pore  network  evolution  in  both  materials  accompanying  the 
decrease in ε(λ) with increasing λ reflects a substantial, albeit abstract, consolidation.    
Another popular tortuosity–porosity relationship is the modified WEISSBERG equation:82 

𝜏(𝜆) = 1 − 𝑝ln(𝜀(𝜆))                           (14) 

This equation (with one adjustable parameter p), however, fails to describe the data set in Fig. 2.9 over 
the entire porosity range, but produces a good fit with p = 0.89 to the data for the porosity range of ε(λ) = 
0.26–0.70 (red line, Fig. 2.9).  

 

Fig. 2.9: Tortuosity τ(λ) = Dm/Deff,K(λ) experienced by finite­size tracers in physical reconstructions of SBA­15 
and KIT­6 silica as a function of the accessible porosity ε(λ). The inset is a zoom into the limited data range fitted 
by eqn (14). 

WEISSBERG87 originally derived eqn (14) with p = 0.5 on a  theoretical basis as  lower bound  for  the 
diffusive tortuosity in random arrangements of freely overlapping spheres. This p­value can indeed be 
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closely matched with fixed and fluidized beds of spherical particles.88–90 The value of p = 0.89 received 
for the data in Fig. 2.9 demonstrates that the model structures used for the derivation of the WEISSBERG 
equation are unsuitable to describe the pore networks and their evolution with increasing λ in the SBA­
15 and KIT­6 silica materials. Their structures cannot be regarded as a random arrangement of spheres 
creating  a  material  with  porosity  ε0.  Still,  the  description  of  hindered  diffusion  through  random 
mesoporous silicas has been improved by embedding the RENKIN equation91 for diffusion of spherical 
tracers through a single, cylindrical pore of constant cross­section in a model that employs the modified 
WEISSBERG  equation  to  account  for  the  evolution  of  the  pore  network  morphology  (and  diffusive 
tortuosity)  with  increasing  λ.92  Tortuosity–porosity  relationships  are often  tested  against  self­similar 
materials, which can be prepared with systematically adjusted pore size and porosity without inducing 
major changes in topology. A prominent example are fluidized beds of particles undergoing a smooth, 
stepwise densification from loose to dense packings, which leaves the topological properties of the pore 
network largely unaffected.89 On the other hand, relationships valid for pointlike tracers in self­similar 
materials  do  not  necessarily  capture  the  topological  evolution  of  a  given  material  experienced  by 
different­sized  tracers.  Fig.  2.10  demonstrates  how  different  the  same  mesoporous  SBA­15  silica 
material looks to tracers according to their size. The top row shows the actual geometrical models of the 
pore space at four different dtracer­values and the bottom row highlights the associated pore networks (for 
clarity, only accessible pores in the first ten images of the stack are visualized).  

 

Fig.  2.10:  Geometrical  models  representing  increasingly  eroded  pore  spaces  (top  row)  and  the  associated 
accessible pore networks (bottom row) available  to  tracers of  increasing size  in a  reconstructed SBA­15 silica 
block (image stack A, ESI†) with dimensions (x × y × z) of 196 × 189 × 38 nm3. For better visibility the accessible 
pore networks (bottom row) refer only to the first ten images of the stack (i.e., to a volume of 196 × 189 × 5 nm3). 

As λ increases, the network drastically thins out as whole pathways are closed off. This confirms that 
the decrease of Deff,K(λ) in Fig. 2.8B reflects the removal of entire branches from the pore network and 
therefore complex topological changes in the accessible pore space. These also cause the steep increase 
in τ(λ) seen in Fig. 2.9 for ε(λ) < 0.26. The large changes in tortuosity (by a factor of ~20, from τ0 = 1.3 
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at ε0 = 0.70 up to τ(λ) = 25 at ε(λ) = 0.03) are much better captured by the exponent in eqn (13) than the 
logarithm in eqn (14). However, in the range of ε(λ) = 0.26–0.70 (when the curve flattens out), eqn (14) 
provides a better description of the tortuosity–porosity relationship than eqn (13), as the inset in Fig. 2.9 
illustrates.    
To summarize, the analysis of the τ(λ)–ε(λ) data in Fig. 2.9 shows that eqn (13) and eqn (14) can be used 
as  tortuosity–porosity  relationships  in  the  low­porosity  and  high­porosity  regions  of  the  data, 
respectively. These equations may therefore be applied directly (without assuming a pore geometry or 
pore network model), and as an alternative to eqn (12), to estimate local and global hindered diffusion 
coefficients,  Deff,K(λ)  =  Dm/τ(λ)  and  Deff,H(λ)  =  Deff,K(λ)ε(λ),  of  passive,  finite­size  tracers  from  their 
accessible  porosity  data  ε(λ).  Information  on  ε(λ)  can  be  obtained  experimentally,  for  example,  by 
characterizing  particulate  materials  in  a  fixed­bed  apparatus  using  methodology  from  inverse  size­
exclusion chromatography.92 

2.4 Conclusions 

To our knowledge, eqns (8), (10), and (12) are the first expressions in the literature quantifying hindered 
accessibility  and  diffusion  in  realistic  geometrical  models  of  ordered  mesoporous  silicas.  Using  the 
individual  values  for  ε0  and  τ0  (Table  2.2)  in  eqn  (12),  this  expression  quantifies  the  hindrance  to 
diffusion  through  the  respective  material  experienced  by  finite­size,  passive  tracers  for 
λ = dtracer/dmeso < 0.5. Alternatively, local and global hindered diffusion coefficients Deff,K(λ) and Deff,H(λ) 
can be estimated with eqns (13) and (14) at high and low λ­values (low and high accessible porosities), 
respectively,  as  identified  by  analyzing  the  tortuosity–porosity  relationships  for  the  tracers.  
Importantly, the morphology–transport relationships of a material are its signature, reflecting individual 
morphological  features  as  the  consequence  of  its  preparation  history.  Ideally,  the  relationships  are 
known  before  a  material  is  used  as  support  structure  in  a  specific  application.  The  elaborate 
reconstruction–simulation approach required to establish this knowledge is justified considering that the 
highly  detailed  information  provided  can  become  the  decisive  factor  to  boost  the  performance  of  a 
process  that  depends  on  efficient  and  selective  transport  of  finite­size  solutes.  This  applies  to  the 
immobilization of bulky species (enzymes, polyhedra, clusters) on the internal surface of a material, the 
transport of educts to and products from active surface sites, and the controlled release of drugs from 
the pore space. In each case, the effects of hindered diffusion, decoupled from interfacial effects, should 
be quantitatively known.    
The morphology–transport relationships established for SBA­15 and KIT­6 silicas in this work improve 
the current perception of ordered mesoporous materials. An ordered pore space is often believed to result 
in superior transport properties, such as higher diffusivities and selectivities, compared to a disordered 
structure.  Fig.  2.8A  has  demonstrated  that  the  ordered  mesoporous  silicas  provide  indeed  a  more 
selective  access  to  the  mesopore  space  (closing  off  sharper  for  larger  tracers)  than  the  random 
mesoporous  silicas.  But  further  analysis  suggests  that  this  is  solely  due  to  the  narrow  pore  size 
distribution. From this perspective,  the main benefit of a material synthesis using structure­directing 
agents lies in realizing a narrow pore size distribution that can in turn increase the size­selectivity of a 
process. On the other hand, the current ordered silica samples, which possess highly interconnected 3D 
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mesopore networks and optimized pore size distributions (only ~10% relative standard deviation and 
negligible  tailing), do not enable faster hindered diffusive  transport  than random mesoporous silicas 
(Fig. 2.8B). The morphological analysis has shown that diffusion of finite­size tracers in a pore space 
with a narrow size distribution becomes highly sensitive to local nonuniformities in pore shape, size, 
and  connectivity,  including  the  remaining  constrictions  in  the  primary  pores.  Hindered  diffusion 
observed on a macroscopic scale therefore depends critically on local aspects of pore interconnectivity 
and the deviation from idealized pore geometries. This sensitivity  to details of the actual pore space 
indicates that structural properties as manifested in XRD patterns and a narrow pore size distribution are 
unreliable predictors of the transport dynamics. Further research is needed for a critical appraisal of the 
benefits  of  ordered  vs.  random  pore  structures  and  narrow  vs.  wide  pore  size  distributions  for  the 
transport properties of mesoporous silica materials employed as support structures. 

 

2.5 Supporting Information  

 

 

Fig. S2.1: Five blocks from different regions of the SBA­15 crumb (cf. Fig. 2.2A in the main text) used as final 
image stacks. Their dimensions are summarized in Table S2.1. 

 

 

 

Fig. S2.2: Five blocks from different regions of the KIT­6 crumb (cf. Fig. 2.2B in the main text) used as final image stacks. 
Their dimensions are summarized in Table S2.1.  
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Table S2.1: Dimensions of the final  image stacks used as geometrical models in the pore­scale simulations of 
hindered diffusion. 

Image stack  SBA­15 (x × y × z nm3)  KIT­6 (x × y × z nm3) 

     A  196 × 189 × 38  212 × 212 × 203 

B  122 × 126 × 71  157 × 159 × 207 

C  122 × 126 × 71  282 × 151 × 144 

D  113 × 243 × 90  188 × 138 × 139 

E  235 × 235 × 85  116 × 118 × 221 

 

 

 

Fig.  S2.3:  Low­angle  XRD  patterns,  which  show  the  expected  reflections  and  symmetry  groups  for  the  two 
materials, that is, hexagonal p6mm symmetry for SBA­15 silica and cubic Ia3̅d symmetry for KIT­6 silica.  
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Fig. S2.4: NLDFT cumulative pore volumes for the SBA­15 and KIT­6 silica powder samples. 

 

 

Fig. S2.5: Pore size distributions derived from the desorption branches of the physisorption isotherms. The dmeso­
values for SBA­15 and KIT­6 silica refer to the mode of the distributions. 
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Abstract 

Spatial confinement effects on hindered  transport  in mesoporous silica particles are quantified using 
reconstructions of their morphology obtained by electron tomography as geometrical models in direct 
diffusion  simulations  for  passive,  finite­size  tracers.  We  monitor  accessible  porosity  and  effective 
diffusion  coefficients  resulting  from  steric  and  hydrodynamic  interactions  between  tracers  and  pore 
space confinement as a function of λ = dtracer/dmeso, the ratio of tracer to mean mesopore size. For λ = 0, 
pointlike tracers reproduce the true diffusive tortuosities. For λ > 0, derived hindrance factors quantify 
the extent to which diffusion through the materials is hindered compared with free diffusion in the bulk 
liquid.  Morphology–transport  relationships  are  then  discussed  with  respect  to  the  immobilization, 
formation,  and  transport  of  key molecular  species  in  the  ring­closing metathesis  of  an  α,ω­diene  to 
macro(mono)cyclization product  and oligomer,  with a 2nd­generation Hoveyda–Grubbs  type  catalyst 
immobilized inside the mesopores of the particles. 

3.1 Introduction 

The  restricted  access  to  and  hindered  diffusion  within  mesoporous  materials  are  key  transport 
phenomena of many technological and environmental processes. It includes the immobilization of bulky 
species (molecular catalysts, enzymes, polyhedra, and clusters) on the internal surface of a material,1 
the transport of substrates to and products from the catalytically active sites,2 or the controlled release 
of  drugs,  macromolecules,  and  cells.3  For  example,  the  partial  size­exclusion  of  solutes  from  the 
mesopore  space  of  a  material  during  surface  functionalization  diminishes  its  mass  loadability  and 
reduces the active surface area, and when the interfacial kinetics are fast, hindered diffusion to and from 
the active surface sites becomes limiting to the process performance.4 Therefore, the impact of restricted 
access and hindered diffusion, the result of steric and hydrodynamic interactions between solutes and 
the  confinement,  should  be  quantitatively  known  to unravel  the  complex  interrelationships  between 
morphology,  transport,  and  reaction  and  thereby  allow  for  a  better  tuning  of  efficiency,  yield,  and 
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selectivity.  This  is  particularly  important  for  multifunctional  catalysts  designed  to  achieve  higher 
catalytic activity or enantioselectivity, because transport under confinement triggers synergistic effects 
between  different  functionalities  immobilized  on  the  same  surface.5  
Hindrance  to  diffusion  of  passive,  finite­size  tracer  molecules  (or  solutes)  within  and  through  a 
mesopore network is commonly expressed by the local and global hindrance factors Kd(λ) and H(λ) as a 
function of λ = dtracer/dmeso, the ratio of the hydrodynamic diameter of the tracers (dtracer) and the mean 
mesopore diameter (dmeso):6 

𝐻(𝜆) =
𝐷eff,H(𝜆)

𝐷m
= 𝐾d(𝜆)𝜀(𝜆)                (1) 

The effective diffusion coefficient Deff,H(λ) in Eq. (1) characterizes the global process of diffusion into 
and through the mesopore network (subscript H) and H(λ) is the global hindrance factor relating Deff,H(λ) 
to Dm,  the diffusion coefficient  for  free diffusion  in  the bulk  liquid.  In  contrast,  Kd(λ) = Deff,K(λ)/Dm 
denotes the local hindrance factor representing effective diffusion locally within the mesopore network 
(subscript K). ε(λ) in Eq. (1) characterizes the accessible mesoporosity as a function of λ. Accordingly, 
Deff,K(λ) = Deff,H(λ)/ε(λ).   
Hindered  diffusion  behavior  expressed  by  Eq.  (1)  and  the  underlying  morphology–transport 
relationships can be elucidated by a combination of experimental approaches.[7] For example, pulsed 
field­gradient  nuclear  magnetic  resonance  (PFG­NMR)  has  been  widely  applied  to  study  effective 
diffusion coefficients in mesoporous materials used in heterogeneous catalysis.[8] The ensemble­tracking 
offered  by  PFG­NMR  can  be  complemented  by  single­molecule  tracking  data  from  spectroscopic 
techniques that allow to monitor trajectories of single molecules and localize their position of sorption 
and reaction,9 or even track diffusion in a pore while imaging the pore itself and thus quantify diffusion 
at  the  single­pore  level.10  IR  microimaging,  on  the  other  hand,  has  been  implemented  to  determine 
effectiveness factors of catalyst particles in a single measurement by recording the evolution of substrate 
and  product  concentration  profiles  emerging  from  the  direct  interplay  of  diffusion  and  reaction.11  
However,  utilizing  spatial  confinement  effects  in  the  best  possible  way  implies  that  the  three­
dimensional  morphology  of  the  mesopore  space  under  consideration  is  known,  for  example,  by 
reconstruction  via  electron  tomography12  (becoming  increasingly  popular  as  part  of  multiscale 
characterization studies13), and transferred into geometrical models that are subsequently used in direct, 
i.e., pore­scale simulations of diffusion, sorption, and reaction. This approach provides a comprehensive 
picture regarding the spatiotemporal dynamics of relevant physicochemical phenomena under explicit 
consideration of the three­dimensional morphology of a material, reflecting its individual preparation 
history. Further, diffusion can be simulated independently from processes coupled in experiments, such 
as sorption and reaction, by using passive tracers that neither interact with the material surface nor with 
one another. It allows to quantitatively resolve key aspects of diffusive transport in porous media, such 
as the effect of increasing tracer size on hindrance to diffusion. In addition, transport properties can be 
related to morphological characteristics, for example, to the porosity and connectivity of the accessible 
pore space, which leads in a straightforward manner to the morphology–transport relationships reflected 
by the local hindrance factor Kd(λ) and the accessible porosity function ε(λ) in Eq. (1).    
We  recently  presented  a  three­step  methodology  comprising  (i)  the  physical  reconstruction  of  the 
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mesopore  space  by  electron  tomography,  (ii)  morphological  analysis  of  the  reconstructed mesopore 
space regarding, for example, its pore network connectivity and the presence of dead ends, constrictions, 
and closed pores, and (iii) direct numerical simulation of tracer diffusion in the reconstructed mesopore 
space with systematic variation of the fundamental parameter λ to derive expressions for the hindrance 
to diffusion in ordered and random mesoporous silicas.14 These studies provided two main conclusions. 
First, expressions for hindered diffusion decrease more strongly with increasing λ than predicted by the 
equations of RENKIN15 and DECHADILOK and DEEN.6b Those equations were derived for diffusion of 
spherical  particles  in  a  cylindrical  pore  by  resolving  the  problem  of  enhanced  drag  from  the 
hydrodynamic Stokes­friction effect. It appears unsurprising that three­dimensional networks of pores 
with varying shape, size, and connectivity close off at  lower  λ due  to bottle­necking at smaller pore 
openings  than  a  single,  straight  cylinder.  Morphology–transport  relationships  of  a  material  are  its 
signature, reflecting individual morphological features as the consequence of its often highly specific 
preparation history.14c Second, ordered mesoporous silicas enable more selective access to the mesopore 
space  (closing  off  sharper  with  larger  tracer  size)  than  random  mesoporous  silicas.14d  From  that 
perspective the main benefit of the studied ordered silicas (SBA­15 and KIT­6) is a narrow pore size 
distribution that, in turn, increases the size­selectivity of a process. On the other hand, ordered silicas 
did not provide faster diffusion than the random silicas. Morphological analysis has shown that diffusion 
in a pore space with a narrow pore size distribution becomes highly sensitive to local nonuniformities 
in pore shape, size, and connectivity. Thus, hindered diffusion observed on a macroscopic scale depends 
critically  on  local  aspects  of  pore  interconnectivity  and  the  actual  deviations  from  idealized  pore 
geometries. This sensitivity to details of the pore space suggests that structural properties as manifested 
in XRD patterns and a narrow pore size distribution are unreliable predictors of the transport dynamics 
and that quantitative measures of morphological and transport properties require physical reconstruction 
of the mesoporous materials.14d   
Motivated by these results, we initiated further research on potential benefits of ordered vs. random pore 
structures and narrow vs. wide pore size distributions for the transport properties of mesoporous silica 
materials employed as support structures in heterogeneous catalysis. An additional goal was to establish 
a continuous­flow microreactor platform to complement the insight coming from simulations and for 
comparison  with  batch  experiments.  In  this  work,  we  present  our  first  results  on  the  morphology–

transport–functionality relationships for spherical silica particles with a mean particle size of ~5 µm and 
mean nominal mesopore sizes of 6 and 10 nm employed in the ring­closing metathesis (RCM) of an 
α,ω­diene  with  a  2nd­generation  Hoveyda–Grubbs  type  catalyst  selectively  immobilized  inside  their 
mesopores. While catalyst immobilization generally aims at improved catalyst stability, its separation 
from  the  reaction  mixture,  and  reuse  of  the  catalyst,  and  many  approaches  are  available  today  for 
immobilization of Grubbs­type catalysts on organic and inorganic supports,16 another highly attractive 
feature of catalyst immobilization in mesopores of different size and shape is the confinement effect and 
its impact on catalyst selectivity.5,17 This has been highlighted with a biomimetic approach that employs 
a biological concept to reduce the drawbacks related to macrocyclization.18 The concept is based on pore 
size­selective  immobilization  of  a  ruthenium­based  olefin  metathesis  catalyst  inside  the  pores  of  a 
support. If the pore diameter is tuned such that only one substrate molecule can approach the catalyst 
inside a pore at a given time, then macro(mono)cyclization (MMC) will prevail and oligomerization (O) 
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should be suppressed. With this approach, the ratio between MMC product and all undesired O products 
resulting  from  acyclic  diene  metathesis  polymerization  (ADMET)  could  be  increased  from  0.55, 
corresponding to 35% MMC product obtained with the homogeneous catalyst, up to 1.49, corresponding 
to 60% MMC product.18  Importantly, a  relationship between  the MMC/O ratio and  the substrate­to­
pore­size ratio has been established in that work.   
Here, we build on our accumulated experience with the reconstruction–simulation approach as applied 
to mesoporous silicas14 and the experimental results achieved with the biomimetic approach to fine­tune 
spatial confinement effects for increased macrocyclization selectivity.18 We start with illustrating the 
individual steps necessary to establish morphology–transport relationships for the silica particles, which 
– due  to  their different pore size distributions and mean mesopore sizes at conserved particle  size – 
reveal different solute­size selective transport properties. Confinement effects on hindered transport are 
quantified  by  using  the  reconstructed  pore  space  morphologies  as  geometrical  models  in  diffusion 
simulations with finite­size  tracer molecules. Accessible porosity and effective diffusion coefficients 
are recorded systematically as a function of λ, the ratio between the hydrodynamic diameter of the tracers 
and  the  mean  mesopore  diameter.  These  relationships,  general  functions  of  λ  for  the  investigated 
materials, are then specifically discussed in view of the transport properties of the homogeneous catalyst 
(relevant  for  its  immobilization  inside  the  mesopores),  substrate  molecules  (relevant  for  catalyst 
accessibility  in  the  pores),  as  well  as  the  MMC  and  O  products  (relevant  for  their  formation  via 
competitive olefin metathesis pathways). 

3.2 Results and Discussion 

3.2.1 Properties of the bulk mesoporous silica samples 

Structural properties of the silica particles with nominal mesopore diameters of 60 and 100 Å (the two 
materials  are  accordingly  denoted  as  Si60  and  Si100  furtheron)  obtained  by  nitrogen  physisorption 
measurements are summarized in Table 3.1.    
The nitrogen physisorption isotherms of the mesoporous silicas in Figure 3.1A can be assigned to type 
IV(a)  isotherms  with  H2(a)  hysteresis  loops.19  Derived  pore  size  distributions  in  Figure  3.1B  have 
relative  standard  deviations  of  59%  (Si60)  and  38%  (Si100).  These  are  typical  values  for  random 
mesoporous  silicas used,  for  example,  in  liquid  chromatography,  where  pore  size  distributions  with 
relative  standard  deviations  of  30–50%  and  higher  are  found,  often  with  a  tail  towards  larger  pore 
sizes.14b,14c,20  
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Table 3.1: Structural properties of the silica powder samples. 

Parameter a  Si60  Si100 

Vt [cm3g–1]  0.82  1.19 

SBET [m2g–1]  793  417 

dmeso
 [nm]  5.9  13.0 

εintra [–]  0.63  0.72 

a Total pore volume (Vt), specific surface area (SBET), and mode mesopore diameter (dmeso) derived from nitrogen 
physisorption analysis; intraparticle porosity (εintra) calculated by εintra = Vt / (Vt + δSi

–1), where δSi is the density of 
amorphous silica (2.13 g cm–3). 

The mean pore size is 5.9 nm for Si60 and 13.0 nm for Si100 (corresponding here to the mode of each 
distribution). These values were subsequently used to calculate λ = dtracer/dmeso. Cumulative pore volumes 
shown in Figure 3.1C confirm the existence of micropores in Si60 (see inset). However, the amount of 
micropores is negligible compared to the main pore volume starting with pores of diameters above 2 nm. 
For Si100, the smallest pore size  is 6 nm. Finally,  the  intraparticle porosities  εintra  (intraparticle void 
volume fractions) are 63% and 72% for Si60 and Si100, respectively. The higher porosity of the Si100 
material can be rationalized with its preparation history based on a pore widening of small­pore silica 
(hydrothermal treatment), which also leaches some of the solid, resulting in wider pores and a somewhat 
higher porosity. 

 

Fig. 3.1: Nitrogen physisorption isotherms (A), pore size distributions (B), and cumulative pore volumes (C) for 
the two silica powder samples. 

3.2.2 Three­dimensional reconstruction of mesopore space morphologies 

For  statistical  reasons,  four  reconstructions  were  prepared  in  total  from  each  silica  material  using 
electron tomography. Two of the reconstructions are shown in Figure 3.2, and the complete set of the 
(eight)  reconstructions  as  well  as  their  dimensions  are  summarized  in  the  Supporting  Information 
(Figure  S3.1  and  Table  S3.1).  Reconstructions  highlighted  in  Figure  3.2  and  Figure  S3.1  clearly 
visualize the different geometrical properties of Si60 and Si100 materials with respect to their mean 
pore  size.  Average  void  volume  fractions,  i.e.,  the  fractions  of  void  voxels  in  the  two  sets  of 
reconstructions, are 60.1% for Si60 and 69.1% for Si100, values that are very similar to the porosities 
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based on nitrogen physisorption analysis (εintra in Table 3.1). The reconstructions slightly underestimate 
the nitrogen sorption data (by ~ 4%), possibly due  to  some smaller pores or pore necks  that escape 
proper identification as void during the reconstruction procedure. For both silicas, we observe a highly 
interconnected, random mesopore network confined by solid (impenetrable) silica walls. 

 

Fig. 3.2: Selected reconstructions of the Si60 (A) and Si100 (B) mesoporous silicas (see Figure S3.1, Supporting 
Information, for the complete set of reconstructions). 

3.2.3 Morphology–transport relationships: Accessible porosity and hindered diffusion 

Direct (pore­scale) simulations of hindered diffusion are conducted for pointlike and finite­size tracers 
that move through the pore space of the reconstructions driven by random Brownian motion. The pore 
space accessible to an ensemble of pointlike tracers equals the entire void volume of a reconstruction. 
However, due to their steric interactions with the pore walls, finite­size tracers have access to only parts 
of that volume, depending on λ = dtracer/dmeso, the ratio of the hydrodynamic diameter of the tracers (dtracer) 
and the mean mesopore diameter (dmeso). To model the accessible porosity for finite­size tracers, the 
reconstructions (serving as geometrical models in the simulations) are subjected to a dilation procedure 
illustrated  by  Figure  3.3.14,21  The  reduction  of  accessible  pore  space  for  finite­size  tracers  can  be 
accounted for by eroding the pore space accessible to pointlike tracers with a structuring element of size 
dtracer. Accordingly, a layer of solid is added to the silica walls (Figure 3.3A and 3.3B), where the layer­
thickness equals half the diameter of the finite­size tracer (Figure 3.3C). By this procedure, the steric 
interaction between the center (point) of the finite­size tracer and the hard silica wall is modelled, i.e., 
the center of a hard spherical tracer particle of diameter dtracer is excluded from a region of thickness 
dtracer/2 at the wall. The remaining porosity corresponds to the volume accessible to a tracer of size dtracer. 
In other words, the void space accessible to the center of a finite­size tracer becomes identical to the 
void space accessible to a pointlike tracer if the pore size reduced by dtracer. The void volume fractions 
(accessible porosities) resulting for finite­size tracers of different diameter dtracer are shown in Figure 
3.4A for both sets of reconstructed silicas. Error bars represent 95% confidence intervals for the four 
reconstructions of each set. Pointlike tracers (dtracer = 0) experience an average porosity of ε0 = 0.601 in 
Si60 and ε0 = 0.691 in Si100. 
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Fig. 3.3: Schematic  illustration of  the dilation procedure. Solid phase (white)  in  the binary  image stack (A)  is 
dilated by a layer of additional solid (red) with thickness ½ dtracer (B). This represents the region adjacent to the 
pore walls, from which the center of a spherical, finite­size tracer of size dtracer is excluded (C). The situation is 
equivalent to using pointlike tracers in the remaining (black) pore space. 

These values are (expectedly) identical to the average porosities calculated from the fractions of void 
voxels in the four reconstructions for each silica. It confirms that the pointlike tracers have access to 
every voxel in a reconstruction. Due to the smaller mean pore size the accessible porosity declines faster 
with increasing dtracer for Si60 than for Si100 (Figure 3.4A). Effective diffusion coefficients simulated 
for  finite­size  tracers  in  the  available pore  spaces  are  shown  in Figure  3.4B  (normalized by  D0,  the 
diffusion coefficients for pointlike tracers in the reconstructions). As expected, the mobility of larger 
tracers is reduced compared to pointlike tracers and decreases more quickly for Si60 than for Si100. In 
both  materials,  diffusion  is  hindered  by  the  intrinsic  confinement  including  the  tortuosity  and 
constrictivity of the pores. 

 

Fig. 3.4: Accessible porosity (A) and simulated effective diffusion coefficient (B) as a function of tracer size dtracer 
for  Si60  and  Si100.  Diffusion  coefficients  are  normalized  by  D0  ≡ Deff,K(dtracer  =  0),  the  diffusion  coefficient 
obtained in the limiting case of dtracer = 0. The gray­shaded regions will become relevant to the discussion in Section 
3.2.4. 
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Figure 3.5 (top row) highlights pore constrictions in both materials (red circles), which become closed, 
i.e., inaccessible pathways for larger tracers (bottom row) due to their enhanced steric interaction with 
the pore walls. Also the zoomed regions (green frames) highlight the substantial decrease of accessible 
porosity and the expected increased hindrance to diffusion for different tracer sizes (top row: pointlike 
tracers, bottom row: dtracer = 1.4 nm and 1.9 nm), as illustrated here by the dilation procedure. Finally, 
the green cylinders in Figure 3.5 with  the respective mean diameter dmeso (from each silica)  indicate 
consistency  between  the  physisorption­based  analysis  and  pore  sizes  found  in  the  physical 
reconstructions.    

 

Fig. 3.5: (Top row) Front view on two original reconstructions of Si60 (A) and Si100 (B) and after application of 
the dilation procedure (bottom row). Red circles highlight constrictions that are open for pointlike tracers (top row, 
dtracer = 0), but closed for finite­size tracers (bottom row, dtracer = 1.4 nm and 1.9 nm). 

To quantify the hindrance to diffusion in both silicas in a dimensionless form, the accessible porosity, 
normalized  by  ε0  ≡  ε(λ  =  0),  and  the  simulated  effective  diffusion  coefficient,  normalized  by 
D0 ≡ Deff,K(λ = 0), were plotted as a function of λ = dtracer/dmeso, the ratio between tracer size and mean 
mesopore diameter. The resulting data sets are summarized in Figure 3.6 and the following second­order 
polynomial fits are obtained for the porosity (Figure 3.6A): 

𝜀(𝜆)

𝜀0
= 1 − 2.02𝜆 + 0.86𝜆2   (Si60)              (2) 
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𝜀(𝜆)

𝜀0
= 1 − 2.56𝜆 + 1.49𝜆2   (Si100)              (3)  

For  the effective diffusion coefficients, we  receive  the  following  third­order polynomial  fits  (Figure 
3.6B): 

𝐷eff,K(𝜆)

𝐷0
= 1 − 1.09𝜆 − 2.54𝜆2 + 3.17𝜆3   (Si60)          (4) 

𝐷eff,K(𝜆)

𝐷0
= 1 − 0.89𝜆 − 5.64𝜆2 + 7.42𝜆3   (Si100)          (5) 

With Eqs. (2) – (5) we can construct the global hindrance factor H(λ) for the two materials (cf. Eq. (1)). 
Because Kd(λ) = Deff,K(λ)/Dm = Deff,K(λ)/D0τ0 (where τ0 denotes the diffusive tortuosity of the pointlike 
tracers), we obtain for Si60: 

𝐻(𝜆) = 𝐾d(𝜆)𝜀(𝜆) =
𝜀0

𝜏0
(1 − 3.11𝜆 + 0.52𝜆2 + 7.36𝜆3 − 8.58𝜆4 + 2.73𝜆5)      (6) 

And for Si100 we receive: 

𝐻(𝜆) = 𝐾d(𝜆)𝜀(𝜆) =
𝜀0

𝜏0
(1 − 3.45𝜆 − 1.87𝜆2 + 20.5𝜆3 − 27.4𝜆4 + 11.1𝜆5)     (7) 

Inserting the respective values of ε0 and τ0, i.e., accessible porosity and diffusive tortuosity for pointlike 
tracers (ε0 = 0.601 and τ0 = 1.69 for Si60, ε0 = 0.691 and τ0 = 1.41 for Si100) into Eqs. (6) and (7) allows 
to quantify hindered diffusion of finite­size, passive tracers into and through both silicas for any λ­value.  

 

Fig. 3.6: Dependence of the accessible porosity (A) and effective diffusive mobility (B) in physical reconstructions 
of the two mesoporous silicas on λ, the ratio of tracer size to mean mesopore diameter. Data are normalized by the 
corresponding values for pointlike  tracers,  ε0 ≡ ε(λ = 0) and D0 ≡ Deff,K(λ = 0) = Dm/τ0 (with  τ0,  the diffusive 
tortuosity of the pointlike tracers). 
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With the panels in Figure 3.6, the global hindrance factor H(λ) = Deff,H(λ)/Dm is obtained by multiplying 
the ε(λ)/ε0­value at a λ of interest in Figure 3.6A with the corresponding value for Deff,K(λ)/D0 in Figure 
3.6B, the result of which represents the value given by the bracket expression in Eq. (6) or Eq. (7), and 
then multiplying the value with ε0/τ0 for a particular material. 

3.2.4 Implications for molecular transport and selectivity in olefin metathesis 

With the available relationships between pore space accessibility, effective diffusion, and tracer size for 
the two mesoporous silicas, Eqs. (2) – (7), we can now discuss consequences for the immobilization, 
formation, and transport of the key molecules 1 – 4 shown in Figure 3.7. Using hydrodynamic diameters 
dH  in bulk solution determined via DOSY­NMR measurements  (see Experimental Section), we  first 
identify corresponding values of λ = dH/dmeso and then calculate accessible porosity ε(λ) as well as the 
local and global hindrance factors Kd(λ) and H(λ) for all species 1 – 4. These data are summarized in 
Table 3.2 for Si60 and Si100. 

 

Fig. 3.7: 2nd­generation Hoveyda–Grubbs type catalyst 1 immobilized inside the mesopores of the silica particles 
and olefin metathesis pathways for  the  α,ω­vinylic prolacton 2  (substrate)  to MMC product 3 and oligomer 4. 
RCM:  ring­closing  metathesis,  ADMET:  acyclic  diene  metathesis,  ROMP:  ring­opening  metathesis 
polymerization, CDP: cyclodepolymerization. 

We begin our discussion with the immobilization of the catalyst 1  inside the mesopores of the silica 
particles, where the trimethoxysilyl tether (Figure 3.7) serves for covalent bonding to the surface silanol 
groups.  Importantly, already  this  first  step (catalyst  immobilization)  is a pore  size­selective process. 
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Caused by a hydrodynamic diameter of dH = 1.08 nm in benzene (solvent used for immobilization), the 
catalyst may be actually size­excluded from a significant number of pores. In Si60, 1 experiences an 
effective porosity of ε(λ) = 0.395. This corresponds to only ~66% of the porosity seen by the pointlike 
tracers (ε0 = 0.601). In contrast, the porosity of the catalyst in Si100 (0.554) still represents 80% of the 
void space accessible to pointlike tracers (ε0 = 0.691). Therefore, we predict a significantly lower loading 
with 1 for the Si60 than for the Si100 particles. In addition, diffusion of 1 into and through the pore 
space of Si60 is only half as fast as for Si100, as shown by the H(λ)­values in Table 3.2. For example, 
the H(λ) of 0.172 for Si60 means that the effective diffusion coefficient of 1, characterizing its global 
process of diffusion into (entering) and through (translocation) the mesopores, is only 0.172 × Dm (cf. 
Eq. (1)), where Dm denotes  the free diffusion coefficient of 1  in  the bulk solution. This implies that 
correspondingly more time should be reserved in protocols for catalyst immobilization in Si60 than in 
Si100. In general, these times should also consider (together with the diffusive hindrance factors) the 
actual  size  of  the  particles  that  shall  be  uniformly  modified  to  accurately  fine­tune  the  overall 
immobilization step. 

Table 3.2: Transport properties of catalyst 1, substrate 2, macro(mono)cyclization product 3, and oligomer 4 (cf. 
Figure 3.7) in bulk solution and in the mesoporous silicas. 

Material  Parameter a  1  2  3  4 

Si60 

dH [nm]  1.08  1.01  0.87  1.49 

λ = dH /dmeso  0.18  0.17  0.15  0.25 

ε(λ)  0.395  0.407  0.432  0.327 

Kd(λ)  0.435  0.447  0.470  0.364 

H(λ)  0.172  0.182  0.203  0.119 

Si100 

dH [nm]  1.08  1.01  0.87  1.49 

λ = dH /dmeso  0.083  0.078  0.067  0.115 

ε(λ)  0.554  0.562  0.579  0.505 

Kd(λ)  0.628  0.634  0.646  0.590 

H(λ)  0.348  0.357  0.374  0.298 
 

a Hydrodynamic diameter dH in bulk solution (deuterated benzene for 1, deuterated toluene for 2–4) from DOSY­
NMR  diffusion  measurements  using  the  Stokes–Einstein  equation;  accessible  porosity  ε(λ),  local  and  global 
hindrance factors Kd(λ) and H(λ) calculated by Eqs. (2), (3), (6), and (7) with ε0 = 0.601 and τ0 = 1.69 for Si60 and 
ε0 = 0.691 and τ0 = 1.41 for Si100. 

While the partial size­exclusion of the catalyst from the mesopore space of the silica particles during 
immobilization  reduces  their  loadability  and  thus  their  active  surface  area,  it  should  not  affect  the 
availability of immobilized catalyst by substrate 2 during the RCM experiments (in absolute toluene as 
solvent). As seen in Table 3.2, the substrate is slightly smaller than the catalyst (1.01 vs. 1.08 nm) and 
should therefore experience no difficulties to reach all catalyst groups that could be immobilized on a 
silica surface. This is confirmed by the very similar accessible porosities  ε(λ) for 1 and 2 in the two 
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silicas  (Table  3.2).  Also  for  the  targeted  MMC  product  3,  still  smaller  than  the  substrate  (0.87  vs. 
1.01 nm),  moving  away  from  the  catalyst  (where  it  is  formed)  and  out  of  the  particles  should  be 
unproblematic. The situation changes, however, for the oligomer 4 that forms by ADMET in sufficiently 
large pores.[18] Due to steric reasons, 4 in Figure 3.7 (and Table 3.2) mostly represents the dimer. With 
dH = 1.49 nm, 4 is ~70% larger than 3 and has, with ε(λ) = 0.327 in Si60 (0.505 in Si100), access to only 
~76% (87%) of the pore space seen by the MMC product 3. Expectedly, also local diffusion of 4 within 
the mesopores (reflected by the Kd(λ)­values in Table 3.2) is slower than for 3 and, as with the accessible 
porosity,  the  mobility  difference  increases  with  the  increasing  confinement  (decreasing  dmeso)  from 
Si100 to Si60: For Si100 (Si60), the local hindrance factor of 4 is 0.590 (0.364) and has thus dropped to 
~91% (~77%) of the corresponding value for 3.   
The  enhanced  confinement  effect  on  the  accessible  porosity  and  effective diffusivity  can  be  clearly 
recognized in Figure 3.4. The gray­shaded region in each panel indicates the range of hydrodynamic 
diameters dH of the key molecular species (cf. Figure 3.7 and Table 3.2) involved in the RCM of interest. 
Importantly, porosity and diffusivity data for Si60 in Figure 3.4 are not just below those for Si100, but 
also their slopes are steeper. That is, the sensitivity of the increased spatial confinement to the size of 
the diffusing tracers is enhanced, as confirmed by the data in Table 3.2. On the other hand, it should be 
realized  that  the  better  size­selectivity  of  Si60  comes  at  the  price  of  a  reduced  mobility  (adversely 
affecting  conversion).  For  example,  the  H(λ)­value  characterizing  diffusion  of  substrate  2  into  and 
through Si60  (0.182) has dropped  to almost 50% of  the value  for Si100  (0.357). As a consequence, 
adequate supply of 2 may not be maintained. Depending on particle size and catalyst reactivity, this can 
severely increase the risk of diffusion limitations.4    
Figure  3.8  illustrates  the  important  effect  of  accessible/excluded  pore  volumes  associated  with 
increasing tracer size (data based on Figure 3.4A for Si60). The representations, obtained for pairs of 
neighbored dtracer­values (cf. Figure 3.4A), highlight pore volumes that become unaccessible when the 
tracer size is increased from dtracer,1 to dtracer,2. The top row visualizes accessible pore networks for a two­
dimensional slice through Si60 (green and red: accessible to dtracer,1, green: accessible to dtracer,2), while 
the bottom row extracts three­dimensional pore volumes that are unaccessible to tracers of size dtracer,2 
(in each panel). Interestingly, tracer sizes in Figure 3.8C represent (approximately) the hydrodynamic 
diameters of substrate 2 and product 4, so that the excluded pore volume visualized in the bottom panel 
of Figure 3.8C identifies pores, in which 2 selectively reacts to the MMC product 3 via direct RCM. 
Due to its size, 4 can neither be formed in these pores nor enter (and diffuse through) them. Figure 8 
therefore reveals that the diffusion networks for 2 and 3 are rather similar, but still significantly different 
(regarding  pore  volume,  network  connectivity,  and  tortuosity)  from  the  transport  pathways  of  4.  A 
conclusion from the previous analysis is that it should be possible to locally trap oligomer 4  in pore 
environments, which enable catalyst 1 and substrate 2 to enter but prevent the larger oligomer molecules 
from leaving. 
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Fig.  3.8:  Accessibility  of  pores  in  Si60  to  tracers  of  different  size.  Red  pores  indicate  regions  that  become 
unaccessible when steric interaction between tracers and solid silica increases from dtracer,1 to dtracer,2. The top row 
highlights accessible pores for dtracer,1 and dtracer,2 in a single slice through the reconstruction, while the bottom row 
visualizes the size­exclusion volumes in cubic boxes with dimensions of 463 nm3. 

The existence of such regions is confirmed with Figure 3.9, which highlights (light blue) two examples 
of pore space in Si60 that can only be entered and left through constrictions (black circles). As indicated 
in the figure, these constrictions are wide enough for 1 (during immobilization) and 2 (during RCM) but 
too narrow for 4. Consequently,  it  is  possible  that oligomer  forms  locally due  to ADMET and  then 
becomes sterically trapped.  

 

Fig. 3.9: Local trapping of oligomer 4 in Si60. Highlighted in light blue are pore regions from which oligomer 
(formed locally within these regions) cannot escape by diffusion, because pore necks (black circles) are too narrow. 
On the other hand, these constrictions are wide enough to allow the passage of catalyst 1 (during immobilization), 
substrate 2 (during RCM), and MMC product 3. Oligomer formed outside these regions (indicated in yellow) can 
move freely through the remaining pore system. 

On the other hand, this occasional event does not comprise a dead end in the synthetic pathway of 4. 
Because  of  equilibrium  RCM,  which  can  convert  the  oligomer  into  MMC  product  through 
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cyclodepolymerization  (Figure  3.7),[22]  and  the  reversibility  of  ADMET  (particularly  in  a  compact 
microreactor without efficient loss of ethylene as the olefinic coproduct), organic matter stored locally 
in the oligomer may not be trapped forever, but can be released via further reactions of 4 resulting in 
sufficiently small molecules. The dynamics of the coupled reactions highlighted in Figure 3.7 becomes 
particularly  important when high substrate concentrations move  into  the  focus due  to  the  increasing 
significance of ADMET. 

3.3 Conclusions 

Presented Eqs. (2) – (7) quantify hindered accessibility and diffusion in realistic geometrical models of 
random mesoporous silicas intended for subsequent application as support in the RCM of substrate 2, 
with catalyst 1 selectively immobilized inside their mesopores. Using the individual values for ε0 and τ0 
in Eqs. (6) and (7), i.e., ε0 = 0.601 and τ0 = 1.69 for Si60 (dmeso = 5.9 nm) and ε0 = 0.691 and τ0 = 1.41 
for Si100 (dmeso = 13 nm), the two expressions quantify the hindrance to diffusion through the respective 
silica material as experienced by finite­size, passive tracers for λ = dtracer/dmeso (ratio between tracer size 
and mean mesopore diameter) up to ~0.7 (Si60) and ~0.6 (Si100). Within these ranges, Eqs. (2)–(7) can 
be  generally  used  to  predict  transport  properties  for  the  employed  silica  particles.  Ideally,  the 
relationships are known before a material  is used as support  in a specific application. The elaborate 
reconstruction–simulation  approach  underlying  the  establishment  of  this  knowledge  is  justified 
considering  that  the  highly  detailed  information  provided  can  become  a  decisive  factor  to  boost 
performance  in  a  process  that  depends  on  efficient  and/or  selective  transport  of  finite­size  solutes.  
The  quantitative  morphology–transport  relationships  have  been  adapted  to  discuss  pore  space 
accessibility and hindered diffusion with respect to the immobilization, formation, and transport of key 
molecular species 1–4 in the RCM of the α,ω­vinylic prolacton 2 to MMC product 3 and oligomer 4, 
with catalyst 1 selectively immobilized inside the mesopores of the silica particles. This analysis has 
shown that already the immobilization step requires careful consideration of size­exclusion effects (on 
the achievable loading of the particles with the catalyst) and diffusive hindrance factors (governing the 
timescale of uniform immobilization throughout the particles). The next important consideration was 
the substrate size. Substrate 2 is slightly smaller than the catalyst, so that it can access all catalytic centers 
inside  the  mesopores.  If  substrate  molecules  were  larger  than  the  catalyst  during  immobilization,  a 
fraction of the catalytic centers would be unavailable for the metathesis pathways. Based on molecular 
size, direct macrocyclization of the substrate 2 to the MMC product 3 is favored over oligomerization 
to 4 in both studied silicas, as reflected by the differences in accessible porosities and diffusive hindrance 
factors for 3 and 4 (Table 3.2). This preference is more distinctive for Si60 than for Si100, which is 
recognized in the steeper slopes of the accessible porosity and hindered diffusivity curves for Si60 in 
the range of tracer sizes relevant to 3 and 4 (gray­shaded regions in Figure 3.4). However, improved 
size­selectivity  is accompanied by  reduced mobility and  there appears  to be a fine  line between  the 
antagonistic effects on selectivity and conversion induced by the confinement. As a particular feature of 
the confinement, we identified by direct image analysis pore regions in the silica that could trap (locally 
formed) oligomer due to size­selective passage resulting from constrictions. To what extent this trapped 
matter will be detected after an experiment depends on the timescale of the experiment and the dynamics 
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of further reactions of the oligomer.    
Importantly,  the  derived  morphology–transport  relationships  can  be  applied  to  other  catalysts, 
substrates,  and  products  in  straightforward  manner  to  assess  transport  resistances  and  transport 
selectivities. For that purpose, the hydrodynamic diameters of key molecular species (required for λ in 
Eqs.  (2)  –  (7))  can  be  estimated  in  the  solvent  of  interest  using  DOSY­NMR.  Together  with 
complementary experimental data on conversion and selectivity, the deduced information will help to 
unravel complex reaction pathways and to better understand mechanistic aspects of olefin metathesis, 
as  addressed  in  Figure  3.7.  Furthermore,  if  also  relevant  reaction  kinetic  data  are  available,  the 
morphology­based  hindered  diffusion  coefficients  will  improve  relationships  between  effectiveness 
factor and Thiele modulus and thereby the predictions on microreactor operation (diffusion vs. reaction 
control) and the proper selection of the particle size.    
Currently, we use the silica particles in RCM experiments in a continuous­flow microreactor setup that 
allows to assign conversion and selectivity to well­defined reaction times and, thus, to analyze multiple 
reaction times and temperatures fully automated in a single flow experiment. The combined approach 
to morphology–transport–functionality relationships will pave the way to fast and systematic screenings 
of  catalyst  activity,  conversion,  and  selectivity  in  dependence  of  spatial  confinement  effects,  in 
particular, with a straightforward variation of the mesoporous support, type and loading of catalyst, type 
and concentration of substrate, solvent, reaction time, and temperature. Results will be reported in due 
course. 

3.4 Experimental Section 

Material characterization: Nitrogen physisorption measurements were carried out at 77 K on a Thermo 
Scientific Surfer gas  adsorption porosimeter  (Thermo Fisher Scientific, Waltham, MA). Prior  to  the 
measurements, the samples were evacuated for 10 h at 250 °C. Total pore volumes (Vt) were obtained 
with the Gurvich rule at a relative pressure of p/p0 = 0.95, specific surface areas (SBET) were determined 
by  means  of  the  Brunauer–Emmett–Teller equation in the range of 0.05 ≤ p/p0  ≤ 0.3. Pore  size 
distributions were derived from the adsorption branches of the physisorption isotherms using the non­
local density functional theory (NLDFT) method with a cylindrical pore model.23 

Electron tomography: Scanning transmission electron microscopy (STEM) tomography was employed 
to obtain physical reconstructions of two silicas.12 For preparation, powder samples were ground in a 
mortar and subsequently dusted over a holey Cu grid  (Quantifoil Micro Tools,  Jena, Germany). Au 
fiducial markers were added to the sample crumbs from aqueous solution. Electron tomographic data 
were collected using an image­corrected Titan 80–300 transmission electron microscope (FEI, Hilsboro, 
OR) at an acceleration voltage of 300 kV. The microscope was operated in STEM mode with a nominal 
beam diameter of 0.2 nm. STEM tilt­series were recorded over a range of ±76° (Si60) and from –76° to 
70° (Si100) in increments of 2° with a high­angle annular dark­field collector. Images were then aligned 
in IMOD 4.7 by means of the Au reference markers.24 The average alignment error was <1 pixel. After 
applying a sufficient number of iterations (100 for Si60 and 25 for Si100) of the SIRT­algorithm to the 
aligned image series using the Xplore3D software package (version 3.0, FEI), the reconstructions were 
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denoised  by  employing  the  ImageJ  plugins  Filters/Median  3D  and  Enhance  Local  Contrast 
(CLAHE).25,26  Final  reconstructions  were  received  after  applying  10  iterations  of  DART  using  FiJi 
software  and  the  ImageJ  plugin  TomoJ.27  For  each  sample,  four  cuboids  were  prepared  as  final 
reconstructions, covering different regions within the crumbs that were used for electron tomography. 
Image stacks had a voxel size of 0.233 nm3 for Si60 and 0.323 nm3 for Si100 (Figure S3.1, Supporting 
Information)  and  served  as  geometrical  models  for  morphological  analysis  and  the  pore­scale 
simulations of hindered diffusion. 

Diffusion  simulations:  A  random­walk  particle­tracking  (RWPT) method  was  employed  to  simulate 
diffusive transport through the pore volumes of the eight reconstructions (cf. Figure S3.1).28 The method 
is based on the equivalence of the diffusion equation (where D denotes the diffusion coefficient and c 
concentration) 

𝜕𝑐(𝐫,𝑡)

∂𝑡
= 𝐷∇2𝑐(𝐫, 𝑡)                  (8) 

and the stochastic differential equation describing the random walk of a  tracer, presented in discrete 
form as 

𝐫(𝑡 + δ𝑡) = 𝐫(𝑡) + ∆𝐫 =  𝐫(𝑡) + 𝛂√6𝐷δ𝑡            (9) 

In Eq. (9), r(t) denotes tracer position at time t, δt is the elementary time step of the random walk, and 
𝛂 is a vector with random orientation in space and a length governed by a Gaussian distribution with 
zero mean and unity variance. Initially, a large number of N = 106 passive (nonadsorbing, nonreacting) 
tracers  was distributed  randomly  and  uniformly  in  the  entire  void  space of  a  reconstruction.  At  the 
external faces of a reconstruction, the tracer displacement was handled by a mirror­boundary condition, 
which has been validated before and assumes that tracers crossing an external face of a reconstruction 
continue their motion in the mirror image of the original domain.14a It enables complete sampling of the 
accessible  pore  space  by  the  tracer  ensemble  at  any  time  and  therefore  reflects  spatially  averaged 
morphological properties of a reconstruction along with their impact on diffusive transport. During each 
time step δt (small enough that the mean diffusive displacement did not exceed Δh/10, where Δh is the 
spatial resolution of a reconstruction), the displacement of every tracer by random diffusive motion was 
determined according to Eq. (9). A multiple­rejection boundary condition at  the surface was used to 
model  passive  interaction  of  the  tracers  with  the  impermeable  pore  walls:29  when  a  tracer  hit  the 
impermeable  wall  during  an  iteration,  the  displacement  was  rejected  and  recalculated  until  tracer 
position was in the void space. The accuracy of the employed RWPT­approach including the described 
boundary conditions has been confirmed by comparing diffusion coefficients simulated in regular arrays 
of  spheres30  with  values  determined  by  the  analytical  approach.31  After  each  time  step,  all  tracer 
positions were monitored and used to calculate a time­dependent diffusion coefficient D(t) from 

𝐷(𝑡) =
1

6𝑁

d

d𝑡
∑ [∆𝑟𝑖(𝑡)]2𝑁

𝑖=1                 (10) 
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where Δri(t) represents the displacement of the ith tracer after time t. The targeted effective, i.e., time­
independent  diffusion  coefficients  Deff,K  (locally  within  the  mesopore  network  of  a  reconstruction, 
subscript K) were determined from the asymptotes of the transient diffusion curves observed in the long­
time limit, as illustrated in Figure S3.2 (Supporting Information).    
In contrast to pointlike tracers, which can access the entire void space in a reconstruction, the void space 
accessible to finite­size tracers is smaller due to their steric interaction with the impermeable pore walls. 
Considering the case of diffusion in a cylindrical pore, the center of a hard spherical tracer particle of 
diameter dtracer is excluded from an annular region of thickness dtracer/2 at the wall, i.e., the void space 
accessible to the center of a finite­size tracer becomes identical to the void space accessible to a pointlike 
tracer if the pore diameter is reduced by dtracer. An extension of this isomorphism concept to random 
porous media has been proposed.21 According to this approach, the reduction of accessible pore space 
for finite­size tracers can be accounted for by eroding pore space accessible to pointlike tracers with a 
structuring element of size dtracer (as illustrated in Figure 3.3).14a,32 We implemented this mathematical 
morphology operation to generate accessible mesopore space  in  the eight reconstructions (cf. Figure 
S3.1) for dtracer­values from 0 to 4.14 nm in 0.46­nm steps (Si60) and from 0 to 7.68 nm in 0.64­nm steps 
(Si100).    
By adapting this approach for the RWPT­simulations, accessible porosity ε(λ) and effective diffusion 
coefficient within a reconstruction Deff,K(λ) were determined for different tracer sizes as a function of 
λ = dtracer/dmeso, the ratio between tracer size and mean mesopore diameter (cf. Figures 3.4 and 3.6). With 
the λ­dependent values of the local hindrance factor, Kd(λ) = Deff,K(λ)/Dm, the associated global hindrance 
factor H(λ) could be derived (cf. Eqs. (6) and (7)) through the accessible porosity ε(λ), determined as the 
void volume fraction of an eroded pore space at this λ. The program realization of the RWPT­algorithm 
was implemented as a parallel code in C language using the Message Passing Interface (MPI) standard 
on a supercomputing platform at the Philipps­Universität Marburg (Marburg, Germany). 

Diffusion measurements: Diffusion coefficients in bulk solution (deuterated benzene for 1, deuterated 
toluene for 2–4) were determined via DOSY­NMR carried out at 298 K on a Bruker Avance III 400 
spectrometer. For  these measurements, homogeneous catalyst  1  and a mixture of  substrate 2, MMC 
product 3, and oligomer 4 have been prepared as described previously.18 Hydrodynamic diameters dH 
(cf. Table 3.2) were then calculated from these diffusion coefficients in bulk solution by application of 
the Stokes–Einstein equation. 
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3.5 Supporting Information 

 

 

 

Fig. S3.1: Cuboids from different regions of the Si60 sample (top row) and the Si100 sample (bottom row), which 
were used as final reconstructions for morphological analysis and the pore­scale simulations of hindered diffusion.  

 

Table S3.1: Dimensions of the final reconstructions (x × y × z nm3). 

Crop  Si60  Si100 

A  100 × 117 × 124  96 × 116 × 116 

B  97 × 95 × 96  96 × 102 × 108 

C  77 × 76 × 72  132 × 156 × 156 

D  79 × 76 × 166  135 × 148 × 141 
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Fig. S3.2: Transient diffusion coefficient D(t) normalized by Dm, the diffusion coefficient for free diffusion in the 
bulk liquid, simulated for dtracer = 2.56 nm in crop A of sample Si100 (cf. Figure S3.1). After a short diffusion time 
(< 10 µs), the tracers have sampled the microstructural heterogeneity of the pore space and the normalized effective 
diffusion coefficient Deff,K/Dm is reached (dashed red line).  

Figure S3.2 illustrates the simulation of Deff,K for dtracer = 2.56 nm in one of the crops from sample Si100, 

assuming a tracer diffusion coefficient in the bulk liquid of Dm = 1.5 × 10–10 m2 s–1. The figure reveals a 

steep decrease of D(t) at the beginning of the simulation (see inset) due to initial contact of the tracers 

with the impenetrable walls. However, within less than ten microseconds, the tracers have experienced 

the microstructural heterogeneity of the pore space and asymptotic behavior is observed. The diffusion 

time required to reach the asymptotic behavior translates into a diffusion length ldiff = (2Dmt)0.5 of a few 

tens of nanometers or 3–4 pore diameters  (dmeso),  also  indicated  in  the  figure. This diffusion  length 

reflects a pore space characterized by a short­ranged heterogeneity, which is quickly sampled by the 

tracers  to  attain  quasi­homogeneous  behavior  and  document  effective  diffusion  properties.  The 

asymptotic D(t)­value corresponds to Deff,K. 
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Abstract 

Hypothesis:  Although  many  synthetic  pathways  allow  to  fine­tune  the  morphology  of  dendritic 
mesoporous  silica  nanoparticles  (DMSNs),  the  control  of  their  particle  size  and  mesopore  diameter 
remains a challenge. Our study focuses on either increasing the mean particle size or adjusting the pore 
size distribution, changing only one parameter (particle or pore size) at a time. The dependence of key 
morphological features (porosity; pore shape and pore dimensions) on radial distance from the particle 
center has been investigated in detail. 

Experiments:  Three­dimensional  reconstructions  of  the  particles  obtained  by  scanning  transmission 
electron microscopy (STEM) tomography were adapted as geometrical models for the quantification of 
intraparticle  morphologies  by  radial  porosity  and  chord  length  distribution  analyses.  Structural 
properties of  the different synthesized DMSNs have been complementary characterized using TEM, 
SEM, nitrogen physisorption, and dynamic light scattering. 

Findings:  The  successful  independent  tuning  of  particle  and  pore  sizes  of  the  DMSNs  could  be 
confirmed  by  conventional  analysis  methods.  Unique  morphological  features,  which  influence  the 
uptake and release of guest molecules in biomedical applications, were uncovered from analyzing the 
STEM  tomography­based  reconstructions.  It  includes  the  quantification  of  structural  hierarchy, 
identification of  intrawall openings  and pores,  as well  as  the distinction of pore  shapes  (conical vs. 
cylindrical). 

4.1 Introduction 

Mesoporous  silica  nanoparticles  (MSNs)  have  received  great  attention  during  the  past  decades, 
documented by applications in drug delivery, nanomedicine, catalysis, and separation science.1–4 Many 
characteristics  like  their biocompatibility,  stability, high surface area, easy surface  functionalization, 
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and  a  tunable  morphology  (e.g.,  particle  diameter  and  shape;  mesopore  size)  make  them  attractive, 
widely  applicable  materials,  and  research  on  MSNs  with  singular  shapes  as  well  as  unique 
morphological features has triggered novel applications.5–18 Particularly the investigation of MSNs as 
efficient, smart drug delivery systems has been intensively pushed forward during the past decade,1–3,19–

27 where the incorporation of large guest molecules such as proteins or genes requires tailorable pore 
sizes and enhanced access to the intraparticle surface.28–31    
Since 2010, MSNs with peculiar architectures in the form of well­organized, dendrimer­like structures 
have been increasingly spotlighted because of their large accessible surface,  tunable pore size, small 
particle diameter, and good stability.30,32–37 The center­radial hierarchical pore system of such dendritic 
mesoporous silica nanoparticles (DMSNs) consists of mesopores with adjustable diameters (from 3 to 
25  nm)  and  their  radially  altering  morphology  enables  the  controlled  release  of  encapsulated 
molecules.23,29,37 Further, their large pore volume and increased surface accessibility allow the uptake of 
large molecules including DNA, RNA, or insulin, for example.10,30,35,38    
Two parameters still limit the applicability and performance of silica nanocarriers: the particle size and 
mesopore  diameter.2,26,29  During  transport  through  the  human  body,  nanocarrier  size  is  important  in 
crossing cellular membranes, escaping from the reticuloendothelial system, or prolonging the sojourn 
time in blood circulation.2,39 On the other hand, the extent of drug uptake and controlled release strongly 
depends on intraparticle pore sizes and the architecture of the silica support.33,40 Although many studies 
reveal solutions for tuning a single parameter, e.g., pore size control by employing swelling agents,41–43 
use of co­surfactants  to obtain small particle  sizes,44,45 or  the application of base catalysts  to ensure 
narrow particle size distributions and high yields,46,47 their independent control to combine features like 
a narrow particle size distribution, small particle sizes, and large mesopore diameters in a single MSN 
system is still a challenge.33,37,48,49 In our study, we focus on two complementary approaches to control 
and  tune  (i)  the  particle  size,  while  maintaining  the  pore  size  distribution,  and  (ii)  the  pore  size 
distribution  while  preserving  particle  size.  The  syntheses  have  been  carried  out  in  a  typical 
microemulsion system, containing a cationic surfactant (cetyltrimethylammonium chloride, CTAC) and 
a base catalyst  (triethanolamine, TEA)  in  the aqueous phase and an organic solvent  (cyclohexane or 
hexane) with the silica source (tetraethylorthosilicate, TEOS) as oil phase.37–50 Formation of the DMSNs 
is assumed to proceed through several steps (Fig. 4.1). TEOS species diffuse from the oil phase into the 
aqueous phase, where  they are hydrolysed. Silicate  species  then  interact with CTA+  cations  to  form 
hybrid micelles. At least two hybrid micelles aggregate to primary particles. Due to pH reduction and 
enhanced competition between chloride and silicate ions for CTA+ micelles, partially silicated micelles 
begin to form and add to primary particles, resulting in larger mesopores. After template removal and 
thermal  treatment,  mesoporous  DMSNs  are  obtained.  The  proposed mechanism,  based  on  the  well­
established {S+, mX–, (1–m)I–}­pathway (S+ ≡ CTA+; X– ≡ Cl–; I– ≡ SiO–), is adapted in our study for 
the interpretation of morphological features.37,50,51 
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Fig. 4.1: Schematic illustration of the formation mechanism for DMSNs.37 Step 1: Diffusion of silica species into 
the water phase and interaction between silica and micelles. Step 2: Aggregation of silicated micelles and formation 
of primary particles. Step 3: Partly  silicated micelles  add  to primary particles  and  lead  to dendritic  growth of 
surfactant­filled mesopores. The DMSNs are obtained after removal of the surfactant and calcination. 

As illustrated in Fig. 4.2, the synthesis of DMSN1 served as starting point for the preparation of all other 
DMSNs. Accordingly, a seed­growth synthesis was carried out with DMSN1 as seed particle (left side, 
Fig. 4.2) to increase particle size without affecting the distribution of pore sizes (DMSN@DMSN1). 
The multigenerational seed­growth technique has been reported35,52–54 as a straightforward approach to 
enlarge particle size in a controllable manner by using as­synthesized DMSNs prior to template removal, 
drying, and calcination as nucleus particles and initiate further growth of silica onto their surfaces. On 
the other hand, the synthesis of DMSN1 was adapted with reduced stirring rate to realize a monomodal 
pore size distribution (DMSN2), and this protocol was applied with a different organic solvent to adjust 
the mean mesopore size (DMSN3). In both cases, mean particle size should be little affected (right side, 
Fig. 4.2).    
Although electron tomography is an established tool for elaborate studies of nanomaterials, for example, 
regarding intraparticle morphological properties or the location of guest molecules in the mesoporous 
structures,55–59  data  on  a  discrete  radial  evolution  of  structural  parameters  within  DMSNs  are  still 
lacking.  By  using  scanning  transmission  electron  microscopy  (STEM)  tomography,  we  derived 
reconstructions  of  four  different  DMSNs  prepared  under  the  scope  of  the  approaches  in  Fig.  4.2, 
focusing on tuning either the particle size or the pore size distribution. These spherical reconstructions 
served as geometrical models in the subsequent radial analysis of morphological descriptors, including 
porosity (void volume fraction) and the size and shape of the intraparticle mesopores. As geometrical 
measure  of  pore  morphology,  we  employed  the  mean  chord  length  obtained  from  chord  length 
distributions (CLDs). CLD analysis can be used to investigate arbitrary structures without the need to 
assume an underlying geometrical model.60 We implemented CLD analysis within concentric shells of 
the  reconstructions  to  derive  conclusions  about  the  radial  propagation  of  pore  morphology  and,  in 
particular, to differentiate between cylindrical and conical pore shapes. This fine insight into the pore 
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space  morphology  is  of  fundamental  interest,  as  the  performance  of  DMSNs  in  all  applications  is 
strongly related to their pore­scale structural features, which govern species uptake by (and release from) 
the particles. 

 

Fig. 4.2: Synthesis strategies for the DMSNs under the aspect of particle size tuning (DMSN@DMSN1) and pore 
size  tuning (DMSN2, DMSN3). DMSN@DMSN1 is prepared by seed­growth using DMSN1 as  seed particle. 
DMSN2 is obtained by following the synthesis protocol for DMSN1 with reduced stirring rate, and for DMSN3 
the synthesis protocol for DMSN2 was adapted with different organic solvent. Shown are cuts through the particles 
of each sample reconstructed by electron tomography. 

The radial analysis procedures were combined and consolidated with results from conventional methods, 
involving  transmission  electron  microscopy  (TEM),  scanning  electron  microscopy  (SEM),  nitrogen 
physisorption, and dynamic light scattering (DLS). We uncover and quantify important morphological 
features of the DMSNs at the nanoscale, filling a gap within existing studies. The presented approaches 
and data offer the prospect of systematic improvements in DMSN performance. An extension of this 
work  to  pore­scale  simulations  of  solute  diffusion,  using  the  reconstructions  obtained  by  electron 
tomography  as  realistic  geometrical  models,  will  help  to  better  predict  transport  properties  of  the 
DMSNs, which are intrinsically related to their morphological characteristics. 

4.2 Methodology 

4.2.1 Material synthesis 

Synthesis of DMSN1. First, the aqueous starting solution (cf. Fig. 4.1) was prepared by mixing 8 ml of 
cetyltrimethylammonium chloride (CTAC, 25 wt % in H2O, Sigma­Aldrich), 360 mg of triethanolamine 
(TEA, > 98%, Alfa Aesar), and 72 ml of H2O for 1 h at 60 °C (IKA stirring plate, stirring rate: 375 rpm). 
Then, the organic solution was prepared, i.e., 32 ml cyclohexane (> 99%, Fisher Scientific) and 8 ml 
tetraethylorthosilicate (TEOS, > 99%, Sigma­Aldrich) were mixed and added dropwise to the aqueous 
phase.  The  resulting  solution  was  kept  overnight  at  60 °C  under  continuous  stirring.  Prior  to 
centrifugation at 10,000 rpm for 20 min, the organic phase was removed. The resulting white product 
was dried overnight at 100 °C and used in the subsequent synthesis. For characterization, ~100 mg of 
the  as­made  product  were  collected  and  the  surfactant  was  extracted  with  100  ml  ethanol 



Chapter 4 

110 
 

containing   1 drop HCl (37%, Sigma­Aldrich) under reflux at 90 °C. Finally,  the nanoparticles were 
dried overnight at 100 °C and calcined under air at 550 °C for 5 h (ramp: 1 °C min­1).    
Synthesis of DMSN@DMSN1. The synthesis was carried out as described above, except that 300 mg of 
the  as­made product  were  added  to  the  aqueous  phase.  After  a  first  drying  step,  the  surfactant  was 
extracted under reflux for 2 h with 100 ml ethanol containing 1 drop HCl (37%). After drying overnight 
at  100 °C,  the  resulting  nanoparticles  were  calcined  under  air  at  550 °C  for  5 h  (ramp:  
1 °C min–1).    
Synthesis of DMSN2. The synthesis was carried out similar to that of DMSN1, except that the stirring 
rate was reduced from 375 to 150 rpm.   
Synthesis of DMSN3. The synthesis was carried out similar to that of DMSN2, except that cyclohexane 
was replaced by hexane (> 99%, Fisher Scientific) in the organic phase. 

4.2.2 Material characterization 

The DMSN powder samples were characterized using TEM, SEM, nitrogen physisorption, and DLS. 
For TEM, the particles were dispersed in ethanol (0.7 mg ml–1). This suspension (4 μL) was deposited 
onto a carbon­coated copper grid and images were recorded with a transmission electron microscope 
(Philips CM200) at an acceleration voltage of 200 kV. TEM images of the samples are shown in Fig. 
S1 and S2 (Supplementary Material). Complementary SEM images of the DMSN@DMSN1 particles 
are provided by Fig. S3. They were obtained for the uncoated powder sample on a Merlin VP Compact 
FE­SEM (Carl Zeiss, Jena, Germany) at acceleration voltages of 1­2 kV.    
Nitrogen physisorption isotherms were measured at –196 °C using an Autosorb­iQ3 sorption analyzer 
(Quantachrome Instruments, Boynton Beach, FL). Prior to that analysis, DMSNs were outgassed for 
10 h at 150 °C. Specific surface area SBET was determined using the Brunauer–Emmett–Teller (BET) 
equation  in  the  relative  pressure  range  0.05 ≤ p/p0 ≤ 0.2  and  total  pore  volume  was  determined  at 
p/p0 = 0.95.  Pore  size  distributions  were  calculated  using  the  non­local  density  functional  theory 
(NLDFT) method with a cylindrical pore model.61,62 DLS experiments were performed on a Malvern 
DTS Nano Zetasizer 173°. Samples were dispersed in water (0.7 mg ml–1), shaken, and sonicated prior 
to the measurements of the hydrodynamic diameters (equilibrium time, 3 min; three measurements for 
each sample). 

4.2.3 Electron tomography 

Physical  reconstructions  of  the  DMSNs  were  obtained  by  STEM  tomography.63–65  Prior  to  the 
tomography,  aggregated powder  samples were ground  in  a mortar  and dusted over  a holey Cu  grid 
(Quantifoil Micro Tools,  Jena, Germany). Au nanoparticles were  added  to  the  sample  crumbs  from 
aqueous solution and served as fiducial markers for the reconstruction. Electron tomography was run on 
an  image­corrected  Titan  80–300  transmission  electron  microscope  (FEI,  Hillsboro,  OR)  at  an 
acceleration voltage of 300 kV. The microscope was operated  in STEM mode with a nominal beam 
diameter of 0.2 nm. STEM tilt­series of the samples were recorded over a range of ±76° (DMSN1 and 
DMSN3), from –74° to 76° (DMSN@DMSN1), and from –70° to 76° (DMSN2) in increments of 2° 
using a high­angle annular dark­field (HAADF) collector. Images were aligned in IMOD 4.7 by means 




